# Design and Performance Analysis of Passive Optical Networks using Dicode coding

Submitted By

### MUKESH KUMAR GUPTA

Research Scholar

(ID-2012REC9041)

Under Supervision of

Dr. Ghanshyam Singh

Associate Professor, Dept. of ECE, MNIT Jaipur

Submitted for the partial fulfillment of degree of

**Doctor of Philosophy** 

То



### DEPARTMENT OF ELECTRONICS & COMMUNICATION ENGINEERING MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR

Sept. - 2017

# Design and Performance Analysis of Passive Optical Networks using Dicode coding

Submitted By

## MUKESH KUMAR GUPTA

Research Scholar

(ID-2012REC9041)

Under Supervision of

Dr. Ghanshyam Singh

Associate Professor, Dept. of ECE, MNIT Jaipur

Submitted for the partial fulfillment of degree of

**Doctor of Philosophy** 

То



### DEPARTMENT OF ELECTRONICS & COMMUNICATION ENGINEERING MALAVIYA NATIONAL INSTITUTE OF TECHNOLOGY JAIPUR

Sept. - 2017

© Malaviya National Institute of Technology Jaipur - 2017.

All rights reserved

### Certificate

It is certified that:

- 1. This thesis has not been submitted in part or full to any other University or Institute for the award of any degree.
- 2. I have fulfilled all the requirements for submission of the thesis.

-frankesh

Mukesh Kumar Gupta ID: 2012REC9041

This is to certify that the thesis entitled "DESIGN AND PERFORMANCE ANALYSIS OF PASSIVE OPTICAL NETWORKS USING DICODE CODING" being submitted by Mr. Mukesh Kumar Gupta to the Department of Electronics and Communication Engineering, Malaviya National Institute of Technology Jaipur for the award of the Degree of Doctor of Philosophy, is a bonafide research work carried out by him under our supervision and guidance. The results obtained in this thesis have not been submitted to any other University or Institute for the award of any other Degree.

Dr. Ghanshyam Singh Associate Professor & Supervisor Department of ECE, MNIT, Jaipur

Thesis was examined and approved for the award of the Degree of Doctor of Philosophy.

Dr. Ajeet Kumar Assistant Professor & External Examiner Department of Applied Physics, Delhi Technological University (DTU), Formerly DCE, New Delhi, 110042

Scanned by CamScanner

# **Declaration of Authorship**

I, Mukesh Kumar Gupta, declare that, this thesis titled as, 'Design and Performance Analysis of Passive Optical Networks using Dicode coding' and the work presented in it are my own. I also confirm that:

- This work was done wholly or mainly while in candidature for a research degree at this University.
- Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated.
- > Where I have consulted the published work of others, this is always clearly attributed.
- Where I have quoted from the work of others, the source is always given. Except such quotations, this thesis is entirely my own work.
- ➤ I have acknowledged all main sources of help.

Mukesh Kumar Gupta

## Acknowledgements

Pursuing Ph.D. has been a truly life-changing experience for me and it would not have been possible to do without the support and guidance that I received from many people. Here I would like to thank all those who have become an integral part of my memories while working on my Thesis.

First and foremost, I thank my academic supervisor, Dr. Ghanshyam Singh, Associate Professor, Department of Electronics and communication engineering, MNIT Jaipur, for accepting me into his group. During my tenure, he contributed to a rewarding graduate school experience by giving me intellectual freedom in my work, supporting my attendance at various conferences, engaging me in new ideas, and demanding a high quality of work in all my endeavors. His motivation and encouragement have driven me this far. He always stood before me in all the instance of my life whether it is a professional or a personal one. His motivating words always energize me to go beyond and perform well in every field. Without his guidance and constant feedback, I could not have done anything.

I am especially indebted to my thesis evaluation committee members, who have been simply unreal. I would like to thank DREC members Dr. Vijay Janyani, Dr. Ritu Sharma and Dr. Ravi Maddila for their time and support for providing me their valuable comments and advice. During a discussion on my ongoing work, their valuable comments always motivate me to think extensively on the topic. They have always been very gracious and generous with their time, ideas, and recommendation. Thank you so much to all of them.

I would like to express my gratitude to the Head of Department Prof. KK Sharma, Prof. Vineet sahula (Ex. HOD.) and DPGC convener Prof. D. Boolchandani who were always helpful and provided me the required research-oriented environment. I would also like to thank all the faculty members of ECE dept. MNIT Jaipur for their academic support.

I also greatly appreciate Dr. Dashkov Michael, from Povolzhskiy State University of Telecommunications and Informatics, Samara (Russia), for his invaluable suggestions and comments on my research article and shared his vast experimental knowledge related to optical networks. I am certainly appreciative and thankful to all my friends Amit Garg, Nikhil Deep Gupta, Sourabh Sahu, Shanky Saxena, Ambrish Kumar, Bipin Kumar, Sandeep Vyas and Shruti Kalra for the numerous ideas and for helping me enormously throughout this work. Apart from that, I am also thankful to all other fellows of my lab "Photonic THz Communication" for untimely discussion on the topics.

I am deeply grateful to my parents for always believing in me and encouraging me to follow my dreams. Thank you so much for encouraging me to pursue my education, for teaching me the value of experiences, and for their love. I would also extend my thanks to my parents-in-law and my brother-in-law for their unyielding support and helping in whatever way they could during this challenging period. A special word of thank to my sister in law Manisha and Ashishji for their support whenever needed.

Most of all, many many thanks to my wife, Jyoti, who has been by my side throughout this PhD, living every single minute of it, and without whom, I would not have had the courage to embark on this journey in the first place. She has graced my life with her support, understanding, and love. I thank her for her sacrifice as I pursued my goals and dreams, hope to return such generosity to you in our new life together. I am blessed to have become a father during my Ph.D. Thank you son for your company, laughter, and love. I wish to thank my brother Dr. Rajesh and sister in-law Dr. Ishita for their care and motivation.

Finally, I express my deep sense of gratitude to God Almighty for giving me the strength, knowledge, ability and opportunity to undertake this research study and to persevere and complete it satisfactorily. May God give me ample prudence and strength to realize my dreams.

My thanks go to all the people who have supported me to complete the research work directly or indirectly, a big "Thank you" from the bottom of my heart.

Mukesh Kumar Gupta

## **Abstract**

The rapid growth in access network demands huge bandwidth to provide voice, data and video (triple play) services for residential and corporate subscribers. The passive optical network (PON) has emerged as one of the most promising solution for higher bandwidth requirement for the optical broadband access network. Passive optical network is a point-to-multipoint tree topology network that connects optical line terminal (OLT) with many optical network units (ONUs) via a long fiber feeder and many short distribution fibers. Passive Optical Networks is cable to provide a lucrative, highly flexible and scalable network for end subscribers. Practical implementation of PON gives rise to various technical challenges, such as colorless ONUs, burst mode transmission, bi-directional transmission with mitigated backscattering noise, long-reach PON, and integrating network functionalities. All such technical requirements are motivated by the "original requirements" of telecommunication – faster, cheaper, and more robust.

To fulfill the technical requirements, researchers investigated various aspects of system designs and studied the upcoming technologies such as devices, system architectures, network protocols, etc. This thesis deals with the performance evaluation of PON systems by designing new structures, finding out suitable modulation format to improve the overall performance by mitigating & analyzing the crosstalk and implementing with different topologies with tracking several system level challenges while acknowledging the limitations of existing networks. In specific, we have studied Non-return to zero (NRZ), Dicode coded NRZ, Duobinary, and Manchester formats, including the modulation/demodulation techniques, transmission properties, and system applications. The research topics are classified according to the capacity of the network with different applications.

In the investigation at the system level, where many components are involved, it is difficult to describe the system performance fully. The system performance is investigated with the aid of split-step Fourier (SSF) simulation. The performance is evaluated on the basis of two criteria. The first criteria sets the probability of bit error below  $10^{-9}$  and in second, timing recovery is facilitated by preserved eye profile at the receiver output.

Firstly, dicode – coded NRZ format is proposed for optical signal transmission. It has desirable properties of large dispersion and Raman crosstalk tolerance (as compared to Conventional NRZ/RZ) and finite optical power in each bit. In this study, we first show the advantages of dicode coded NRZ and the corresponding modulation techniques. We analyzed the effects of dispersion on conventional wavelength division multiplexing-time division multiplexing (WDM-TDM) system at a data rate of 10 Gbps using NRZ and dicode. Further, we demonstrate a 10-Gbps video overlay PON system to provide voice, data, and video (triple play) services to end user in two different wavelength bands (1480 nm and 1550 nm) for data and video. The video has high power, and this band lies in the Stokes shift ~100 nm of silica hence the data signal act as a pump for video, resulting in the reduction of the power of data signal. This reduction in power can be minimized by dicode, and remote Raman pump can be used as an amplifier to increase the reach of network.

Secondly, we propose a design of high capacity PON using dicode as ring WDM-PON and conventional WDM-PON. Ring WDM-PON can be used in mitigation of Rayleigh backscattering power by using downstream and upstream signal in the same direction as well as using the concept of remodulation at ONU as a light source used at OLT to avoid the problem of colorless ONU. Optical network unit is simple to operate and do not requires laser sources for generation of the carrier at its end. A carrier at the central office is remodulated as upstream transmission at the subscriber end. In this way, by using the additional carrier at Optical Line Terminal (OLT), we have reduced the laser sources equal to the number of ONUs. Dual fiber ring is used to avoid any fiber fault. An 80 Gbps conventional WDM-PON system is designed with two 40 Gbps WDM channels using FBG (Fiber brag grating). Dicode coded NRZ and FBG as a dispersion compensator, can achieve 1500 Km long transmission system.

In the present scenario, concepts of hybrid (WDM/TDM) PON which uses the advantages of both TDM as well as WDM network, are under investigation by the scientific community at large. It is utmost necessary to design, characterize and investigate hybrid PON for high capacity access systems. Finally, we design and analyzed numerically a cost effective 77 KM reach flexible hybrid WDM-TDM-PON system for 64×4 subscribers at 10 Gbps down -stream and upstream signal. The cost of OLT is better shared among the ONUs as used in higher splitting ratio. The proposed

system is very simple, flexible, expandable and profitable. This design uses a costeffective system which solves the problems of colorless ONU, bi-directional transmission, and long-reach simultaneously. However, there are difficulties in interfacing with existing GPON network which would be overcome in the near future.

# Table of Contents

| Certificate                                 | i    |
|---|------|
| Declaration of Authorship                   | ii   |
| Acknowledgments                             | iii  |
| Abstract                                    | V    |
| List of Tables                              | xii  |
| List of Figures                             | xiii |
| List of Acronyms                            | xvii |
| List of Symbol                              | xxi  |
| 1. Introduction                             | 1    |
| 1.1 Introduction                            | 1    |
| 1.2 A Fiber- optic communication system     | 1    |
| 1.2.1 Optical Transmitter                   | 2    |
| 1.2.2 Communication channel                 | 2    |
| 1.2.3 Optical receiver                      | 3    |
| 1.3 Access Network                          | 3    |
| 1.4 Motivation and Objective                | 5    |
| 1.5 Organization of Thesis                  | 6    |
| 2. Passive Optical Networks: Overview       | 8    |
| 2.1 Introduction                            | 8    |
| 2.2 Evolution of Passive Optical Networks   | 9    |
| 2.2.1 Past Generation PONs (APON & BPON)    | 9    |
| 2.2.2 Current Generation PONs (EPON & GPON) | 11   |

| 2.2.3 Next Generation PONs (10G EPON, XG-PON1 & XG-PON2) 12 | 2 |
|---|---|
| 2.2.4 Future PONs (WDM - PON)                               | 4 |
| 2.3 Performance Requirements of PON17                       | 7 |
| 2.4 Impairments in optical fiber Transmission18             | 8 |
| 2.4.1 Linear Impairments                                    | 8 |
| 2.4.1.1 Attenuation   | 8 |
| 2.4.1.2 Chromatic Dispersion                                | 9 |
| 2.4.1.3 Polarization Mode Dispersion                        | 0 |
| 2.4.1.4 Amplified spontaneous emission noise                | 0 |
| 2.4.1.5 Linear crosstalk in WDM systems                     | 0 |
| 2.4.2 Non- Linear impairments                               | 1 |
| 2.4.2.1 Stimulated Raman Scattering (SRS)                   | 2 |
| 2.4.2.2 Stimulated Brillouin Scattering (SBS)               | 2 |
| 2.4.2.3 Self Phase Modulation (SPM)                         | 2 |
| 2.4.2.3 Cross Phase Modulation (XPM)                        | 3 |
| 2.4.2.5 Four Wave Mixing (FWM)                              | 4 |
| 2.5 Summery   | 4 |
| 3. Modulation Techniques for optical communication          | 5 |
| 3.1 Introduction  | 5 |
| 3.2 Line coding   | 5 |
| 3.3 Optical Modulation                                      | 8 |
| 3.3.1 NRZ- ON-OFF keying                                    | 0 |

| 3.3.2 RZ-OOK  | 32 |
|---|----|
| 3.3.3 Manchester Coding   | 33 |
| 3.3.4 Differential Phase shift keying                                   | 34 |
| 3.3.5 Multilevel signaling  | 37 |
| 3.3.5.1 Duobinary   | 37 |
| 3.3.5.2 Dicode  | 39 |
| 3.5 Summary   | 42 |
| 4. Investigation and Mitigation the various effects in WDM TDM Networks | 43 |
| 4.1 Introduction  | 43 |
| 4.2 Raman Crosstalk   | 43 |
| 4.2.1 Generation of partial response coding                             | 44 |
| 4.2.2 Simulation Setup and Results                                      | 45 |
| 4.2 Spectral Analysis   | 46 |
| 4.2.1 Modeling and Equations Analysis                                   | 48 |
| 4.2.2 Simulation Model  | 51 |
| 4.2.3 Results and Discussion  | 53 |
| 4.3 Analysis of Group Velocity Dispersion (GVD)                         | 55 |
| 4.3.1 Network architecture and simulation setup                         | 58 |
| 4.3.2 Results and Discussion  | 60 |
| 4.4 Summary   | 64 |
| 5. Design of High Capacity PON using Different Topology                 | 66 |
| 5.1 Introduction  | 66 |

| 5.2 2×40 Gbps WDM-PON Long Haul System  |
|---|
| 5.2.1 Simulation Setup  |
| 5.2.2 Results and Discussion  |
| 5.3 Ring based WDM PON Network72  |
| 5.3.1 Setup of dual ring WDM-PON75  |
| 5.3.2 Results and Discussion  |
| 5.4 Design Hybrid WDM PON77   |
| 5.4.1 Hybrid WDM-PON Setup  |
| 5.4.2 Results and Discussion  |
| 5.5 Summary   |
| 6. Enhancement of Links Performance by Incorporating EDFA and Dicode coding. 82 |
| 6.1 Introduction  |
| 6.2 Importance of flexible PON  |
| 6.3 Hybrid WDM PON with Dicode Coded None Return to Zero (NRZ)                  |
| 6.3.1 Flexible Hybrid PON Setup:  |
| 6.3.2 Results and Discussion  |
| 6.3.3 Summery   |
| 7. Conclusion and Future Works95  |
| Bibliography97  |
| Publications  |
| Profile Summary   |

# List of Tables

| Table 2.1 Comparison of various PON technologies                     | . 16 |
|--|------|
| Table 4.1 Generation of Balance Duo-Binary Code                      | . 44 |
| Table 4.2 Generation of Dicode Code                                  | . 45 |
| Table 4.3 for Balance Duobinary Coding                               | . 46 |
| Table 4.4 for Dicode Coding  | . 46 |
| Table 4.5 BER performance with respect to GVD                        | 61   |
| Table 5.1 BER and Q Factor measurements for NRZ and Dicode at -10dBm | . 76 |
| Table 5.2 BER and Q Factor measurements for NRZ and Dicode at -4 dBm | . 77 |
| Table 5.3 BER and Q Factor measurement for Manchester coding         | . 79 |
| Table 5.4 BER and Q factor measurement of Dicode-coded NRZ           | . 79 |
| Table 6.1 Extinction ratio analysis                                  | . 88 |
| Table 6.2 Effects in the presence of thermal power density           | . 88 |
| Table 6.3 GVD Analysis at 10 Gbps data rate                          | . 89 |
| Table 6.4 Power Budget Analysis                                      | . 91 |

# List of Figures

| Fig. 1.1 Generalized fiber- optic communication system               | 2  |
|--|----|
| Fig. 1.2 Optical Fiber   | 3  |
| Fig. 1.3 Network Architecture  | 4  |
| Fig. 2.1 Basic PON Architecture                                      | 9  |
| Fig. 2.2 PON Evolution   | 10 |
| Fig. 2.3 GPON Network  | 12 |
| Fig. 2.4 XGPON frequency spectrum band as ITU-T G.987 recommendation | 13 |
| Fig. 2.5 NGPON 2 spectrum band                                       | 14 |
| Fig. 2.6 WDM-PON Architecture  | 15 |
| Fig. 2.7 (a) Out-band cross talk (b) in-band cross talk              |    |
| Fig. 2.8 SRS transfer of lower wavelength power to higher wavelength |    |
| Fig. 2.9 Self phase modulation                                       |    |
| Fig. 2.10 Cross phase modulation                                     |    |
| Fig. 2.11 FWM with two signal frequency                              | 24 |
| Fig. 3.1 Line coding to convert data to digital signal               |    |
| Fig. 3.2 Common line coding  |    |
| Fig. 3.3 Power spectral density of different line codes              |    |
| Fig. 3.4 ASK modulation  |    |
| Fig. 3.5 FSK modulation  |    |
| Fig. 3.6 PSK modulation  |    |
| Fig. 3.7 NRZ- OOK transmitter  | 30 |

| Fig. 3.8 Optical signal spectrum   |
|--|
| Fig. 3.9 Block diagram of a RZ transmitter   |
| Fig. 3.10 Power spectra of RZ-OOK  |
| Fig. 3.11 Binary to Manchester generation of waveform  |
| Fig. 3.12 Manchester optical spectrum  |
| Fig. 3.13 NRZ- DPSK (a) Encoder (b) Modulator  |
| Fig. 3.14 DPSK Receiver circuit  |
| Fig. 3.15 Power spectrum of DPSK   |
| Fig. 3.16 Duobinary Encoder  |
| Fig. 3.17 Generation of duobinary waveform   |
| Fig. 3.18 Power spectra of Duobinary after modulation  |
| Fig. 3.19 Dicode encoder   |
| Fig. 3.20 Power Spectra of (a) conventional NRZ (b) dicode-coded NRZ 40  |
| Fig. 3.21 Electrical waveform (a) NRZ signal of binary data (b) Output after Precoder (c) Encoded dicode signal  |
| Fig. 3.22 Dicode optical signal (a) Intensity (b) Phase (c) Power spectra42  |
| Fig. 4.1 A WDM TDM system with remote Raman pump 45  |
| Fig. 4.2 Simulation model of proposed distributed Raman amplified WDM-PON architecture. 52   |
| Fig. 4.3 Received power and eye diagram analysis with and without Raman pump 53  |
| <ul><li>Fig. 4.4 (a) Raman gain, (b) Optical Signal to noise Ratio, (c) Effective Raman gain</li><li>(d) spectral efficiency analysis with increasing fiber length</li></ul> |
| Fig. 4.5 Power spectra of dicode signal  |

| Fig. 4.6 Simulation setup of WDM TDM PON                         | 59 |
|--|----|
| Fig. 4.7 Electrical Dicode signal                                | 60 |
| Fig. 4.8 Downstream signal after multiplexer                     | 61 |
| Fig. 4.9 Optical received power                                  | 62 |
| Fig. 4.10 Fiber length vs Q factor                               | 62 |
| Fig. 4.11 Eye diagram for maximum length for 64 users for 82 KM  | 63 |
| Fig. 4.12 Fiber length vs Q factor for 128 user                  | 63 |
| Fig. 4.13 Eye diagram for maximum length for 128 users for 72 KM | 63 |
| Fig. 5.1 Simulation Setup of FBG Post compensation.              | 69 |
| Fig. 5.2 Block diagram of (a) OLT (b) ONU                        | 69 |
| Fig. 5.3 BER performance (a) 1550 nm & (b) 1551 nm               | 70 |
| Fig. 5.4 Q factor performance (a) 1550 nm & (b) 1551 nm          | 71 |
| Fig. 5.5 Eye pattern of Dicode-NRZ scheme with FBG for 2000 Km   | 71 |
| Fig. 5.6 PRBS dependency on dicode NRZ                           | 72 |
| Fig. 5.7 Ring architecture WDM-PON with fibre-fault protection   | 73 |
| Fig. 5.8 Power Spectra of dicode-coded NRZ                       | 74 |
| Fig. 5.9 Simulation Setup of dual ring WDM-PON                   | 75 |
| Fig. 5.10 BER performance of NRZ and Dicode-coded NRZ            | 77 |
| Fig. 5.11 Simulation Setup of HPON                               | 78 |
| Fig. 5.12 BER Performance of Manchester and Dicode coded NRZ     | 80 |
| Fig. 5.13 Eye diagram at 40 km for Dicode-coded NRZ              | 80 |
| Fig. 6.1 Flexible Hybrid PON                                     | 85 |

| Fig. 6.2 (a) A Flexible Hybrid PON System (b) Optical Network unit (ONU) | 86 |
|--|----|
| Fig. 6.3 Optical received power vs BER                                   | 90 |
| Fig. 6.4 Length v/s Q at 10 Gbps data rate                               | 92 |
| Fig. 6.5 Q factor analysis with length and number of user                | 92 |
| Fig. 6.6 Length vs maximum No. of User                                   | 93 |
| Fig. 6.7 Eye diagram for 64 users at 77 Km Dicode coding                 | 93 |
| Fig. 6.8 Eye diagram for 128 user at 67 KM for dicode                    | 94 |

# List of Acronyms

| AMI   | Alternate Mark Inversion                   |
|-------|--|
| AON   | Active Optical Network                     |
| APD   | Avalanche Photodiode                       |
| ASE   | Amplified Spontaneous Emission             |
| ASG   | Analog Signal Generator                    |
| ASK   | Amplitude Shift Keying                     |
| ATM   | Asynchronous Transfer Mode                 |
| AWG   | Array Waveguide Grating                    |
| B&S   | Broadcast and Select                       |
| BER   | Bit Error rate                             |
| BPON  | Broadband Passive Optical Network          |
| CAPEX | Capital Expenditure                        |
| CATV  | Cable TV                                   |
| CDMA  | Code Division Multiplexing                 |
| CL    | Correlative Level                          |
| СО    | Central Office                             |
| CPE   | Customer Premise Equipment                 |
| CSRZ  | Carrier Suppressed Return to Zero          |
| CW    | Continuous Wave                            |
| CWDM  | Coarse Wavelength Division Multiplexing    |
| DB    | Duobinary                                  |
| DC    | Direct current                             |
| DCF   | Dispersion Compensation Fiber              |
| DD    | Direct Detection                           |
| DML   | Directly Modulated Laser                   |
| DMT   | Dispersion Managements technique           |
| DPSK  | Differential Phase shift Key               |
| DQPSK | Differential Quadrature Phase Shift Keying |
| DS    | Downstream                                 |
| DSF   | Dispersion Shifted Fiber                   |
| DSL   | Digital Subscriber Line                    |

| DWDM  | Dense Wavelength Division Multiplexing                  |
|-------|---|
| EDFA  | Erbium Doped Fiber Amplifier                            |
| EOM   | Electro-Optic Modulator                                 |
| EPON  | Ethernet Passive Optical Network                        |
| ER    | Extinction Ratio  |
| FBG   | Fiber Bragg Grating                                     |
| FDM   | Frequency Division Multiplexing                         |
| FEC   | Forward Error Correction                                |
| FSAN  | Full Service Access Network                             |
| FSK   | Frequency Shift Keying                                  |
| FTTB  | Fiber to the Building                                   |
| FTTC  | Fiber to the Curb                                       |
| FTTH  | Fiber to the Home                                       |
| FTTX  | Fiber to the X  |
| FWM   | Four Wave Mixing  |
| GBPS  | Gigabit per Second                                      |
| GEM   | GPON Encapsulation Method                               |
| GEPON | Gigabit Ethernet Passive Optical Network                |
| GPON  | Gigabit Passive Optical Networks                        |
| GVD   | Group Velocity Dispersion                               |
| HDTV  | High Definition Television                              |
| HPON  | Hybrid PON  |
| IM    | Intensity Modulation                                    |
| IP    | Internet Protocol                                       |
| IPTV  | Internet Protocol Television                            |
| ITU   | International Telecommunication Union                   |
| LAN   | Local Area Network                                      |
| LASER | Light Amplification by Stimulated Emission of Radiation |
| LED   | Light Emitting Diode                                    |
| LPF   | Law Pass Filter   |
| MZDI  | Mach-Zehnder Delay Interferometer                       |
| MZI   | Mach- Zehnder Interferometer                            |
| MZM   | Mach- Zehnder Modulator                                 |

| NGPON | Next Generation Passive Optical Networks   |
|-------|--|
| NLSE  | Nonlinear Schrodinger Equation             |
| NRZ   | Non Return to Zero                         |
| NTSC  | National Television System Committee       |
| OADM  | Optical Add Drop Multiplexer               |
| ODN   | Optical Distribution Network               |
| OFC   | Optical Fiber Cable                        |
| OFDM  | Orthogonal Frequency Division Multiplexing |
| OLT   | Optical Line Terminal                      |
| OMI   | Optical Modulation Index                   |
| ONT   | Optical Networks Terminal                  |
| ONU   | Optical Network Units                      |
| OOK   | ON-OFF Keying                              |
| OSNR  | Optical Signal to Noise Ratio              |
| OSU   | Optical Subscriber Unit                    |
| P2MP  | Point to Multipoint                        |
| P2P   | Point to Point                             |
| PM    | Phase Modulation                           |
| PMD   | Polarization Mode Dispersion               |
| PolSK | Polarization Shift Keying                  |
| PON   | Passive Optical Network                    |
| PR    | Partial Response                           |
| PRBS  | Pseudo Random Binary Sequence              |
| PSD   | Power Spectral Density                     |
| PSK   | Phase Shift Keying                         |
| Q     | Quality Factor                             |
| QAM   | Quadrature Amplitude Modulation            |
| QPSK  | Quadrature Phase Shift Keying              |
| RB    | Rayleigh backscattering                    |
| RF    | Radio Frequency                            |
| RN    | Remote Node                                |
| RSOA  | Reflective Semiconductor Amplifier         |
| RZ    | Return to Zero                             |

| SBS  | Stimulated Brillion Scattering            |
|------|---|
| SCM  | Sub Carrier Multiplexing                  |
| SMF  | Single Mode Fiber                         |
| SNR  | Signal to Noise Ratio                     |
| SPM  | Self Phase Modulation                     |
| SRS  | Stimulated Raman Scattering               |
| SSMF | Standard Single Mode Fiber                |
| TDM  | Time Division Multiplexing                |
| TDMA | Time Division Multiple Access             |
| TWDM | Time and Wavelength Division Multiplexing |
| US   | Upstream                                  |
| VoD  | Video on Demand                           |
| VoIP | Video over Internet Protocol              |
| WAN  | Wide Area Network                         |
| WDM  | Wavelength Division Multiplexing          |
| WDMA | Wavelength division Multiple Access       |
| WS   | Wavelength Select                         |
| XPM  | Cross Phase Modulation                    |

# List of Symbol

| α                | Loss in optical fiber               |
|------------------|-------------------------------------|
| Dm               | Material dispersion                 |
| τ                | Pulse broadening                    |
| λ                | Wavelength of light                 |
| n                | Refractive index                    |
| c                | Velocity of light in vacuum         |
| $D_{wg}$         | Waveguide dispersion                |
| f                | Frequency of wave                   |
| R                | Data rate                           |
| ω                | Angular frequency                   |
| β                | Propagation Constant                |
| V                | Mode V number                       |
| $A_{\text{eff}}$ | Effective area of fiber             |
| Ge               | Raman gain                          |
| g <sub>R</sub>   | Raman gain coefficient              |
| $\gamma_R$       | Nonlinear strength                  |
| P <sub>p</sub>   | Pump power                          |
| Ps               | Signal power                        |
| Δf               | Channel spacing                     |
| γ                | Nonlinearity coefficient            |
| А                | Slowly varying electric field       |
| $\beta_2$        | Group velocity dispersion parameter |
| $T_0$            | Pulse width                         |
| L <sub>D</sub>   | Dispersion length                   |
| P <sub>0</sub>   | Peak power of input pulse           |
| Т                | Absolute temperature                |
| Ω                | Raman shift                         |

Dedicated to my family members for extending their unconditional support, strength and love to accomplish the task.

## Introduction

### **1.1 Preface**

Communication is defined as a process involving the exchange of information from one point to another. In the ancient time, a piece of information was conveyed from smoke and fire signal, reflecting mirrors and more recently signaling lamp [1]. The use of low capacity communication links invented in the early part of the twentieth century was limited due to the line of sight and atmospheric disturbance [2-3]. Low-frequency electromagnetic wave is considered suitable for transmission up to the considerable distance, but information and bandwidth varied as per their carrier frequency. It may be noted that the optical communication is the only available choice for potential large bandwidth which is 10<sup>4</sup> times larger than the microwave frequency. The system which uses the light carrier for the communication is known as light wave system. Fiber- optic system is the example of the light wave system, in which the carrier frequencies are selected from the optical range (particularly the infrared part) of the electromagnetic spectrum [4].

### 1.2 A Fiber- optic communication system

Fiber optics is an essential unit in the telecommunication infrastructure. Characteristics such as high bandwidth capabilities, low attenuation make it suitable for gigabit or higher transmission rates [5]. With the recent developments, fiber optics offers a lossless medium for transmission of information over long distances using light as the carrier wave. Conventional transmission using copper wire is electrical in nature while fiber optics is not the same. A standard fiber optics system mainly consists of three parts transmitter, optical fiber cable as a channel and receiver. The electric signal is converted into a light signal at the transmitter end which is carried by optical fiber cable and finally received at the receiver [1]. The receiver converts the received light back into an electric signal. The generalized fiber optical transmitter, communication channel and an optical receiver.



Fig. 1.1 Generalized fiber- optic communication system [5].

The original information input which can be in any physical form, e.g., voice, video, or data is converted to electrical input signal using suitable transducer at the transmitter [1].

#### **1.2.1 Optical Transmitter**

The optical transmitter consists of the optical source, modulator and channel coupler. The light source such as a semiconductor laser or Light-emitting diode (LED) is compatible with optical communication channel, hence used to generate the light pulse [3]. The light pulse could be modulated by two-ways. First, directly modulated laser, by changing the injection current and another is external modulator [5]. The modulation by changing the injection current is economical and simple but it can produce chirp especially in the high data rate signal. Now, external modulator is used to modulate high data rate signal so that chirp can be avoided. Modulated light signal focuses to optical fiber efficiently by channel coupler that uses the micro-lens. The launch power is a critical parameter, by increasing it, repeaters and amplifier spacing can be increased and it is limited by the nonlinear effects. As per usual standards, dBm is used as a unit for launched power with reference level as 1mW [5].

#### **1.2.2** Communication channel

Most of the light wave systems use the optical fiber as a communication channel that is used to transport optical signal from the transmitter to receiver with less distortion and other possible transmission losses. Usually, it is made by silica material that has property of less attenuation in useful bands. It shows 0.2 dB/Km at 1550 nm band hence it is very useful as a communication channel for lightwave system [5]. The basic

arrangement of core clad as an optical fiber is shown in Fig. 1.2. The optical fiber has cylindrical glass rod known as the core, which is used to propagate light.



Fig. 1.2 Optical Fiber; reproduced from ref. [4].

This core is surrounded by the cylindrical coaxial shell of glass, known as cladding, which has less refractive index [4]. The buffer coating is used to protect the fiber from damage and moisture [1].

#### 1.2.3 Optical receiver

An optical receiver converts the optical signal into the original electrical signal. It consists of a coupler, a photodetector, and a demodulator. The coupler focuses the received optical signal at the photodetector, which converts it into the electrical signal [5]. Design and type of demodulator depends on the format which is used in lightwave system. Frequency shift key (FSK) and phase shift key (PSK) format generally uses the heterodyne and homodyne demodulation techniques. Most of the light wave systems use the Intensity Modulation (IM) format which is known as amplitude shift keying (ASK). As per the received modulated signal, the decision circuits are used to detect the 1 and 0 at the receiver end. The signal to noise ratio (SNR) of the signal generated by the photodetector at receiver is the critical parameter defining the accuracy of decision circuits [5].

### **1.3 Access Network**

Access networks are responsible for connecting the service providers to their customers (either business and or residential). Most common local access networks such as the telephone networks are still based on the twisted- pair technology and coaxial cable [6-8]. The bandwidth required by customers has been increasing quickly in the last few years and twisted- pair / coaxial cable technologies are not able to realize such high

speed requirements [9]. However the demand for bandwidth is increasing exponentially due to advanced internet application and standard high definition Television (HDTV). To solve these problems, fiber was taken close to the residential and small business houses. There are several ways to deploy the optical network, which is commonly referred as Fibre to the X (FTTX)[10-14]. FTTX network has an arrangement of twisted pair, coaxial cable and optical fiber as shown in Fig. 1.3 and these types of arrangments and upgrading of network showed the bottleneck in term of data rate and bandwidth. To overcome this problem, renewal and reconfiguration of access network is required so that bandwidth could be freed up. In the contemporary scenario, many players are letting go off their traditional copper networks, paving the way for the optical fiber network. The optical fiber is coming closer to the end users, giving a faster connection. Services like fiber to the home (FTTH) emerge as clear choice for the long term because of its capability to accommodate the future bandwidth on the same hardware thus future proofing popular services like video on demand (VoD), online gaming, HDTV and video over internet protocol (VoIP) [15].



Fig. 1.3 Network Architecture; reproduced from ref. [17].

The foundation of fiber to the home lies on 100% deployment of optical fiber, from center office (CO) to subscriber end in the access network. It has mainly two types of configuration. First, it provides a dedicated connection to each user by individual fiber and this kind of topology known as Point to Point (P2P) connection. This type of network has high cost and is not able to use the available bandwidth of the single fiber

effectively. In second type of single network, fiber is shared by the multiple users varying from 16 to 32 by splitting the signal by power splitter. These types of network are known as passive optical networks (PON) [17]. Each of these networks has advantages and disadvantages involving financial, bandwidth and component parameters. The both architectures discussed above are the solutions to be deployed as FTTH for a major operator. The PON is mainly recognized for access networks due to the economic and future-safe solution and overcome the problem of bandwidth bottleneck [18].

### **1.4 Motivation and Objective**

With increase in continuous bandwidth demand for supporting high-speed applications such as Video on demand, High-definition television (HDTV), video conferencing, etc., the broadband access networks are the most promising solutions [19-22]. The researchers are focusing more on the design and implementation of innovative access network architectures, sustaining huge data at high speed with larger bandwidth for expected future demands [23-24]. PONs are attractive as they provide a profitable, highly flexible and scalable system for all end users [25-30]. Optical wavelength division multiplexing networks offer tremendous promise in meeting this demand. While going through several research article, got familiar with the challenges in the optical network like the bidirectional transmission in optical fiber, burst mode transmitter-receiver, colorless ONU, reach, network flexibility, integration to the legacy of PON, energy saving and cost of the networks [31-35]. Over the past decade, the scientific community had suggested various techniques to handle these challenges [36-40].

In the initial phase of fiber optic communication, the modulation techniques used was mainly intensity modulation (IM) [35, 41]. The techniques were very simple to implement but suffered from dispersion, crosstalk and nonlinear effects along the length of the channel. Recently, it has been realized that alternative forms of modulation format should be utilized to improve system capacity so that it can handle the ever-increasing bandwidth demand with low cost. Performance evaluation and enhancement of Passive Optical Networks may be improved by designing new architectures and incorporating suitable modulation formats (Line coding) [42]. In such system, the performance is a function of characteristics of the modulation schemes employed. The

numerous significant applications and developments inspired us to study the different line coding schemes [42].

Despite the merits like less bandwidth requirements, high noise immunity, low crosstalk, etc. of dicode code, the application of this coding has not been adequately addressed in high-speed optical communication [43-48]. An objective of proposed research work is to design and analyze the Dicode code in optical domain in high-speed optical communication. Simultaneously, various system level challenges and limitations of existing networks are considered during the design. Major objectives of this thesis are:

- Investigation and study of Dicode coded non return to zero (NRZ) coding including the mechanism for Dicode signal generation.
- > Investigate the effects of Dicode coding in PON Architectures.
- To keep simplified transmitter and receiver configuration for low cost and long reach transmission

This thesis work aims at system level innovation for the next generation optical access network, with main focus on modulation format and its corresponding transmitter and receiver to fulfill the system requirement. There are challenges in optical networks such as colorless ONUs, burst mode transmission, bi-directional transmission, long-reach PON, and cost. Next generation optical access network requires joint consideration of all the challenges mentioned above. Our research work aims at solving one or multiple research problems with innovation in signal modulation format, transmitter, and receiver techniques.

### **1.5 Organization of Thesis**

Chapter 2 reviews the basics of optical communication and passive optical networks as well as differentiate between past, current and next Generation PON (10G PON and WDM-PON) in terms of standardization, maturity, cost and power consumption. It describes the main requirement to improve the performance of PONs and various channel impairments like dispersion, attenuation, nonlinearity, etc.

Chapter 3 explains different line coding and their characteristics that would match with channel frequency response and make them suitable for the different application and developments. The conversion of the electrical signal into an optical signal by the suitable modulation format are studied which impetus the use of low loss transmission window of optical fiber effectively.

Chapter 4 describes effects of dispersion on the WDM-TDM networks utilizing NRZ and dicode. How this will affect system bit error rate (BER) and Quality factor is also discussed. It will also analyze the effect of Raman crosstalk in the communication channel using the multilevel coding like Duobinary and dicode.

Chapter 5 depicts and summarizes the design of high bandwidth WDM-PON and ring WDM-PON systems. Ring networks upstream and downstream signal will travel in the same direction. Hence it minimizes the occurrence of Rayleigh backscattering noise. In this chapter, implementation of long haul network using dispersion compensation with FBG is designed.

Chapter 6 study the implementation of a cost-effective system which can fulfill future bandwidth demand as well as flexibility. A flexible hybrid PON system is analyzed in the presence of dicode coding.

Chapter 7 concludes the results and observations made in this work and summarizes possible solutions. Some future trends and possibility of enhancement of PONs working and application areas also presented, that may encourage the future research community to work upon.

Chapter 2

## **Passive Optical Networks: Overview**

### **2.1 Introduction**

In this chapter, details of passive optical networks and their evolution with time is presented. As bandwidth demand increases, new access networks have been developing to support and fulfill this growth. This chapter also explains important parameters such as flexibility, scalability, etc. which are necessary to improve the system performance. Different impairments in the channel are also studied.

In today's world, the high-speed internet has become as essential part of life. This development is driving research and innovation for possible upgradation of access technologies to meet true broadband requirements [50-53]. Thus the market itself is moving in the direction of rolling out high volume optical fiber-based access networks by major operators [54-55].

At present, FTTH seems the only choice for long term that accommodates the future required bandwidth. Passive Optical Networks do fall in the category of one important architecture of FTTH [23],[56]. PONs have evolved to provide much higher bandwidth than previously deployed access networks such as digital subscriber line (DSL) and cable television (CATV) [57-59]. In a PON, a single fiber between the optical line terminal (OLT) and the remote node (RN) is shared by all users, which are connected to it. The transport path of the network is passive because all the information traveling from OLT to ONU is in optical form but the components used between both units are passive, which means no power is required from outside nor electrical devices are used.

A PON is a fiber network which utilizes fiber and passive components like splitter and combiner, instead of the active components like shaping circuits, repeaters, and amplifiers [18]. Networks based on PON technologies are cost effective, but their application is limited by the shorter range of coverage due to weak signal strength. Optical network with active component (active optical network: AON) is suitable for upto 100 km transmission as compared to passive optical network, which is typically limited to 20 km. A standard passive optical network is a point to multipoint network

where multiple subscribers of size 16 to 128 receive the internet or television services from central optical line terminal. The signal is split into equal low- power signals to distribute it to the multiple users. An ONU terminates the PON at the customer's home. Usually, ONU communicates with ONT and therefore can be considered as a single unit also [24].



Fig. 2.1 Basic PON architecture,

The point-to-multipoint (P2MP) structure of a PON is depicted in Fig. 2.1. The optical line termination (OLT) distributes data in the downstream direction toward the optical network units (ONUs), located at the subscriber's premises. The subscriber uses time-division-multiple-access (TDMA) scheme to send upstream data toward the OLT [60]. There is continuous development in the passive optical networks as per requirements of the customers from time to time.

#### 2.2 Evolution of Passive Optical Networks

In past three decades, various types of PONs have evolved and are classified as past (APON & BPON), current (EPON & GPON), next (10G EPON, XG-PON1 & XG-PON2) and future generation of PONs (WDM-PON).

#### 2.2.1 Past Generation PONs (APON & BPON)

In the mid -1990s, an initiative of the full-service access networks (FSAN) group developed the first PON system referred as Asynchronous Transfer Mode Passive Optical Networks (APONs) [61]. The electrical layer is built on Asynchronous Transfer Mode (ATM) for commercial availability. How to extend the high-speed services with speed? British Telecom in 1995 addressed this topic while organizing FSAN coalition which aimed at developing the most economical and efficient way to perpetuate the emerging market services like Internet Protocol (IP), 10/100 Ethernet, video etc to business and residential customers globally. ATM and PON appeared as two most logical choices for the protocol and physical plant. Multiple protocols were best served by ATM while PON was the most inexpensive broadband solution catered optically. FSAN used APON format which was later accepted as an International Telecommunications Union (ITU) standard (ITU-T Rec. G.983.x series) [62-63].



Fig. 2.2 PON evolution; reproduced from ref. [64].

ITU-T updates and recommendations on G.983 series started in 1998 dealing with the BPON based broadband optical access system [64]. The last mile of communication infrastructure is addressed by PONs between the service provider's Center office (CO), head end, or point of presence (POP) and business or residential customer locations. The APON format previously developed soon became a basis for international standard released by ITU-TS (Rec. G.983.x) also called as BPON (broadband PON), and it is enhanced subsequence of APON, in which the downstream and upstream data rate is 622 Mbps and 155 Mbps respectively [65-67]. It added some feature like dynamic bandwidth distribution, protection, and other feature. However, it was realized with early deployment that BPON ODN could not upgrade to the next upgraded technologies in the future. The evolution of PON is shown in Fig. 2.2.

#### 2.2.2 Current Generation PONs (EPON & GPON)

Two visionary startups perceived the weakness in APON standard viz its lack of video capabilities, low bandwidth, complexity, and cost. With the market moving towards fast Ethernet, gigabit Ethernet, and 10-gigabit Ethernet, these startups soon realized that the need for conversion in the wide area network (WAN)/ local area networks (LAN) connection between ATM and IP protocols would be eliminated by Ethernet passive optical networks (EPONs) which uses similar optical frequencies as TDMA and Giga bit passive optical network (GPON). In fact, (EPON) 802.3ah specifies comparable passive optical network such as those based on Ethernet standard 802.3 with a range up to 20 km [68]. In such networks, the data rate as high as 1.25 Gbps could be obtained in both uplink and downlink.

A frequency band 1550 nm reserved for future extensions and video services, a 1490 nm based bidirectional 1 Gbps link is provided by EPON [69]. Owing to its compatibility with other Ethernet standards, handshake becomes easy and avoids any conversions or encapsulations. Ethernet being the primary networking technology used in local area networks and MAN (Metro area networks), payload up to 1518 bytes is efficiently served by the same Ethernet frame. A PON system which supports the data rate of 2.5 Gbps downstream and 1.25 Gbps in upstream is recommended by ITU-T G 984 series with the specified protocol is known as Gigabit Passive Optical Networks [70]. EPON is based upon Institute of Electrical and Electronics Engineers (IEEE) 802.3ah and GPON on ITU-T G984.x standard.

GPON is the most widely used PON architecture that uses single fiber for both downstream and upstream transmission by utilizing wavelength division multiplexing [71] with operational wavelength ranges viz. downstream: 1480-1500 nm, upstream: 1260-1360 nm and downstream RF video distribution: 1550-1560 nm. A 1490nm laser wavelength transmits over the channel for downstream while the wavelength of 1310 nm transmits as upstream data as shown in Fig. 2.3. At remote end, each user is allotted a time slot specific to them using time division multiple access (TDMA) format by GPON, which not only allows each ONU to get a downstream data



Fig. 2.3 GPON network; reproduced from ref. [71].

rate of 2.488 Gbps but also divides the bandwidth to ensure that each user gets a fraction of 100 Mbps speed depending on allocation by the service provider [71-72]. However, upstream rate is less than the maximum due to its sharing with other ONUs in a TDMA scheme. OLT determines the time and distance delay for each subscriber and paves the way of allotting upstream time slots to each user. Each fiber can serve up to 32 or 64 subscribers based upon its typical split of 1:32 or 1:64 respectively which extends to 1: 128 in some systems. The most practically deployed system use class B+ devices which have a length of 20 KM and split ratio of 1:32. It has a transmission data rate of 2488 Mbps and 1244 Mbps for downstream and upstream respectively, and this types of specification lead to a power budget of 28 dB [70].

#### 2.2.3 Next Generation PONs (10G EPON, XG-PON1 & XG-PON2)

In late 2006, FSAN group begin to work on a system that would proceed after GPON, however in 2007 the focus shifted to define new system itself. Different architecture such as TDM-PONs, WDM-PONs, and CDMA-PONs were investigated. There was a difficulty to compare these systems in a reasonable way because they were quite different in architecture and service profile. Due to high bandwidth demand required from consumers and business applications, the most crucial requirements for next generation PON (NGPON) is to provide higher bandwidth than GPON. The migration GPON to NGPON of the subscribers must be smooth, and NGPON must be able to protect the major investments spent on time and money [72-73]. Fig.2.4 gives the solution to the problem by dividing the system proposal into two groups. The first group
NGPON1 systems could coexist with GPON by sharing the same optical distributed network ODN. NGPON2 included all system that either required a different ODN or that required new technologies.

10 GEPON: - To increase the data rate of EPON from 1 Gbps to 10 Gbps, 10 GEPON was developed on IEEE P802.3av PON by keeping ethernet interface [73]. 10 GEPON with EPON shared many protocols and coexisted with EPON and operated at the same ODN. It operates at a downstream transmission rate of 10 Gbps and upstream data rate 2.5 Gbps. For downstream two wavelength band (1480-1500 nm and 1575-1580 nm) are used for 1 Gbps and 10 Gbps channels respectively. In the upstream direction, two bands are overlapped [73-74].

XGPON1: - XGPON (X stands for10) is one of the most promising solutions for NGPON1. A mechanism of migration for acquiring a signal for 10 Gbps for downstream and 2.5 for upstream transmission is standardized in ITU-T Rec. G987.x for XGPON technology. NGPON2 technology is still evolving with growing number of technological suppliers and interested operators [75]. At 10 Gbps data rate, a high-speed burst mode trans receiver and dispersion are the most significant challenges for the realization of XGPON. XGPON is better as compared to WDM-PON regarding capacity and delay. XGPON2 has a symmetric data rate of 10 Gbps for downstream and 10 Gbps for upstream transmission and provides high capacity for upstream signal [76-77].



Fig. 2.4 XGPON frequency spectrum band as ITU-T G.987 recommendations; reproduced from ref. [78].

NG-PON 2: The basic requirement of NGPON2 is to support the 40 Gbps downstream for the residential and commercial application. FSAN group considered many option for NGPON2 which supports 40 Gbps aggregate capacities like WDM-PON, UDWDM, OFDM-PON, hybrid system and TWDM [78]. TWDM-PON technology was chosen by FSAN to be the prominent solution for NG-PON2 as from the operators' perspective it is more reliable, less disruptive and cost effective than the other solutions, which provides a significant advancement in the NG-PON2 standardization [79]. This type of advanced system can increase PON bandwidth capacity by a minimum 40Gbit/s and can provide 1Gbit/s or more services that may be rolled out in the near future [78].

As we study the coexistence of the various PON architectures, the planning of wavelength band with which NG-PON2 technology has to coexist must be taken into account. Fig. 2.5 gives the spectrum for NG-PON2. Here we consider Four wavelengths of 10 Gbit/s which are combined by the multiplexer in the Central Office and transmitted as a downstream signal [78]. At the remote end, ONU chooses the required wavelength by selecting one of the wavelengths from downstream signal. For the upstream, the ONU/ONT operate in one of the four upstream wavelengths, which was chosen by the OLT receiver. By placing the WDM filter at the center office it can coexist with the legacy of PON, but ODN is left untouched in all possible NG-PON 1 solutions.



Fig. 2.5 NGPON 2 spectrum band; reproduced from ref. [78].

Several studies revealed that TWDM PON [80] is the most lucrative solution for fulfilling the requirements. ITU-T G.989.1 recommended in March 2013, the possible wavelength bands as per requirement of operators.

### 2.2.4 Future PONs (WDM-PON)

Wavelength division multiplexing passive optical network (WDM-PON) is known as a promising technology for future networks to provide high data rates to the end users. The Optical Add/Drop Multiplexer (OADM) improves the performance of a WDM- PON Network architecture, which supports high dense geographical area with unlimited scaling [81-82]. Higher installation and maintenance costs can be reduced by the types of WDM architecture as shown in Fig. 2.6. The network design can support 10 Gbps downstream and upstream for distance up to 30 km. WDM-PON gives an alternative to the GPON scheme by providing each ONT transmission and reception at a specific wavelength [82].

As WDM-PON standards are still under development, manufacturers and operators anticipate that it will be using the same physical network as used by PON systems, having an OLT serving of 32 ONTs. Each ONT will be provided a different wavelength, which enables each customer to get higher bandwidth. WDM-PON expands on the idea of increasing the number of wavelengths in the fiber from the OLT to a nearby node [83-86]. Therefore, the key difference between WDM-PON and GPON is that WDM-PON may not use the GPON protocol but can use point-to-point communication Gigabit Ethernet.



Fig. 2.6 WDM-PON Architecture; reproduced from ref. [82].

In the near future a high bandwidth will be required by the end users for high definition TV (HDTV). Daily activities that are now in use also consume a significant amount of bandwidth to support data voice and video services [83]. The basic difference among the various PON technologies and generation is summarized in Table 2.1 given below.

|                              | A/BPON              | EPON, 10<br>EPON  | GPON                 | XGPON1/<br>XGPON2                    | WDM-<br>PON                                   |
|------------------------------|---------------------|---|----------------------|--------------------------------------|---|
| Standard                     | ITUG<br>983/FSAN    | IEEE 802.3-<br>ah (1Gbps)<br>IEEE 802.3-<br>av (10<br>Gbps) | ITU-T<br>G-984       | ITU-<br>TG987/FS<br>AN               | ITU-T G-<br>983                               |
| Protocol                     | ATM                 | Ethernet  | ATM,<br>Ethernet     | GEM                                  | Protocol<br>independen<br>t                   |
| Subscriber<br>cost           | Low                 | Only 10%<br>of GPON<br>subscriber<br>cost                   | Moderat<br>e         | High but<br>less than<br>WDM-<br>PON | Very high                                     |
| Managemen<br>t system        | Single layer        | Single layer  | Three<br>layer       |                                      | Dual<br>manageme<br>nt<br>mechanism           |
| Multiplexin<br>g             | TDM                 | TDM   | TDM                  | TDM                                  | FDM   |
| Downstrea<br>m bitrate       | Maximum<br>622 Mbps | 1.25 Gbps,<br>10.3 Gbps                                     | 1.2, 2.5<br>Gbps     | 10 Gbps                              | 1-10<br>Gbps/chan<br>nel                      |
| Upstream<br>bitrate          | 155/622<br>Mbps     | 1.25 Gbps,<br>1.25 or 10.3<br>Gbps                          | 1.2, 2.5<br>Gbps     | 2.5 Gbps,<br>10<br>Gbps              | 1-10<br>Gbps/chan<br>nel                      |
| Downstrea<br>m<br>wavelength | 1490 nm,<br>1550 nm | 1490, 1550<br>nm  | 1490 nm              | 1577 nm                              | Separate<br>wavelength<br>for each<br>channel |
| Upstream<br>wavelength       | 1310 nm             | 1310 nm   | 1310 nm              | 1270 nm                              | Separate<br>wavelength<br>for each<br>channel |
| ONUs                         | 32                  | 32 and 64<br>(with FEC)                                     | 128                  | 128                                  | 16-32   |
| Per user<br>BW               | 20-40 Mbps          | 30-60<br>Mbps, >100<br>Mbps                                 | 40-80<br>Mbps        | >100<br>Mbps                         | 1-10 Gbps                                     |
| Reach                        | 20 Km               | $\leq$ 20 Km  | $\leq 60 \text{ Km}$ | $\leq$ 20 Km                         | $\leq$ 20 Km                                  |

Table 2.1 Comparison of various PON technologies [62-83].

The main advantages of WDM-PON [82] are summarized below :-

- It provides separate bandwidth to users by using the different wavelength i.e. no bandwidth scheduling is needed.
- Transmission bandwidth of the fiber is used effectively, up to 64 subscribers/fiber.
- A low loss AWG is used in contrast to the high-loss power splitter which ensures longer reach.
- Physical separation of subscriber signals.

The cost of the WDM-PON is increased, as a separate wavelength requires individual transmitters. At the user end, it will effect severely. However, at the center office side, the cost can be lowered by optical integration.

# 2.3 Performance Requirements of PON

**Compatibility and Flexibility**: - Designed system should be integrated with legacy PON network and should be adaptable to new upcoming technologies and protocols. Flexibility is related to the possibility of each user in the network, who can use any wavelength that is provided by the transmitter end that will help to support the random traffic pattern [87].

**Scalability:** - The network is said to be scalable if a particular user can access the variable bandwidths according to the immediate traffic demand and a growing amount of work [88].

**Resiliency:** - The network is called resilient when services are continued without any interruption in case any wavelength failure, this is achieved by diverting any available wavelength [89].

**Immunity to interference:** - The PON is immune to electromagnetic interference that is produced by same and different media [90].

**Optimization of Networks:** - The number of users can be increased by reducing the channel spacing, but it also increases interference due to overlapping of the signals in the channel [90,91]. So, there is a requirement to optimize amplifiers and other components to utilize maximum bandwidth and receiver capability.

**Improving the capacity of Networks:** - The existing WDM network with large spacing wavelengths can be converted into dense wavelength division multiplexing (DWDM) network by multiplexing these wavelengths with narrower spacing without disturbing the existing fiber [85].

## 2.4 Impairments in Optical Fiber Transmission

In this section, we will discuss various linear as well as nonlinear impairments which affect the quality of the signal such as reduction in amplitude, distortion as well as contamination.

### 2.4.1 Linear Impairments

The optical fiber offers huge transmission bandwidth. The application of optical fiber system for longer distance and the higher bit rate suffers from linear effects of the fiber such as attenuation and dispersion imposing limiting factor. In order to have high performance, high bit rate, long distance point to point optical transmission networks, both linear and nonlinear effects should be managed well. The present chapter gives the brief overview of various fiber induced impairments and their adverse effects on the performance of the networks.

### 2.4.1.1 Attenuation

Attenuation in an optical fiber plays very crucial role in designing a light wave system. This causes the decay of signal strength and loss of light power over fiber. Scattering, absorption, and bending loss are the main source of attenuation in the optical fiber [5]. The other reasons that contribute to attenuation are extrinsic, such as stress during the manufacturing process, environmental effects as well as physical bending [4-5].

When light travels in fiber over a length then power at the origin of fiber is P(0) and at the distance z is P(z) that can be related as given in Equation (2.1) [5]

$$P(z) = P(0)e^{-\alpha Z}$$
(2.1)

Then attenuation loss as Equation (2.2)

$$\alpha(dB/Km) = \frac{10}{z} \log\left(\frac{P(0)}{P(z)}\right)$$
(2.2)

The loss in optical fiber is expressed in dB/Km as shown in Equation (2.2). For silica material at 1550 nm (C-band) and 1310 nm (O-band), the possible losses are 0.25 dB/km and 0.4 dB/km respectively.

#### 2.4.1.2 Chromatic Dispersion

A light pulse which has a fixed spectral width represents data and information. We consider a laser light as a monochromatic, but in practice it has some different wavelengths. Hence when this light pulse travels through the fiber, then each different component of light will travel at a different velocity at fiber end, and all wavelengths will reach at the various time hence pulse become broadened. This phenomenon is known as chromatic dispersion. If this is not managed properly, then light pulse gets overlapped resulting in inter-symbol interference in an adjacent bit, and at the receiver end, error free signal detection is not possible [92].

The main causes of chromatic dispersion or intramodal dispersion (also known as Group velocity dispersion: GVD) are material dispersion and waveguide dispersion. Material dispersion arises due to intrinsic properties of glass such as the variation in refractive index of the core. Refractive index is the function of wavelength hence pulse spreading occurs even though different wavelengths follow the same path. Material dispersion in term of material dispersion parameter given by [5]

$$Dm = \frac{1}{L} \frac{\tau}{\Delta \lambda} = \frac{\lambda}{c} \left| \frac{d^2 n}{d \lambda^2} \right|$$
(2.3)

L is the fiber length,  $\Delta\lambda$  is the full width at half maximum at peak spectral power of the source at  $\lambda$ , c is the velocity of light in vacuum, n is a refractive index, and  $\tau$  is pulse broadening of pulse. Waveguide dispersion arises mainly due to the bound structure of the fiber, where degree of waveguide dispersion depends on the fiber design. The waveguide dispersion due to nonlinear b-V diagram is obtained as [5]

$$D_{wg} = -\frac{n_2 \Delta}{c\lambda} V \frac{d^2(bV)}{dV^2}$$
(2.4)

Generally, V and b is the substitution of  $\omega$  and  $\beta$  respectively in the two normalized parameter of fiber. The total chromatic dispersion is given by Equation (2.5)

$$D = D_{mat} + D_{wg} \tag{2.5}$$

### 2.4.1.3 Polarization Mode Dispersion

When we move towards high data rate (more than 10 Gbps) light wave system to fulfill the future bandwidth demand, then polarization mode dispersion (PMD) limit the reach of the system. In practical optical fiber, the core is not completely cylindrical hence there is non-uniformity in core diameter along the fiber. This leads to differences in propagation constant of two polarization components and fiber become birefringence. Light pulse travelling through single mode fiber, has two orthogonally polarized components with two different velocity  $V_{gx}$  and  $V_{gy}$  arriving at the end of the fiber at different times [5].

$$\Delta T = \left| \frac{L}{V_{gx}} - \frac{L}{V_{gy}} \right| \tag{2.6}$$

This above Equation (2.6) is valid for the only assumption of constant birefringence. For conventional single mode, fiber PMD induce pulse broadening can be given by

$$\Delta T = D_{PMD}\sqrt{L} \tag{2.7}$$

Where  $D_{PMD}$  unit is ps/km<sup>1/2</sup>, value varies from 0.01 to 10 ps/km<sup>1/2</sup>, L is the length of the fiber, and  $\Delta T$  is a time delay. This difference in propagation time leads to pulse broadening, and degrade the transmission performance through Inter-symbol interference [5].

### 2.4.1.4 Amplified spontaneous emission noise

For long haul network, signal travelling through fiber gets degraded due to loss. This loss can be compensated using erbium doped fiber amplifier (EDFA). During amplification amplified spontaneous emission (ASE) noise is generated by the spontaneous decay of electrons in the higher energy levels to lower energy levels. Hence the wide frequency range of photon emitted [93].

### 2.4.1.5 Linear crosstalk in WDM systems

As advances in optical networks leads towards use of transmission bandwidth of fiber more effectively, more than two wavelengths are used and multiplexed into the optical fiber. As spacing of the wavelength is reduced for dense wave division multiplexing, adjacent channel cross talk is very common for a communication system. This crosstalk is very significant which affects the minimum separation of two channel, and there is a partial overlap of the channel in the frequency domain. Linear crosstalk in WDM-PON mainly arises due to channel spacing, filtering characteristics of the multiplexer as well as de-multiplexer [94]. Linear crosstalk is classified as in-band and out –band as per their passband of an optical fiber [95]. The optical out band is defined in WDM for different wavelength hence this is not much harmful due to easy separation by the narrow band optical filter as shown in Fig. 2.7(a). In the in-band crosstalk, the desired signal and leaked signal have nominal identical wavelength shown in Fig 2.7 (b).



Fig. 2.7 (a) Out-band cross talk (b) in-band cross talk; reproduced from ref. [95].

### 2.4.2 Non- Linear impairments

A high bit rate system operated at high transmitter power, suffers from several nonlinear effects. These effects are the results of the intensity dependence of refractive index of the medium. In WDM-PON, multiple wavelength multiplex into the single optical fiber due to which transmitter power and bit rate start to play role for the nonlinearity. There are nonlinear phenomena which occur due to intensity dependence of refractive index known as Kerr effects, and another one is inelastic scattering phenomena. Kerr effect includes self-phase modulation (SPM), cross phase modulation (XPM) and Four-wave mixing (FWM). Stimulated Raman scattering (SRS) and stimulated Brillouin scattering (SBS) is the part of inelastic scattering [5]. As the fiber length increases, the nonlinear effects start to decay due to power loss in the fiber hence nonlinear phenomena are more dominant at the start of the fiber. The nonlinearity will contribute towards the signal impairment in the optical communication system, and additional power is required at the transmitter to maintain the same BER.

### 2.4.2.1 Stimulated Raman Scattering (SRS)

SRS is a phenomenon that occurs when two and more wavelength signals travel into the optical fiber, lower wavelength channel power is transferred to higher wavelength channel [96]. It is an interaction between light waves and molecular vibration of silica. In this interaction, the scattered light is generated which is downshifted by Stokes frequency. If any signal is present at higher wavelength, then it will experience gain means it gets amplified and this gain process is known as SRS as shown in Fig. 2.8. In WDM system where multiple channels are equally spaced and easily find the separation of 110 nm between two wavelengths, which is appropriate for Raman gain, then low wavelength channel is treated as a pump for higher wavelength channel [96-97].



Fig. 2.8 SRS transfer of lower wavelength power to higher wavelength; reproduced from ref. [4].

### 2.4.2.2 Stimulated Brillouin Scattering (SBS)

It is a nonlinear process occurring in fiber due to the launching of high power optical signal. It includes the interaction of incident light with acoustic phonons in the silica glass and produces the variation in refractive index of fiber [98]. This variation in fiber works as a grating hence a particular frequency gets backscattered and down shifted by 11 GHz from the original signal [4]. The Brillouin line width is 20 MHz at 1550 nm and is very narrow compared to Raman. SBS effect is confined within a single wavelength channel and occurs in WDM in each channel for same power level.

### 2.4.2.3 Self Phase Modulation (SPM)

A single light wave's passage through the optical fiber can be affected by the nonlinearity. Its own optical field modifies the phase of the optical signal. This phenomenon is known as self-phase modulation. The edge of pulse shows the time varying intensity, which rises rapidly from zero to maximum value then return to zero.

Since the phase fluctuation is intensity dependent, different parts of pulse undergo a different phase shifts. This leads to frequency chirping [4]. For some fibers, the time varying phase may result in a power penalty owing to a group velocity dispersion (GVD) induced pulse broadening of the pulse as it travel along the fiber and resulting overlap between itself and the adjacent bit and increases the bit error rate of the system as shown in Fig.2.9.



Fig. 2.9 Self phase modulation; reproduced from ref. [100].

### 2.4.2.3 Cross Phase Modulation (XPM)

XPM is known as another cause of nonlinear effects which affects the WDM transmission performance. In this, signal phase is modulated according to the neighboring channel of DWDM system [101].



Fig. 2.10 Cross phase modulation; reproduced from ref. [102].

This happens due to refractive index change by the power of adjacent channel as well as central power channel in a nonlinear fashion. If XPM occurs in DWDM channel, the SPM will also be present and coupling the power among the DWDM channel as shown above Fig. 2.10. The dispersive walk-off phenomena will also exit due to the different frequency component of DWDM hence affecting the adjacent channel power [4] as shown in Fig. 2.10. Dispersive walk-off among the channels affects the power of the adjacent channel and reduces the effects of XPM [5].

### 2.4.2.5 Four Wave Mixing (FWM)

If three or more frequency signal using WDM travel through optical fiber, a new frequency signal is generated, this phenomenon is known as FWM. It is a third order nonlinearity [103]. Three frequency signal  $f_1$ ,  $f_2$ , and  $f_3$  induce phase shift in the neighboring channel by the intensity dependence of refractive index, together with this it will generate a new frequency as  $2f_i$ - $f_j$  and  $f_i$ - $f_j$ - $f_k$ . In general, two frequency  $f_1$ ,  $f_2$  generates new frequency due to FWM which has little power as compared to another original signal as shown in Fig. 2.11.



Fig. 2.11 FWM with two signal frequency; reproduced from ref. [4].

This critically depends on channel spacing and dispersion of DWDM system in contrast to SPM, XPM which are significant on high bit rate system [4-5].

### 2.5 Chapter Summery

In this chapter, most commonly used passive optical network standards such as APON, BPON, GPON EPON and NG PON are discussed. In view of requisite high bandwidth in futuristic network, a detailed study of WDM-PON is presented. Important parameters such as flexibility and scalability have also been studied. Various fiber impairments like dispersion, attenuation, SRS, SBS, SPM, XPM and FWM have been discussed in detail.

# **Modulation Techniques for optical communication**

### **3.1 Introduction**

In recent times, advancement in optical systems are in the direction to achieve high bit rate and to operate for long distance communication, however the performancedegradation due to impairments in the channel is posing the problems. The improvement in the system performance can be achieved by utilizing the appropriate modulation format. The modulation format finds the transmission quality and spectral efficiency for the optical system. Here we focus mainly on different digital line codes such as NRZ, RZ, Duobinary etc. There is immense scope of research in optical modulation formats. In this chapter, we also discuss various optical modulation formats such as NRZ-OOK, RZ-OOK, DPSK etc. including their important characteristics. The line codes generate the waveform from the binary data information, which is only suitable for transmission in the baseband channels. Modulation is necessary to transmit the information at higher frequency. The Different modulation formats are discussed in the following sections of the chapter.

## 3.2 Line coding

In optical communication, the line coding is techniques where binary digital data is converted to the real temporal digital waveform and transmitted in baseband channel as shown Fig. 3.1. The development in various line coding is required for the different application, channel characteristic as well as a performance requirement. Each line code has its distinct properties.



Fig. 3.1 Line coding to convert data to digital signal; reproduced from ref. [42].

Some essential characteristics of line coding are given below, which are useful for selecting the proper line coding according to the application [42].

**Number of signal levels:** Binary data can be represented by two or more level of digital signal. Digital signal, which has more than two levels, is also known as a multilevel signal. The spectral efficiency can be improved by the increasing the number of level of the signal.

**DC components:** After line coding, there is zero frequency signal in the spectrum, which is known as direct-current (DC) component. The DC component is undesirable because it cannot pass through some components of an optical communication system. This leads to distortion in the signal and unwanted energy loss of the system

**Signal spectrum:** The different line coding leads to different power spectrum and bandwidth of the signal. To avoid attenuation and distortion in the channel, the frequency response of the channel should be matched to signal spectrum. The energy should be contained in small bandwidth, smaller the bandwidth; the higher is the transmission efficiency.

**Synchronization:** To avoid error in received signal, the bit interval of receiver should be exactly same as a transmitter. If any mismatch occurs then signal gets an error and the original signal cannot be recovered. Usually clock signal is generated and synchronized with the received signal. If the signal has minimum one transition per bit interval, then it can be self-synchronized.

**Cost of implementation:** Cost is the hurdle for the practical application of the system so encoder should be kept as simple as possible.

The classification of line coding is done according to the level and polarity of signal [42]. Binary line coding is classified in two-ways (i) level codes (ii) and transition codes. In level codes, the information is in the voltage level as low and high for particular bit intervals like full bit interval or some part of the bit interval. Non- return to zero (NRZ) and return to zero (RZ) are the example of the full bit interval and partial bit interval respectively. The level codes are instantaneous, encode instantaneous binary data to signal waveform and does not depend on the previous binary number. Another types of line coding is transition codes which carry information in the change in the level of the waveform and also possess memory stage, so the generation of waveform

depends on the previous binary bit information [104]. Further, the line codes are classified according to the polarity of the signal such as unipolar, polar and bipolar as shown in Fig. 3.2.



Fig. 3.2 Common line coding; reproduced from ref. [42].



Fig. 3.3 Power spectral density of different line codes; reproduced from ref. [42].

For unipolar signaling, one of the polarity either positive or negative voltage is used to represent the data information. If both positive and negative level of voltages are used to represent information or not using voltage level zero, is known as polar signaling [104]. In bipolar signaling, both negative and positive voltage level along with zero voltage level are used to represent information. The term bipolar signaling is utilized by some researchers in past to designate a specific line coding scheme with positive,

negative, and zero voltage levels [105]. Some of the main line codes shown in the Fig. 3.2. the frequency response of the channel can be matched to power spectral density as dipicted in Fig. 3.3 as a requirement of the application. In the next section, we have explained the modulation techniques for different line codes such as NRZ, RZ, Manchester, and DPSK along with advanced coding like Duobinary, dicode.

### **3.3 Optical Modulation**

To transmit the information through an optical fiber, the line coding has to be transformed in the optical range at THz frequency band to match the frequency response of the channel [106]. The process of transformation is known as modulation [42]. The three physical features of optical field distribution which are considered to transmit information are intensity modulation, phase modulation (as well as frequency) and polarization modulation. The selection of appropriate modulation format is the essential key to build the flexible and cost-effective high-capacity optical networks.

**Intensity modulation (IM)**: - In intensity modulation, the amplitude of the source switches between two states ON and OFF, so it is also known as ON-OFF keying. The amplitude shift keying (ASK) is the type of intensity modulation [107]. The high-frequency optical carrier signal is modulated by the message signal as shown in Fig. 3.4.



Fig. 3.4 ASK modulation; reproduced from ref. [107].

The message signal is imposed on the carrier signal such that binary '1' is represents ON states with a fixed width amplitude and zero voltage level represents binary '0' or the OFF state.

**Frequency Modulation:** - The frequency of carieer signal is varied as per data signal information between two frequencies as shown in Fig. 3.5 known as frequency shift

keying (FSK) [107]. In this technique envelope of the optical signal does not change, so generation and receival of a signal are costly and circuit complexity of FSK is high on comparison to ASK.



Fig. 3.5 FSK modulation; reproduced from ref. [107].

**Phase modulation (PM):** - One of the property (phase) of the carrier signal is changed according to the digital information or binary data as shown in Fig. 3.6. This improves the nonlinear tolerance for the system due to the presence of constant signal envelope and narrow spectral. The phase shift keying (PSK) uses the finite number of phases to represent the unique pattern of binary digits.



Fig. 3.6 PSK modulation; reproduced from ref. [107].

Moreover, each pattern of bits forms the symbol which is represented by the particular phase. The effects produced by the multichannel are very sensitive to this type of modulation. Due to its complex receiver circuit, it did not receive much interest.

**Polarization Modulation:** - In the polarization modulation, two orthogonal polarizations exist which can be changed according to binary bit information like one for 1 bit and other for 0 bit. It is known as polarization shift keying. It also has constant signal envelope which improves the sensitivity and nonlinear tolerant uses the system bandwidth effectively [107]. It is very sensitive to any polarization disturbance in the

channel and receiver and detection of the signal is complex. In the next section, we study the intensity modulation with different line coding. Intensity modulation is very popular for the light wave system because the receiver is simple due to direct detection and less cost at the subscriber end.

#### 3.3.1 NRZ- ON-OFF keying

For a long time, non-return-to-zero on-off-keying (NRZ-OOK) has been the dominant modulation format in intensity modulation/direct detection (IM/DD) fiber-optical communication systems [42]. In NRZ line coding any positive voltage represents binary bit 1 and zero voltage represents bit 0 as shown in Fig.3.2. This is instantaneous level codes, and PSD for an equal probable sequence of 1 and zero is given by Equation (3.1)

$$S_1(f) = \frac{V^2 T}{4} \left(\frac{\sin \pi fT}{\pi fT}\right)^2 + \frac{V^2}{4} \delta(f)$$
(3.1)

Where V is binary 1-bit voltage, T=1/R bit duration and R is the data rate in bit per second. There is several reasons to use this level coding in an optical communication system as it requires electrically low bandwidth compared to RZ for the transmitter and receiver. Additionally, this is less sensitive to laser phase noise as compared to PSK and efficiently used in DWDM system and not affected by disturbance of multichannel. To convey information from the optical fiber channel, this line code is modulated by the light source as Intensity modulation or OOK. The transmitter block diagram is shown in Fig. 3.7.



Fig. 3.7 NRZ- OOK transmitter; reproduced from ref. [109].



Fig. 3.8 Optical signal spectrum.

The block diagram of an NRZ-OOK transmitter is shown in Fig. 3.7, where pseudorandom binary sequence (PRBS) generator generates randomly binary bits as data information, which are converted to the electrical waveform as NRZ level signal [108-110]. Laser diode generates continuous light pulse which is modulated with an intensity modulator (IM) as according to information waveform with using proper biasing at Modulator. Two types of IM which are mainly used in optical communication are Mach Zehnder modulator (MZM) and electro-absorption modulator. Both converts NRZ electrical signal to NRZ-OOK optical signal with same data rate. At receiver, we can detect this NRZ-OOK signal by using PIN or APD photodiode which converts the optical power of signal into electrical signal. These types of detection methods are known as direct detection (DD). This receiver circuit of direct detection is simple hence cost is also less compared to others. The optical signal spectra are shown in Fig. 3.8 indicates the zero frequency components and DC contents, which are undesirable. These types of coding are not good choices for circuits where inductive and capacitive components are used. At high data rate and long distance NRZ is not the best choice in optical communication [111]. The RZ is becoming popular nowadays for high bit rate long haul networks. Previously much attention was not given to this due to larger bandwidth requirements compared to NRZ. However, in recent years, it is shown that RZ has superior performance over NRZ in the presence of chromatic dispersion, fiber nonlinearity and due to other many advantages like these. In recent years, it has been shown that RZ can have superior performance over NRZ in certain regimes where chromatic dispersion and fiber nonlinearities [112]–[114] also exhibit solitons like properties. In addition, RZ has greater tolerance to polarization mode dispersion than NRZ [115].

### 3.3.2 RZ-OOK

Level coding which is Return to zero (RZ) and the optical signal width is less then its bit period. RZ means 'return-to-zero,' so the width of the optical signal is smaller than its bit period. A clock signal with an equal data rate to an electrical signal is used to generate RZ shapes of the optical signal [42]. In this line code, a binary one is represented by a voltage level with half bit interval and 0 bit is represented by the 0 voltage of the full bit duration. Thus this is the instantaneous level codes. The waveform of unipolar RZ signal is shown in Fig. 3.2. The PSD of unipolar RZ line codes for equally likely bit is given by Equation (3.2)

$$S_2(f) = \frac{V^2 T}{16} \left( \frac{\sin \frac{\pi f T}{2}}{\frac{\pi f T}{2}} \right)^2 + \frac{V^2}{4\pi^2} \left[ \frac{\pi^2}{4} \delta(f) + \sum_{n=-\infty}^{\infty} \frac{1}{(2n+1)^2} \delta(f - (2n+1)R) \right]$$
(3.2)

Where V is the binary one voltage level and T = 1/R is the bit periods. Fig. 3.3 shows the spectrum of RZ signal. The advantages of RZ are easy generation, and self-clocking means no need of extra component and allows simple timing recovery [113-115]. The generation of RZ –OOK signal is shown in Fig.3.9.



Fig. 3.9 Block diagram of a RZ transmitter.

First intensity modulator generates the NRZ optical signal followed by another intensity modulator, which generates RZ optical signal using a synchronized electrical pulse train with the same data rate.



Fig. 3.10 Power spectra of RZ-OOK.

The spectrum of RZ-OOK is also shown in Fig. 3.10. It has wide spectra compared to NRZ that leads to less spectral efficiency for WDM system also have the presence of zero Frequency components which lead to DC wander. For continuous zero bit, it could lead to the loss of synchronization. It has no error detection capability hence it will not be suitable for performance monitoring. In the next section, we will review the Manchester coding which has no DC content at zero frequency.

### 3.3.3 Manchester Coding

In the Manchester level coding binary bit (1) is represented by positive half period pulse followed by negative half bit period pulse [42]. Zero bit (0) is represented by the negative half bit periods followed by a positive half bit period as shown in Fig 3.11. It is classified under instantaneous transition code, which has no memory.



Fig. 3.11 Binary to Manchester generation of waveform; reproduced from ref. [42]

For equally likely bit of 0 and 1 then power spectral density (PSD) is given by Equation (3.3)

$$S_3(f) = V^2 T \left(\frac{\sin\frac{\pi fT}{2}}{\frac{\pi fT}{2}}\right)^2 \sin^2\left(\frac{\pi fT}{2}\right)$$
(3.3)

This coding has advantage of no DC components as well as provides good synchronization due to 1 and 0 are represented by the transition from the high to low and low to high respectively. These transitions are sure for each bit of one or zero so easy to extract timing information. The optical modulation can be done by applying the Manchester signal to intensity modulator as discussed in previous techniques. The optical power spectra are shown in Fig 3.12. The major drawback of this coding is that it has twice of the bandwidth of the NRZ or other common code and the twice of the data rate (2Rd). Also, performance monitoring is not possible because it has no error detection capability. In the following section, we will study about the Differential phase shift keying (DPSK) in which information is in the phase of the carrier and useful to improve the spectral efficiency as well as nonlinearity tolerant for optical communication.



Fig 3.12 Manchester optical spectrum

#### 3.3.4 Differential Phase shift keying

The line coding studied till now represents the instantaneous optical power level with intensity modulation. Similarly, the digital information is represented by the phase of carrier signal known as phase shift keying (PSK) [42]. In starting, the laser source was

immature to generate the single frequency and stable phase. Hence phase based modulation was not possible. However, now due to rapid growth in the light source technology, use of PSK is feasible for practical optical system especially differential phase shift keying (DPSK) is most often used modulation format.

Fig. 3.13 (a) shows the block diagram of a typical NRZ-DSPK transmitter. Electrical NRZ signal converted to DPSK signal as Fig. 3.13 (a) then it is modulated by external electro-optic phase modulator. The optical signal 1 bit is represented by phase  $\pi$  and 0 bit is represented by no change in phase.



Fig. 3.13 NRZ- DPSK (a) Encoder (b) Modulator; reproduced from ref. [109].

The optical signal power of DPSK signal is always constant. To detect the signal at the receiver side, a complex circuit is used with Mach-Zehnder- interferometer (MZI) as shown in Fig. 3.14 (b).



Fig. 3.14 DPSK Receiver circuit; reproduced from ref. [109]



Fig. 3.15 Power spectrum of DPSK

The MZI is mainly correlating each bit to its previous neighbor bit to convert phase information into the intensity. It has two port, one is constructive and other is a destructive port. At constructive port, if two consecutive bit is in the same phase it adds and gave high signal power otherwise for different phase, they cancels one another, and it shows a low level of power, Vice versa occurs at the destructive port. The photodiode converts these signal into photocurrent and combine (logically subtracted) them to double the signal strength.

In the NRZ DPSK, the power level is a constant and optical phase shift between 0 to  $\pi$  (as 1 and -1), and average optical field gets zero [116-117]. So there are no carrier components present in power spectrum of NRZ- DPSK as shown in Fig. 3.15 compared to NRZ OOK which has strong carrier components shown in Fig. 3.8. The main drawback of this coding is complex receiver circuits hence cost increases at the user end. In the next section, we will study the Duobinary and dicode which is the part of multilevel signaling.

#### **3.3.5 Multilevel signaling**

Till now we studied about two level (zero and one) signal and their characteristics, but these level of signal can be increased from more than two. The multilevel signal allows higher transmission efficiency and useful in the band limited application [42]. Transmission bit rate can be given as Equation (3.4)

$$R_{b} = R_{\delta} \log_{2}[L] \tag{3.4}$$

Where  $R_b$  is bit rate,  $R_s$  is baud rate or signaling rate and L is the number of level and symbol rate is less than the bit rate. The multilevel signaling compresses the required transmission bandwidth at the same bit rate and allows improvement in transmission efficiency also known as partial response signaling [42]. In the partial response signaling the controlled amount of inter-symbol interference is allowed that can be removed at the receiver end. This allowed interference is used to reshape the spectra that will make it less sensitive to timing error.

### 3.3.5.1 Duobinary

Duobinary Coding is the special case of partial response coding, which requires the minimum bandwidt which is the half of the bit rate. Duobinary Signals can be generated digitally by applying a unipolar binary data signal to the input of a one-bit delay shift register with subsequent addition of the delayed and undelayed versions of the original data signal as shown in Fig. 3.16. The generated electrical duobinary signal is modulated by Intensity modulator and waveform is shown in Fig. 3.17.



Fig. 3.16 Duobinary encoder [118].



Fig. 3.17 Generation of Duobinary waveform [118].

The power spectral density of encoded Duobinary signal is given as Equation (3.5)

$$P(f) = T.\frac{\sin^2(2\pi fT)}{(2\pi fT)^2}$$
(3.5)

The simple intensity modulator can be used for modulation of this signal and optical spectrum after modulator is given as



Fig. 3.18 Power spectra of Duobinary after modulation.

From Fig. 3.18 it is clear that for the signal of 10 Gbps data rate the bandwidth achieved is 5 GHz, which means 10 GHz is compressed to 5 GHz. The Optical Duobinary Coding utilizes the optical phasefor 1 and -1 and intensity profile is same as binary IM system, so it has the following advantages [119-120]. First, a Binary IM-DD receiver can be used at the receiver for the detection of the received signal, and same receiver sensitivity can achieve as a binary IM-DD system. Secondly, it has higher tolerance to dispersion due to reduced spectrum width compare to binary IM signal. Also the optical Duobinary signal has no carrier frequency components, so input power limitation is relaxed due to SBS.

### 3.3.5.2 Dicode

The Dicode is a special spectrum reshaping technique to reduce the low-frequency DC component compared with the conventional NRZ coding which will improve noise performance as well as nonlinearity [43-48]. It is correlative coding which has the advantage to permit some amount of interference at the receiver end to reproduce the original message. The block diagram of the dicode encoder is shown in Fig. 3.19.



Fig. 3.19 Dicode encoder [48].

Dicode coding can be defined mathematically as [45]

A rectangular pulse  $f_1(t)$  can represent as an impulse response

$$\partial(t) = \lim_{t \to 0} f_1(t)$$

Such that

 $A \rightarrow \infty$  $AT \equiv 1$ 

For NRZ we find amplitude spectra

$$\left|\operatorname{sinc}\left(\frac{\omega T}{2}\right)\right|$$
 (3.6)

Then power spectral density (PSD) for NRZ can be express by

$$\left|\sin c^2 \left(\omega T \,/\, 2\right)\right| \tag{3.7}$$

In Dicode

$$\partial T(t) = \partial(t) - \partial(t - T)$$
 (3.8)

T is delay time

The amplitude spectrum can be express as

$$|F[\partial T(t)]| = |1 - e^{-j\omega T}|$$

$$|1 - \cos\omega T + j\sin\omega T|$$

$$[(1 - \cos\omega T)^{2} + (\sin\omega T)^{2}]^{1/2}$$

$$[1 - 2\cos\omega T + \cos^{2}\omega T + \sin^{2}\omega T]^{1/2}$$

Then finally we get

$$2\left|\sin\frac{\omega T}{2}\right| \tag{3.9}$$

The power spectral density (PSD)

$$4\left|\sin^2\frac{\omega T}{2}\right| \tag{3.10}$$

As expressed in equation (3.6 & 3.9), low-frequency component gets reduced which can be verified by a graphical representation of dicode coding as shown in Fig. 3.20. The spectra depicted in Fig. 3.20 have a sine envelope.



Fig. 3.20 Power Spectra of (a) conventional NRZ (b) dicode-coded NRZ.

It is observed that at the Nyquist frequency the amplitude is maximum whereas zero at zero frequency and the double frequency of Nyquist frequency. This type of spectral distribution helps to employ the filter having a sharp cutoff frequency for pulse shaping as shown in band limited dicode waveform and also suitable for transmission over a.c. Coupled-channels [48].

Dicode signal can be modulated by the lithium niobate intensity modulator to convert this electrical dicode signal to optical dicode signal by using proper biasing at the modulator. The transmitter block diagram is same as simple binary IM. Now we analyse the generated waveform at transmitter and receiver as shown in Fig. 3.21 and Fig. 3.22.



Fig. 3.21 Electrical waveform (a) NRZ signal of binary data (b) Output after Precoder (c) Encoded dicode signal.





Fig. 3.22 Dicode optical signal (a) Intensity (b) Phase (c) Power spectra.

The effects of dicode coding are analyzed and described in next chapter with different design architecture for different applications.

# **3.5 Chapter Summary**

Different line coding characteristics have been discussed in this chapter in detail. We have observed the impact of different line coding on the various applications. Also, different modulations formats such as NRZ, RZ, DPSK, dicode and Duobinary also have been studied in detail and their transmission characteristics have been observed. We found that dicode coding is more resilient to dispersion and nonlinearity as compared to other coding methods.

# Investigation and Mitigation the various effects in WDM-TDM Networks

# 4.1 Introduction

As bandwidth demand has increased due to emerging broadband services like video on demand, Internet Protocol television (IPTV) and medical applications, it has motivated the evolution of high-speed passive optical network in order of gigabits. Existing Gigabit Passive Optical Networks (GPON) offers a capacity of 2.5/1.25 Gbps for downstream and upstream for the physical reach around 20 Km with 32 splitters [18]. To minimize the Center offices (CO) in the sparsely populated geographical area, it is necessary to increase the reach and number of users. This further reduces the overall (CAPEX) capital expenditure of networks.

Increasing the number of users at splitters, results in increased power loss and reduction in available bandwidth for users. To address this problem, FSAN and ITU-T proposed NGPON1 and NGPON2 [20]. NGPON1 is upgraded network widely known as XGPON with enhancing capacity supporting higher number of users. In this chapter, we study the effects and mitigation of Raman cross talk that arise due to high power video and data signal with duobinary and dicode, spectral analysis with remotely Raman amplified and effects on dispersion for the long reach network using erbiumdoped fiber amplifier.

# 4.2 Raman Crosstalk

Introducing broadband data, voice and video services to residential and business users together in the same network requires a cost effective and high data rate hybrid WDM-TDM system. [121]. These types of networks works in two different wavelength bands (1480 nm and 1550 nm) for data and video at high power. At short distance ~ 20 Km, the performance of data signal is degraded significantly due to Raman cross talk [122]. Video and data signal lies within the Stokes shift ~100 nm of silica. Hence, the data signal acts as a pump for the video signal, resulting in a reduction of the power of the data signal [123-124]. Some techniques which previously discussed in literature for

reduction in cross talk like dynamically varying power of digital signal [125], preemphasis of lower channels [123], decreasing wavelength spacing [126], using polarizer scrambler [127], Quadrature amplitude modulation (QAM) for modulating the video signals [128] and by compensating the signal attenuation using remote Raman pump [129]. Generally, for a particular transmission link, received signal quality at receiver is different for different data formats due to their different waveforms and spectra [130]. Crosstalk between reflected light and signal is reduced by a three level Partial response (PR) coding by reforming the signal spectra by altering the signal's power distribution with PR coding. This is done by generating a special 1 bit/symbol multilevel signal via superposition of adjacent symbols [48].

### 4.2.1 Generation of partial response coding

Balance Duo- binary coding: Balance duobinary coding can be derived from a binary digital sequence by pre-coding the sequence into another binary sequence  $d_n$  [45] as shown in Table 4.1.

| Binary input          | $a_n$                         |   | 1 | 0 | 0 | 1 | 1 | 0 | 1 | 0  |
|-----------------------|-------------------------------|---|---|---|---|---|---|---|---|----|
| Precoding             | $d_n = a_n \bigoplus a_{n-1}$ | 0 | 1 | 1 | 1 | 0 | 1 | 1 | 0 | 0  |
| Duobinary             | $b_n = d_n + d_{n-1}$         |   | 1 | 2 | 2 | 1 | 1 | 2 | 1 | 0  |
| Balanced<br>duobinary | $c_n = b_n - 1$               |   | 0 | 1 | 1 | 0 | 0 | 1 | 0 | -1 |

Table 4.1 Generation of Balance Duo-Binary Code

Where  $d_n$  is the differential code signal,  $a_n$  is the original binary signal,  $b_n$  duo binary signal and  $c_n$  is the balance duobinary signal.

Dicode coding: - A dicode sequence can be obtained by converting the  $d_n$  sequence to another sequence  $b_n$  by subtracting the previous digit from each member of the  $d_n$  sequence [45]. Dicode coder uses one logic gate (X-OR) and 1 bit delay. Dicode coding is power efficient and more tolerable to dispersion and nonlinearity due to the absence of zero DC components. In the section 3.3.5.2, generation of Dicode code with its power spectra characteristics is explained in detail.

| Binary input | an                            |   | 1 | 0 | 0 | 1  | 1 | 0 | 1  | 0 |
|--------------|-------------------------------|---|---|---|---|----|---|---|----|---|
| Precoding    | $d_n = a_n \bigoplus d_{n-1}$ | 0 | 1 | 1 | 1 | 0  | 1 | 1 | 0  | 0 |
| Dicode       | $b_n = d_n - d_{n-1}$         |   | 1 | 0 | 0 | -1 | 1 | 0 | -1 | 0 |

 Table 4.2 Generation of Dicode Code

Table 4.2 Illustrates pre-coded output  $d_n$  and Dicode output  $b_n$  for the corresponding binary input  $a_n$ .

#### **4.2.2 Simulation Setup and Results**

In the system shown in Fig. 4.1, five wavelengths each at 10 Gbps are multiplexed and transmitted from the optical line terminal (OLT) to a 20 km long optical fiber. Three out of five wavelengths (1558, 1550, 1542nm) for video services are transmitted with 20 dBm power and remaining two wavelengths (1490, 1482nm) for data services at 0 dBm. The transmitted signals are divided into 64 parts using 1:64 power splitters to serve 63 ONUs. At the ONU, a Gaussian filter separates the video and data signals. Instead of  $64^{th}$  ONU, a counter-propagating Raman pump (p~180mW, 300mW) is introduced to amplify the downstream signal.



Fig. 4.1 A WDM-TDM system with remote Raman pump

By comparative study from Table 4.3 and Table 4.4, we observed that the power reduced by scattering in Dicode coding is approximately 0.5 dB less when compared to the balance Duobinary code. Raman scattering and fiber loss can be minimized further

by amplifying the signal using remote Raman pump by more than 8dBm power as shown in Table 4.4 and length of the system can be increased up to 60 Km.

| S. No. | Wave<br>Length<br>(nm) | Power<br>without<br>scatterin<br>g (dBm) | Power<br>with<br>scattering<br>(dBm) | Power<br>reduced by<br>scattering<br>(dBm) | Power with<br>Raman pump<br>(dBm) | Power<br>Improved<br>by Raman<br>pump (dBm) |
|--------|------------------------|--|--------------------------------------|--|-----------------------------------|---|
| 1.     | 1482                   | -10.4                                    | -12.3                                | 1.9  | -4                                | 8.3   |
| 2.     | 1490                   | -10.4                                    | -11.9                                | 1.5  | -2.5                              | 9.4   |

Table 4.3 for Balance Duobinary Coding

| S. No. | Wave<br>Length<br>(nm) | Power<br>without<br>scattering<br>(dBm) | Power<br>with<br>scattering<br>(dBm) | Power<br>reduced<br>by<br>scattering<br>(dBm) | Power<br>with<br>Raman<br>pump<br>(dBm) | Power<br>Improved by<br>Raman pump<br>(dBm) |
|--------|------------------------|---|--------------------------------------|---|---|---|
| 1.     | 1482                   | -16.2                                   | -17.6                                | 1.4   | -9.2                                    | 8.4   |
| 2.     | 1490                   | -16.2                                   | -17.3                                | 1.1   | -7.8                                    | 9.5   |

**Table 4.4 for Dicode Coding** 

Here, we have analyzed the Raman crosstalk and also reduced it by using Dicode coding. By simulation, we observed that stimulated Raman scattering (SRS) dominate more for balance duobinary code then Dicode and this value is 0.5 dB approximately. Raman scattering and fiber losses are also compensated by amplifying the downstream data signal through remote Raman pump.

### 4.2 Spectral Analysis

To cover large geographical area, long reach optically amplified hybrid WDM-TDM PONs are in huge demand due to its low cost and high bandwidth [131-132]. A Basic PON system uses a 1490 nm and 1550 nm wavelength for downstream transmission for combined data/voice and video traffic distribution [8], [133]. A data and video signal together in WDM-TDM system at short distance degrades the performance of data signal significantly due to Raman cross-talk [134].

The Raman amplification used in an optical fiber communication system has attracted much attention as they have broad amplification bandwidth and flexible central wavelength [129,135]. The discrete Raman amplification and distributed Raman amplification are two main categories used in Raman amplification [136]. The gain and typical transmission length for distributed Raman amplification is larger and it improves system performance. The transmission rate of WDM system can be enhanced by either increasing spectral efficiency or a wider optical bandwidth or both. The spectral efficiency limit is determined by the information-theoretic capacity per unit bandwidth. The DPSK modulation technique allows information to encode in two degrees of freedom per polarization and its spectral efficiency limits are much higher in typical terrestrial systems [136-137]. The constant intensity modulated systems have only one degree of freedom, which reduces spectral efficiency [138]. There are some other modulations like Quadrature Phase Shift Keying (QPSK) and Differential Quadrature Phase Shift Keying (DQPSK) as they have high Optical Signal to Noise Ratio (OSNR) sensitivity but they are more complex in design and their implementation cost is very high [109,140-141].

In practice, most of the PON systems have 20 Km physical reach with 32 splits and 28 dB of link budget [70]. A FTTH with GEPON system operating at 2 Gbps is proposed by D. Kochar et al. [142] the system was simulated for different users like 32, 48, and 64 at 20 Km fiber length with booster amplifier. The complete set up depicts better performance for 56 users. Hesham et al. [143] observe the effect of power for a GPON system, they proposed that 9 dBm power is suitable to achieve 128 users for 20 Km fiber length. This methodology if implemented for the WDM systems, where more than one wavelength is multiplexed the average power is increased which results an increase in nonlinearity, and degradation of the system performance. R. Kaler et al. [144] propose a GEPON (Gigabit Ethernet passive optical network) for 0-20 km optical fiber length for a different number of user and achieve 15 Km is suitable length for the 10 Gbps data rate system.

N. cheng et al. [145] demonstrates a 20 Km TWDM PON system for dual rate transmitter, delay modulation using Directly Modulated Laser diode (DML) for 10 Gbps transmission. Power over fiber technique is proposed by Rivael S. Penze et.al [146] to achieve the 10 Gbps data rate system for 32 users at 50 Km. This technique can be deployed to amplify entire optical band, but it fails to amplify individual

selective wavelength. K. Acharya et al. [129] demonstrated a system with a remote Raman amplification where a continuous wave counter-propagating pump is injected from one of the optical network unit. The set up was tested for 1.25 Gbps data rate for 32 or 64 users for 40 km and 25 Km length respectively. As per our knowledge in video overlay PON network the remote Raman pump is the best choice.

In this section, we have discussed the effective distributed Raman amplification in WDM-PON using DPSK modulation for enhancement of reach and system capacity. The benefit of using DPSK modulation along with Raman amplification is that it results in enhanced spectral efficiency of the signal due to DPSK and increased transmission length by Raman amplification. Hence, the signal can be transmitted for long distance with the higher transmission rate. The spectral efficiency is directly proportional to system capacity which is also a most economical mean to increase DWDM system capacity. It has also been stated in previous work that the spectral efficiency is also reduced in constant intensity or direct detection.

### 4.2.1 Modeling and Equations Analysis

**Effective Raman Gain:** - The larger effective area  $A_{eff}$  of the fiber is beneficial for optical amplification as it reduces nonlinear effects when we launch high signal power. Moreover, in Raman pumping the gain increases when effective area of the fiber decreases. The effective Raman gain ( $G_e$ ) is the ratio of Raman gain coefficient over effective area of the fiber. Mathematically it can be defined as  $\frac{g_R}{A_{eff}}$ , which is more practical parameter than Raman gain coefficient in Raman amplifier. We can determine effective area for Raman gain as  $A_{eff} = a_{Eff} \pi \left(\frac{Diameter}{2}\right)^2$ , here  $a_{Eff}$  is a dimensionless number, which is related to actual effective area of the mode to the core diameter. Similarly the Raman gain coefficient  $g_R$  describes how stokes power rises as pump power is transferred to it through stimulated Raman scattering (SRS). The Raman gain in terms of power can be described as Equation (4.1)

$$\frac{dP_s}{dz} = \frac{\gamma_R}{A_{eff}} P_p P_s \equiv g_R P_p P_s \tag{4.1}$$

Here  $\gamma_R$  is the nonlinear strength  $\gamma_R = 2\pi n_2/(\lambda A_{eff})$ ,  $n_2$  is nonlinear refractive index and  $\lambda$  is the signal wavelength. The power evolutions for signal and pump waves along
an optical fiber can be described by using different equations called propagation equations. The signal and pump equation can be expressed as

$$\frac{dP_s}{dz} = g_R P_s P_p - \alpha_s P_s \tag{4.2}$$

$$\frac{dP_p}{dz} = -\frac{\omega_s}{\omega_p} g_R P_s P_p - \alpha_p P_p \tag{4.3}$$

Where  $P_s$  and  $P_p$  are signal and pump powers, respectively,  $g_R$  is the Raman gain efficiency for the wavelength of signal and pump,  $\omega_s$  and  $\omega_p$  are the frequencies of signal and pump waves and  $\alpha_s$  and  $\alpha_p$  are attenuation coefficients of optical fiber at signal and pump wavelength. If we consider the signal power low enough as compared to pump power, pump depletion is negligible. After solving the above Equation (4.2), when pump is off, the obtained output power is attenuated due to fiber loss only.

$$P_{s-off}(L) = P_s(0)exp(-\alpha_s L)$$
(4.4)

When the pump power is as large as compare to signal power than the power of weak signal amplified due to SRS. The amplification of the signal can described from the following Equations (4.5)

$$P_{s-on}(L) = P_s(0)exp\left(\frac{g_R P_0 L_{eff}}{KA_{eff}} - \alpha_s L\right) \equiv G(L)$$
(4.5)

Where the  $P_0$ ,  $P_s(0)$  are input pump power and stock power at zero fiber length, G(L) is the net signal gain, G (L) in dB is given by  $G(L) = 4.34 \left[ \frac{g_R P_0 L_{eff}}{A_{eff}} - \alpha_s L \right]$ , where L is the amplifier length and  $L_{eff} = \frac{1 - \exp(-\alpha_p L)}{\alpha_p}$  is effective length of the fiber. The net signal gain G(L) can be less than 1 (net loss) if the Raman gain is not sufficient to overcome the fiber loss. Than the Raman on-off gain  $G_A$  in dB can be defined as

$$G_A = 10\log_{10}\left(\frac{P_{s-on}(L) \text{ with pump on}}{P_{s-off}(L) \text{ with pump off}}\right)$$
(4.6)

Where L is the length of the fiber and  $P_{s-on}(L)$  with pump on is considered without amplified spontaneous emission noise (ASE) and thermal noise. After substituting the value of Equations (4.3) and (4.4) the effective Raman gain can be obtained as

$$G_e = \frac{g_R}{A_{eff}} = \frac{2\pi n_2}{\lambda A_{eff}^2} = \frac{G_A}{4.343 L_{eff} P_{pump}}$$
(4.7)

Here  $P_{pump}$  is the pump power launched into the fiber,  $\frac{g_R}{A_{eff}}$  is the depolarized effective Raman gain.

**Spectral efficiency limit:** - The performance analysis of broadband Raman amplifiers is affected by several factors which should be controlled. The Raman spontaneous scattering in Raman amplifiers appears as a noise signal as random phase associated with spontaneously generated photons. The spectral efficiency in DWDM can be defined as the ratio of channel capacity to the channel spacing. The channel capacity in communication is the maximum number of bits that can be transmitted through a channel without any error. This is common to all modulation formats because it does not depend upon encoding or decoding schemes. Hence this can also defined as the optimal probability density of the transmitted signal. If we consider S as a spectral efficiency, C capacity per channel, B occupied bandwidth per channel and  $\Delta f$  channel spacing, then  $S = C/\Delta f$  denotes spectral efficiency limit. The ASE spectral density is defined as Equation (4.8),

$$S_{ASE} = n_{sp} h v_0 g_R G(L) \int_0^L \frac{P_p(z)}{G(z)} dz$$
(4.8)

The parameter  $n_{sp}$  is defined as  $n_{sp}(\Omega) = 1/[1 - exp(-h\Omega/\kappa_B T)]$ , where  $\Omega = |\mu - \nu|$  is the Raman shift and T denotes absolute temperature of the amplifier. The spectral density is constant and present at all frequencies (white noise). In practice, the noise exists only over the amplifier bandwidth and further can be reduced by placing an optical filter at the amplifier output. According to this consideration, we can calculate the total ASE power after the amplifier as Equation (4.9),

$$P_{ASE} = 2S_{ASE}B_{opt} = 1.4S_{ASE}R_b \tag{4.9}$$

Here  $B_{opt}$  is the bandwidth of optical filter and  $R_b$  is the bit rate for signal transmission. The factor of 2 indicates two polarization modes of the fiber span. ASE can be reduced by 50%, if a polarizer is used after the amplifier. In our consideration if we are not using any polarizer, the signal to noise ratio of the amplified signal is given by Equation (4.10),

$$SNR_o = \frac{P_{S-on}(L)}{P_{ASE}} = \frac{G(L)P_S(0)}{P_{ASE}}$$
 (4.10)

In DPSK modulation technique the information is encoded in an optical signal having nominal constant intensity. Now at high signal to noise ratio (SNR), the capacity can be defined as Equation (4.11) [9].

$$C = B_{opt} \left[ \frac{1}{2} \log_2(SNR_o) + 1.10 \right]$$
(4.11)

Hence spectral efficiency can be given by Equation (4.12) [9].

$$S = \frac{B_{opt}}{\Delta f} \left[ \frac{1}{2} \log_2(SNR_o) + 1.10 \right]$$
(4.12)

The unit of spectral density is b/s/Hz. It is necessary to consider a heterodyne or phase diversity homodyne with differentially coherent (one-bit delay) demodulation of DPSK signal. This is only applicable for coherent detection because for non-coherent it never offers high spectral efficiency. Homodyne receiver requires bandwidth approximately equal to the symbol rate  $R_s$  which is just half of the heterodyne receiver bandwidth. Hence we prefer homodyne receiver which require a pair of balance receiver for detection. The optical DWDM de-multiplexer is used which provide narrowband filtering of the received signal and ASE. In the absence of fiber nonlinearities, with proper dispersion compensation and matched filtering, all formats of DPSK should provide same SNR and spectral efficiency. Under interferometric detection of DPSK, a Mach-Zehnder interferometric detector is used with path difference of one-symbol duration compares phase transmitted in successive symbols which provides intensity modulated output that is detected using balanced receiver.

#### 4.2.2 Simulation Model

The general features of the proposed topology are illustrated in Fig. 4.2. The highlighted segments for WDM-PON are optical line terminal (OLT) and optical network unit (ONU). The triple play (i.e. voice, video and data) services are considered, with data/voice signals transmission range from 1450 nm-1500 nm and video signals within 1550 nm-1560 nm range. The radio frequency (RF) video signals have wavelength 1542 nm, 1550 nm and 1558 nm and data/voice signals are having wavelength 1482 nm and 1490 nm. All signals are exhibited with pseudo-random bit sequence (PRBS) generator (bitrate-10 Gbps), non-return to zero (NRZ) modulator and phase modulated (PM) continuous wave (CW) lasers. The subcarrier multiplexing (SCM) is the method of combining (or multiplexing) many different communication signals so that they can be

transmitted along the optical fiber. The video transmitter consists of an adder to combine two electrical signals and a CW laser modeled with a phase modulator. Each SCM signal is using two channels from standard national television system committee (NTSC) and cable television (CATV) frequency plan. The channel number (2, 78), (3, 79) and (4, 80) combinations for each 1542 nm, 1550 nm and 1558 nm video signals are taken respectively. Each channel is transmitting separate frequency as shown in Fig. 4.2 within the frequency range 55 MHz to 1000 MHz. The Multiplexed output of all input signals is transmitted via forward input of the bidirectional single mode fiber (SMF) and another backward input is reserved for Raman pumping to achieve Raman amplification.



Fig. 4.2 Simulation model of proposed distributed Raman amplified WDM-PON architecture.

A bidirectional single mode fiber (SMF) with core diameter 8.2 micrometer is used for data transmission. An average fiber attenuation of ~0.25-dB/km is taken into consideration. The nominal channel spacing and Raman gain coefficient are 100 GHz and 9.8E-14 m/W respectively. Raman amplification also depends on two equal and unequal pump power (i.e. E-PP and UE-PP) configurations. In both configuration processes, lower wavelength data signals transfer their energy to higher wavelength signals due to stimulated Raman scattering (SRS) property of the fiber.

A splitter/WDM demux is used to transmit these services to different e.g. 32, 64 or 128, etc. users /splits. The video and data/voice signals are Raman-amplified using a pump of wavelengths 1481 nm, 1497 nm, 1513 nm, 1405 nm and 1415 nm respectively. A WDM-mux is used to combine all these pump wavelengths, and the multiplexed output

are applied via a remote optical node unit to the bidirectional fiber. The backward pumping configuration is preferred for Raman amplification as nonlinear effects are least in this case and the signal power is smallest throughout the link length. Raman gain is calculated and analyzed per signal before service distribution to the subscribers. At optical network unit (ONU), the received signals are separated into video and data signals using a suitable filter because the phase deviation in the modulator has been identified as a parameter that can affect the transmission. It has already been reported in previous research work that the DPSK modulated signal gets more power (approx. >3-dB) at optical network unit (ONU) terminal when using suitable demodulation techniques [138]. It is also to be noticed that in this technique there is no extra component introduced, such as a semiconductor optical amplifier (SOA) or highly nonlinear fiber or EDFA, besides remote Raman pumps.

#### 4.2.3 Results and Discussion

The launching powers, with and without Raman pump powers are shown in Fig.4.3. It was observed that half power (approx.) of the signal is reduced just after 20 Km fiber span. If we are again increasing the transmission length, the complete signal loss can be observed beyond the 40-50 Km fiber length. At receiver, the Eye diagrams of the pump signal can be observed by using electrical BER visualizer. The increased number of users also reduces the received power for each individual user. Hence if Raman pump is in the off-stage (i.e. without pump), the received power with increased number of users will also reduce the transmission length. Thus,



Fig. 4.3 Received power and eye diagram analysis with and without Raman pump.

The higher wavelength signals have higher gain in comparison to lower wavelength data/voice signals as defined in past research work. It is also essential that each signal must have at least 7dB gain enhancement for both pump power configurations to achieve the desired transmission length or number of users or both. If we compare eye height and signal quality of all signals, the higher wavelength signals will always have wider eye opening and better signal quality in comparison to lower wavelength signals. In equal pump power (E-PP) configuration, the 100 mW pump power is taken for each pump wavelength whereas, for unequal pump power (UE-PP) configuration, the pump power decreases with increasing pump wavelength. It can be observed form the plot of Fig. 4.4 (a) that the Raman gain at any fiber length is higher for E-PP than in the UE-PP pumping mode. Raman gain is dominant for shorter fiber length and Raman on-off gain always increases with increasing transmission distance. Raman Gain increases up to the effective length of the fiber and decreases beyond it due to nonlinear effects.





Fig. 4.4 (a) Raman gain, (b) Optical Signal to noise Ratio, (c) Effective Raman gain and (d) spectral efficiency analysis with increasing fiber length.

The Raman on-off gain is calculated in equation 4.6, after observing the received power at fiber end for both on and off stages of Raman pumps. Effective Raman gain is observable from the plot that initially this gain increases and after some fiber length it becomes saturated due to constant Raman gain coefficient and effective area of the fiber. The result analysis for both OSNR and spectral efficiency parameters behaves in the same way as Raman gain with respect to each fiber span. The only difference is that they are oppositely related with the Raman gain at each pump power configuration. This signifies that OSNR, as well as spectral efficiency, is higher and Raman gain is lesser for E-PP case at any fiber span or vice-versa. In each configuration, reach and data transmission rate are dependent on the effective Raman gain and spectral efficiency parameters. The maximum 55 Km reach achieved for 32 users and beyond this length, the received power at the photo-detector is below its acceptable level. Hence no communication can be achieved.

The result analysis of WDM-PON system provides 55 Km transmission reach for 32 users and 0.3 bits/s/Hz higher spectral efficiency with 10 Gbps transmission speed. The network system efficiency is upgraded by increasing the transmission bit rate, distance, and capacity of the system to support a maximum number of users or subscribers. There are many more interesting challenges that remain for the future work. One can use bi-directional Raman pumping configuration for Raman amplification under the effect of temperature variation along the fiber cable. This yields highest system capacity per channel and ultra-long haul transmission distance for advanced optical communication system. Further investigations of non-linear effects, including non-linear phase noise on DPSK systems, are also feasible for longer transmission distance.

## 4.3 Analysis of Group Velocity Dispersion (GVD)

To overcome the challenges in NGPON1 a new network with ODNs, independent of GPON standard was designed as NGPON2 [147, 78]. Wavelength and Time division multiplexing (WDM-TDM- PON) Network is a primary broad band solution which is selected for NGPON2 [148]. It includes the advantages of both Time division multiplexing (TDM) and WDM (wavelength division multiplexing) Passive Optical Networks. In the network, 40-Gbps downstream and 10-Gbps upstream is achieved by combining XG-PONs via multiple pairs of wavelengths [77, 149]. With careful selection of wavelength band, the designed network can function as TWDM PON that

ensures full compatibility with Gigabit-capable PON (GPON), XG-PON and RF-video. The operating wavelengths for TWDM-PON fall in C- (1524 nm~1544 nm) and L + (1596 nm~1603 nm) bands [150-151]. The networks with increased length and higher split ratio were proposed, with the standards for the networks set by ITU-T.G.987.2. The XGPON obeys standards of ITU.T.G.987.2 operates for the maximum length of 25 Km and 128 splitters for with 35 dB maximum optical path loss (OPL) [149]. The performance of the network system can be improved by improving the power budget with the use of erbium doped amplifier, semiconductor amplifier and Raman amplifiers [152-154]. Another technique is power over fiber and tested to offer a complete virtual passive network for 50 Km length and 1:32 split ratio [146].

Takuya T Tsutsumi et [155] al successfully demonstrated 41.3-km-reach and 128-split in an XG-EPON system. However, a phenomenon referred to as group velocity dispersion GVD which leads to the broadening the optical pulses as they propagate through the single mode fiber and thus affecting the bit rate is a considerable issue. Therefore in this chapter, we focus on providing an analytical analysis concerning the GVD-based power penalty to evaluate the maximum allowable bit rate for optical transmission without the need for using any dispersion management techniques (DMTs). A monochromatic optical signal when modulated ideally no longer remains monochromatic and has different frequency components that travel through an optical fiber with different group velocity, resulting in a phenomenon referred as group velocity dispersion or intramodal dispersion. This phenomenon contributes to both material dispersion and waveguide dispersion.

As the waveguide dispersion is fiber dependent and is determined during fiber manufacturing, we mainly focus on material dispersion. In lightwave system, effect of group velocity dispersion on the performance of the receiver can be broadly categorized as below. Firstly, due to pulse broadening the pulse peak power get reduced further reducing the SNR. To obtain the same BER, SNR should remain constant and is possible by increasing the average power at the receiver. This additional power required to obtain desired BER is known as power penalty.Secondly, the broadening of the pulse results in intersymbol interference which degrades the performance of the receiver. To minimize this effect, at the receiver end a decision circuit is so designed that on the application of input signal it corresponds to maintain the transfer function of raised cosine filter. In the practical optical system, the light signal applied do not have perfect rectangular shape, so for analysis and approximations, a bell-shaped Gaussian pulse is preferred over all the pulses. The propagation of optical pulses through single-mode fiber can be mathematically expressed by nonlinear Schrodinger equation (NLSE) as given below [156]

$$i\frac{\partial A}{\partial Z} = -\frac{i\alpha}{2}A + \frac{\beta_2}{2}\frac{\partial^2 A}{\partial t^2} - \gamma|A|^2A$$
(4.13)

Where A is the slowly varying electric field pulse envelope, Z is the direction of propagation; t is time,  $\alpha$  is constant for loss and  $\gamma$  is nonlinearity coefficient. The dispersive and nonlinear effects in the fiber are related to puls width (T<sub>0</sub>) and peak power (P<sub>0</sub>) of input pulse.

To study the effect of GVD on optical pulse propagation the medium is assumed to be linear dispersive and nonlinear losses are ignored. Thus the Equation 4.13 can be modified as

$$i\frac{\partial A}{\partial Z} = \frac{\beta_2}{2}\frac{\partial^2 A}{\partial t^2} \tag{4.14}$$

Where Z is the propagation direction, t is the time; A is the electric field envelope,  $\beta_2$  is group velocity dispersion (GVD) and is defined as Equation (4.15)

$$\beta_2 = \frac{\partial^2 \beta}{\partial \omega^2} \tag{4.15}$$

Equation 4.15 represents the GVD parameter, defined as the second order derivative of the fiber mode propagation constant with respect to the frequency of input signal.

A Gaussian pulse is an incident on the optical fiber can be expressed as [157,158]

$$E(z=0,t) = \sqrt{p_0} \exp(-\frac{t^2}{2T_0^2})$$
(4.16)

Where pulse width  $T_0$  is related to the pulse full width at half maximum by  $T_{FWHM} \approx$  1.665  $T_0$ . The pulse broadening with distance z is expressed as [155]:

$$T(z) = \left[1 + \left(\frac{z}{L_D}\right)^2\right]^{1/2} T_0$$
(4.17)

and, these results to change in the peak power, due to GVD, and can be written as:

$$P(z) = \frac{p_0}{\left[1 + \left(\frac{z}{L_D}\right)^2\right]^{1/2}}$$
(4.18)

$$L_D = \frac{T_0^2}{|\beta_2|} \tag{4.19}$$

Equation (4.19) represents dispersion length; pulse propagates equally to the distance of  $L_D$  and pulse get by a factor of  $2^{1/2}$ .

A 50 Km long communication system operating at 10 Gbps data significantly suffers from GVD. The dicode coding is used to minimize the GVD effects at low frequency; the Fig. 4.5 depicts the spectra of dicode coding [48]. About Equation 3, the dicode coding offers improved performance over conventional NRZ coding in an optical communication system.



Fig. 4.5 Power spectra of dicode signal.

#### 4.3.1 Network architecture and simulation setup

The architecture shown in Fig. 4.6 is used to analyze the effects of dicode coding on group velocity dispersion (GVD). In the proposed work, we have analyzed and observed the results related to downstream signal only. The simulation work was carried out with a CW laser source operating at wavelength 1550 nm, 1550.8 nm, 1551.60 nm and 1552.40 nm with 0 dBm power. At transmitter end, a (PRBS) generator generates a binary signal at 10 Gbps. The binary signal is converted into electrical pulses by using suitable pulse generator. The optical signals are generated by using lithium niobate modulator as a light source. The modulated optical signals are

multiplexed by WDM with insertion loss of 5 dB. The loss due to splitter was of 3.5 Log N, where N is a number of users at the splitter. The losses are compensated by amplifying the signals with erbium-doped fiber amplifier (EDFA). The EDFA offered gain of 20 dB and 4 dB noise figure. Then signal transverses through a fiber of length up to 100 Km with fiber loss of 0.2 dB / Km and dispersion -16.75 Km/nm-ps. During propagation the pulse gets broadened due to GVD, resulting in signal distortion with power loss.



Fig. 4.6 Simulation setup of WDM-TDM-PON.

At the remote end, the signal is split for N users ranging from 2 to 256 users. In the broadcasting mode, all users receive signals of the same wavelength. The setup is used for broadcast mode. At the receiver end signals are filtered using Bessel filters. This optical signal drives the PIN photo detector with a responsivity of 1 A/W and thermal noise density 100e<sup>-24</sup> W/ Hz to generate an electrical signal. The electrical signals are filtered by Low pass filters to remove the noise. A BER analyzer is used to analyze the signal quality in the form of Bit Error Rate and quality factor.

**Power Budget:** An optical network system requires the estimation of power budget with an approach that sufficient power reaches the receiver, and satisfactory performance is maintained for the entire lifetime of the system. The power budget sets the limit for the losses in the system the losses incurred in the system. The power budget is used to estimate the length of the network. The power budget can be mathematically expressed as below [4]. We estimate the power budget for two access network operating

at BER of 10<sup>-10</sup> including the losses due to various passive components. Maximum allowable system loss can be calculated as:

Рмах = output transmitter power – optical receiver sensitivity + Amplifier gain (if used) (4.20)

(I) For NRZ =
$$0-(-31) + 20 = 51 \text{ dB}$$
 (4.21)

(II) For Dicode = 
$$0 - (-36.5)$$
 +20 dB= 56.5 dB (4.22)

As calculated the power budget in Equation (4.21, 4.22), a system with dicode coding shows improvement over NRZ coding both for a number of users and length of the system.

#### 4.3.2 Results and Discussion

To explore the feasibility of improving the downstream reach in WDM- TDM PON system, we analyze the results of simulation measuring group velocity dispersion effects on dicode as well as NRZ signal. The effect of GVD on receiver sensitivity, splitter, and the networks reach was also analyzed with the observations and results. Fig. 4.7 shows the RF spectra of dicode in the electrical signal.



Fig. 4.7 Electrical Dicode signal.



Fig. 4.8 Downstream signal after multiplexer.

The spectra depicts that the zero frequency components power is reduced and the first lobe has the signal bandwith is 10 GHz which is equal to traditional NRZ coding. Fig. 4.8 shows the modulated downstream optical signal with the use of dicode coding in a network for different wavelength passing through the EDFA and optical fiber.

| S. No. | GVD     | NRZ BER | Dicode BER        | Remarks                  |
|--------|---------|---------|-------------------|--------------------------|
| 1      | Absent  | 10-36   | 10-27             | NRZ Perform better       |
| 2      | Present | 10-8    | 10 <sup>-16</sup> | Dicode perform<br>better |

Table 4.5 BER performance with respect to GVD

It is well known that GVD is unavoidable in practice and its effects are significant for a system operating at 10 Gbps and above. The results are tabulated in Table 4.5, it can be inferred that dicode coding performs better than NRZ. This development can lead to improvement in the sensitivity of the receiver along with power budget. Fig.4.9 shows the optical power obtained in the two different cases, one back-to-back (0 km) and another for a fiber length of 60 Km. In the case of a back-to-back the receiver sensitivity improvement by 0.7 dB, whereas 5.5 dB improvement was obtained for a fiber length of 60 Km.





Fig. 4.10 Fiber length vs Q factor.

The length of the fiber was varied in the range from 70 Km to 100 Km to obtain the suitable length matching the value of prescribed quality factor value as 4.10. As represented in Fig. 4.10, the maximum required length is 82 Km. Fig. 4.11 shows the clear eye opening at 82 Km for 64 user, depicts the quality of the signal received at the receiver. The signal can recovered satisfactory at required minimum bit error rate.



Fig. 4.11 Eye diagram for maximum length for 64 users for 82 KM.



Fig. 4.12 Fiber length vs Q factor for 128 users.



Fig. 4.13 Eye diagram for maximum length for 128 users for 72 KM

The Fig. 4.12 shows that there is considerable improvement in a number of users and length. The number of users extended to 128 and the maximum length that can be achieved is 72 Km as depicted in Fig. 4.12. The Fig. 4.13 is the graphical representation of Quality of signal for 128 users. Here, we have analyzed the effects of dicode coding on group velocity dispersion. The system with Dicode coding is observed to be more tolerant to GVD, and it can have power budget up to 5.5 dB with improved receiver sensitivity by 0.7 dB. Thus the improvement in power budget is helpful to improve the length and number of users in a system. The system showed the best performance for 128 users for length up to 72 Km. Thus a cost effective system can be designed with the parameters obtained.

#### **4.4 Chapter Summary**

Here, we have we analyzed the effects of balance duobinary and dicode coding in the presence of nonlinear effects. By simulation, we found that the balance duobinary coding is more affected by stimulus Raman scattering (SRS) by an amount of 0.5 dB as compared to dicode coding. So, dicode coding comparatively provide more tolerance to Raman cross talk in the video Overlay Passive Optical Network (PON). Further in this system 8.3 dB amplification was done by using Raman pump for the data signal to compensate the SRS effect and fiber loss for improving the performance of PON Network. Here we analyzed the spectral efficiency and reach enhancement in next generation passive optical network (NG-PON). The DPSK modulation format mitigates stimulated Raman scattering (SRS) effects in the fiber and boosts channel capacity with multi-giga-bit transmission rate. At 20 km fibre length, a power loss of 8-dB was observed which is compensated to 7-dB with pump power of 180mW. We also analyzed a system at 10 Gbps which has 128 and 32 users for 20 Km and 55 km respectively in permissible BER limit.

The DPSK modulation with interferometry detection technique provides 0.3 (bits/s)/Hz higher spectral efficiency in comparison to other modulation formats. The network system efficiency is upgraded by increasing the transmission bit rate, distance and capacity of the system to support maximum number of users or subscribers. Further investigations of non-linear effects, including non-linear phase noise on DPSK systems, are also feasible for longer transmission distance. This coding has main drawback of complexity of receiver circuits, so in this thesis we mainly focus on intensity

modulation. Here, we have analyzed the effects of dicode coding in a passive optical network based on both wavelength and time division multiplexing (WDM-TDM-PON). By numerical analysis, we observed that network using dicode coding is more tolerant to Group Velocity Dispersion (GVD) as compared to traditional NRZ-OOK coding. In the presence of dispersion, dicode coding has improvements of 0.7 dB in receiver sensitivity compared to conventional NRZ coding. It also improved the system power budget up to 5.5 dB. With the observations from analysis and results, use of Dicode coding doubles the number of users and increase the length of the network by 20 percent in respect to NRZ. Thus a cost effective system can be designed with the parameters obtained.

# **Design of High Capacity PON using Different Topology**

# 5.1 Introduction

With the Evolution of FTTH networks, the 'last mile' bottleneck problem has been broken and due to the exponential growth in the high bandwidth applications, the operators may face new challenges for upcoming bandwidth bottleneck in near future [8]. GPON supports theoretical bandwidth of 2.5 Gbps, however in practice; the effective offered bandwidth per user is limited up to 100 Mbps since it has to be divided among multiple users. The TDM PON is cost effective but it also sacrifices the bandwidth due to high split ratio and provide less security to the applications. Reach is also limited for TDM PON as splitter incorporate higher insertion losses. WDM-PON has emerged as a future proof solution for such high bandwidth applications.

Most of the service operators require highly bandwidth scalable, cost efficient, secured and long distance transmission to serve their users. To resolve the bandwidth, security and reach issue, WDM-PON becomes a most effective solution that transmits multiple wavelengths into the same fiber. Multiplexer multiplexes transmitted wavelengths at the central office and the receiver end demultiplexes by the demultiplexer. By using WDM-PON, high-speed access network services to dedicated business and mobile users can be provided. By providing a dedicated wavelength to each subscriber, service operators can scale the bandwidth requirement cost-effectively that is very essential for today and future services. In this chapter, we discuss about the various solution of WDM-PON.

# 5.2 2×40 Gbps WDM-PON Long Haul System

In this section, we discuss the transmission of high data rate signal. The high data rate signal limits the length of transmission system due to pulse broadening. This phenomena is known as dispersion. At such high bit rate, transmission of the signal is limited up to few kilometers. So the suitable dispersion compensation techniques can be used together with appropriate line coding. These types of networks can be used for long distance communication. Standard single mode fiber (SSMF) is used in the

network, that exhibit chromatic dispersion of the order of 15-17 ps/nm/km at 1550 nm. To compensate the chromatic dispersion, Dispersion-compensated fiber (DCF) [1,159], fiber brag grating [160, 161] and mid-span spectral inversion [108, 162] have been used. Time division multiplexed passive optical network (TDM-PON) and wavelength division multiplexed passive optical network (WDM-PON) are used to provide services in the passive optical network. To achieve large bandwidth and high capacity, WDM is the best technology in an optical communication system that also provides protocol transparency, easy upgradability and high quality of service [163]. The current existing PON network supports the maximum transmission length of 20 km with a data rate of 2.5 Gbps downstream and 1.25 Gbps upstream [163]. 10-Gigabit PON have evolved to support 10 Gbps data rate but not enough to fulfill the increasing future bandwidth demand [164],

In literature, a 40 Gbps or higher PON networks (Next Generation- PON) has been incorporated such as 2.5 Gbps 16 channels and 1.25 Gbps 32 channels [165-166], 4×10 Gbps TDM-PON up to 25 km without any dispersion compensation [35]. 40-Gbps single channel downstream/upstream PON using reduced modulation index [41] for limited reach is also proposed in due to chromatic dispersion. The performance of WDM-PON depends on several factors such as non-linearity, crosstalk, transmission loss, dispersion, etc. Transmission loss and non-linearity can be compensated and reduced by introducing suitable modulation scheme and Erbium-doped fiber amplifier (EDFA) [160]-[163]. To achieve high capacity and long distance, several techniques are used to mitigate dispersion such as DCF, FBG, and dispersion shifted fiber (DSF), etc.

DSF is not a suitable technique for WDM due to infeasibility of installing new fiber cable in the place of existing infrastructure [159]. So, DCF and FBG both are used to compensate dispersion. In WDM-PON, a lot of work on DCF have been done including 40 Gbps & 4×40 Gbps TDM/WDM transmission over 150 and 100 km respectively [167], NRZ & RZ modulation formats of 40 Gbps transmission over 1200 km & 400 km with DCF have also been evaluated [114]. DCF produces non-linearity at a longer distance due to high insertion loss with noise. Therefore, FBG is used with low insertion loss and large bandwidth to mitigate dispersion and crosstalk [158]. Previously, 10 Gbps single-channel transmission over 200 km [168] and 4×10 Gbps WDM channels transmission over 600 km [169] has been demonstrated for FBG. The comparative

analysis between DCF and FBG for 4×40 Gbps transmission over 600 km [170] and single channel 40 Gbps over 4000 km have also been analyzed, and the choice of modulation formats such as NRZ, RZ, CSRZ, Duobinary (DB), etc. [158] have been clarified. Here, we have demonstrated an 80 Gbps WDM-PON architecture using 2×40 Gbps WDM channels with NRZ, Dicode-NRZ, Duobinary schemes for FBG post compensation. The modulation schemes are also compared on the basis of their tolerance against the dispersion and transmission length. The modulation schemes are used to enhance the dispersion tolerance, distance, and performance of the system. Different types of modulation schemes such as NRZ, Dicode-NRZ & Duobinary have been discussed in section 3.3. The brief of these schemes is described in the following section.

NRZ is the simplest modulation scheme that can be used in WDM-PON system. Laser source switching is required to get this modulation. The modulation is performed by Mach-Zehnder modulator (MZM) which receives continuous wave (CW) as carrier and modulation signal from electrical filter. During the transmission of bit 1, signal appears at the receiver. While for bit 0, no signal appears [42]. Dicode coded NRZ is modified form of the NRZ level coding, doesn't have zero dc components, thus it is power efficient with more dispersion and nonlinearity tolerance. The overlapping of signals or interference can be mitigated by dicode-NRZ [48]. The binary signal requires differential encoder which converts binary signal into differential signal to get independent correlated signal level.

One logic gate (X-OR) and 1 bit delay is used in Dicode coder. More detail of Dicode generation including power spectra characteristics is given in the section 3.3.5.2. Receiver complexity, four wave mixing and scattering noise is minimized by Dicode-NRZ scheme. Duobinary code is also similar to Dicode code except its DC component and can be generated by using 1 bit delay and adder. The bandwidth of Duobinary is half as compared to NRZ [118].

#### 5.2.1 Simulation Setup

The proposed 80 Gbps WDM-PON system configuration is depicted in Fig. 5.1. Two 40 Gbps WDM channels with external modulator are used. For this proposed system 1 nm channel spacing is considered. In this set up, each OLT consists of a pseudo random

binary sequence (PRBS) generator, electrical pulse generator, continuous wave (CW) laser source and Mach Zender modulator (MZM) as shown in Fig. 5.2.



Fig. 5.1 Simulation setup of FBG post compensation.



Fig. 5.2 Block diagram of (a) OLT (b) ONU.

A PRBS with 64-bit stream is carried by each OLT to generate electrical waveforms such as NRZ, Dicode-NRZ, Duobinary schemes. For this, intensities of light sources (CW laser) are externally modulated by MZM modulator and electrical signal is converted in to optical signal. The bandwidth for NRZ and Dicode-NRZ is 2R while for Duobinary bandwidth is R, where R represents the data rate. The input power of laser is kept 0 dBm and both WDM channels are multiplexed and transmitted over SSMF. Fiber spans of 50 km of SSMF with dispersion 17 ps/nm/km and dispersion slope +0.075 ps/nm<sup>2</sup>/km is considered for this analyses. EDFA is used for compensating the losses occurring in the fiber. FBGs are also placed at the receiver end (post compensation) to

compensate overall link dispersion. The value of dispersion of FBG is negative and equals to -800 ps/nm that cancels the positive dispersion produced by SSMF. At the receiver, demultiplexer is used to separate the channels. At the ONU, PIN photo detector, converts optical signal into electrical signal and low pass filter (LPF) with bandwidth 0.75R and bit error rate (BER) analyzer to measure the performance of the systems as shown in Fig. 5.3. The performance of the signal is measured in terms of Q factor and BER.

#### 5.2.2 Results and Discussion

The various specific characteristics of modulation schemes affect the BER and Q factor with increasing length. The transmission performance of two 40 Gbps WDM channels for two wavelength 1550nm, 1551nm with FBG post compensation using NRZ, Dicode-NRZ, Duobinary scheme for varying lengths are simulated. In Fig. 5.3, the relation between BER and fiber length is evaluated at different length and for different modulation schemes.



Fig. 5.3 BER performance (a) 1550 nm & (b) 1551 nm



Fig. 5.4 Q factor performance (a) 1550 nm & (b) 1551 nm



Fig. 5.5 Eye pattern of Dicode-NRZ scheme with FBG for 2000 Km

Relation between Q factor and fiber length is given in Fig 5.4 shows the performance for different coding techniques and find that dicode NRZ can achieve the maximum length for required quality factor. Eye pattern at 2000km at 0 dBm power is shown in Fig. 5.5 which shows clear eye opening. With these observations we conclude that Dicode-NRZ can gives better performance as compared to other schemes. For different PRBS sequence length, we have analysed the receiver sensitivities for Dicode coding and found that as the PRBS sequence length increases, receiver sensitivities improves. For BER of 10<sup>-9</sup> receiver sensitivities for PRBS sequence 2<sup>7</sup>-1 and 2<sup>10</sup>-1 are found - 3.75dBm and -5dBm respectively as shown in Fig 5.6.



Fig. 5.6 PRBS dependency on dicode NRZ.

The NRZ, Dicode-NRZ and Duobinary schemes with 80 Gbps WDM-PON systems, is analyzed by using two 40 Gbps WDM channels using FBG and evaluated by numerical method. From the investigation and numerical results, Dicode-NRZ is superior compared to the other schemes which increases in length and reduces the system complexity. In future, FBG will mitigate the dispersion in WDM system with increased channels and length.

#### **5.3 Ring based WDM-PON Network**

In this section we have designed and analysed a ring-based wavelength-divisionmultiplexing passive optical network (WDM-PON), which provides fibre-fault protection by using dual ring architecture as well as Rayleigh backscattering (RB) noise mitigation. For protection and restoration against fibre fault, the dual fibre ring is used. This analysed the characteristics of the downstream signals using 10 Gbps ON–OFF keying (OOK), 10 Gbps dicode-coded NRZ. For the ultra-high speed networks, wavelength division multiplexing (WDM) technique is more efficient than that of the time sharing technique. Use of directly modulated lasers at central office (CO) [171,172] and Reflective Semiconductor Optical Amplifier (RSOA) at the Optical network Unit (ONU) for remodulation and reuse, seeded from the CO [173], is the most desired architecture for future ultra-high speed, high capacity WDM-PONs. However, the Rayleigh Backscattering (RB), which is generated by the downstream signal, is the major issue in the colourless WDM-PONs.

To mitigate the effect of RB, different techniques have been developed, such as using phase and bias-current dithering, utilizing wavelength-shifting technique, employing advanced modulation formats and double laser bands source, etc. [31, 38-39]. These techniques make network more complex, costly and mainly used in tree based WDM-PON. The effect of RB noise on PON is the major issue, which can be minimized by ring based WDM-PON. In the ring based WDM-PON, RB is generated in opposite direction of upstream signal that reduce the effect of RB noise [174]. Hence the ring based WDM-PON is regarded as one of the important architectures for next-generation PON , the architecture is shown in Fig. 5.7.



Fig. 5.7 Ring architecture WDM-PON with fibre-fault protection [173].

Next generation WDM-PON must provide higher data rate with high quality of service. If the fault in the fiber has occurred between CO and ONUs, then data traffic cannot reach affected ONU, which leads to loss of data. Hence, Protection management is the major issue in PON access [175, 176]. Fig. 5.7 represents dual ring architecture, which is one of the best solution for the fibre protection [174]. In the system architecture, dual fibre ring architecture has been used for the fibre protection in which each ONU can select its fibre path automatically. In this work, we have compared the system performance for the 10 Gbps downstream On-Off Keying (OOK) with dicode-coded NRZ signal. For upstream characterization 2.5 Gbps OOK has been used. Dicode-coded NRZ as discussed in section 3.3 dicode-coding is one of the simplest dc-balanced correlative level (CL) coding. Dc-balanced CL codes upshift the lower frequency components to upper frequency, which reduces the overlap between signal and interferer spectra.



Fig. 5.8 Power Spectra of dicode coded NRZ.

In section 3.3, dicode-coded NRZ is generated as shown in Fig. 3.18. The power spectra of dicode NRZ shown in Fig. 5.8. In order to treat the correlated signal levels independently in the decoding process, the original binary message needs to be differentially coded prior to the CL encoding. The differential encoder consists of two gates and one delay element while dicode coder consists of one gate and a delay element as shown in Fig. 3.18 Due to precoding that we observe in given example that there is 1:1 relation in dicode coded data and original binary data. Hence, 0 is mapped to 1 and -1 and 1 is mapped to 0, that reduces receiver complexity, and imposes no bandwidth redundancy unlike 8bl0b code [40] and Manchester code [177]. From Fig. 5.8, we can see that the dicode-coded NRZ has less baseband components as compared to uncoded NRZ that is major portion of lower frequency component has moved to higher frequency components. Since RB noise affects the dc component of the upstream signal by using dicode-coded signal, we can mitigate the effect of RB noise.

#### 5.3.1 Setup of dual ring WDM-PON

Fig. 5.9 shows the general architecture of the dual ring WDM-PON used in simulation setup. We have used three ONUs, at each ONU where there are three optical circulator, two in clockwise and one in anticlockwise, two FBGs for separation of data signal, and upstream continuous wave (CW) seeding light from multiplexed data stream . The data signal and CW seeding signal received at ONU is separated by filter and passed to receiver and RSOA, respectively. The seeding light is used for generation of upstream signal by remodulating it using RSOA.



Fig. 5.9 Simulation Setup of dual ring WDM-PON.

In simulation each ONU is placed at 5 km distance from others, that is we have used a 5 Km SMF to connect ONUs with each other and to CO. A CW laser of wavelength 1550.6nm, 1552.6nm, 1554.6nm is used for seeding RSOA and the data is transmitted at 10 Gbps at wavelengths 1512.2nm, 1522.2nm, 1532.2nm for ONU1, ONU2and ONU3, respectively. We have used a PRBS sequence of length 2<sup>7</sup> -1 as data. The upstream signal of 2.5 Gbps is remodulated using OOK at ONU.

#### 5.3.2 Results and Discussion

In the numerical analysis, we are observing the performance of OOK and dicode-coded NRZ modulation and its effect on the system parameters. The input data is modulated with a 2<sup>7</sup>-1 PRBS using OOK, and its BER and the Q factor at receiver of all three ONUs are noted as mentioned in the Table 5.1. Now by the same procedure we have analyzed the effect of the dicode -coded NRZ and listed the parameters in Table 1. We can observe from the results that in our level coding techniques, the dicode-coded NRZ is superior. System BER is improved approximately 10 times than that of the OOK. It is also observed that Q factor for dicode-coded NRZ is better than OOK.

To analyze the effect of fiber length on BER and Q factor in optical system we performed simulation using a CO and a single ONU. In which the single CW seeding light with wavelength 1550.6 nm and a data at wavelength 1512.2 nm is used. Here, we are varying the length of the SMF which is connected between CO and ONU to carry out our simulation work. 15 Km, 20 Km, 25 Km and 30 Km SMF fibers have been used to observe the effect.

| Sr. NO | User No. | NRZ                      |          | Dicode-Coded NRZ         |          |
|--------|----------|--------------------------|----------|--------------------------|----------|
|        |          | BER                      | Q factor | BER                      | Q factor |
| 1      | ONU 1    | 9.48 x 10 <sup>-11</sup> | 6.39     | 8.37 x 10 <sup>-12</sup> | 6.72     |
| 2      | ONU 2    | 4.66 x 10 <sup>-8</sup>  | 5.33     | 4.55 x 10 <sup>-9</sup>  | 5.74     |
| 3      | ONU 3    | 1.41 x 10 <sup>-7</sup>  | 5.13     | 2.27 x 10 <sup>-8</sup>  | 5.46     |

Table 5.1 BER and Q Factor measurements for NRZ and Dicode at -10dBm

The Table 5.2 lists the observation taken from a single ONU configuration at -4dBm power. From Table 2 and Fig. 5.10, we can observe that the dicode-coded NRZ is better than NRZ even at higher distances.

| Sr. NO | Length<br>(Km) | NRZ                      |          | Dicode-Coded NRZ         |          |
|--------|----------------|--------------------------|----------|--------------------------|----------|
|        |                | BER                      | Q factor | BER                      | Q factor |
| 1      | 15             | 1.02 x 10 <sup>-36</sup> | 12.6     | 2.00 x 10 <sup>-40</sup> | 13.25    |
| 2      | 20             | 2.63 x 10 <sup>-24</sup> | 10.1     | 9.90 x 10 <sup>-27</sup> | 10.63    |
| 3      | 25             | 4.00 x 10 <sup>-16</sup> | 8.09     | 8.13 x 10 <sup>-18</sup> | 8.51     |
| 4      | 30             | 4.30 x 10 <sup>-11</sup> | 6.48     | 4.94 x 10 <sup>-12</sup> | 6.8      |

Table 5.2 BER and Q Factor measurements for NRZ and Dicode at -4 dBm



Fig. 5.10 BER performance of NRZ and Dicode-coded NRZ.

In this previous section, we have analyzed and simulated a ring-based WDM-PON providing both RB noise mitigation and fiber-fault protection. We analyzed the characteristics of downstream signal using 10 Gbps OOK and dicode-coded NRZ, and it is observed that there is significant improvement in BER and Q factor while using dicode-coded NRZ. Cost and complexity of receiver at the user side is less as compared to other coding techniques such as DPSK, OFDM.

## 5.4 Design Hybrid WDM-PON

In this section, we have compared the performance of the hybrid passive optical network (H-PON) for different coding techniques such as Manchester code, and Dicode coded Non-Return to Zero (NRZ). The above stated two codes are employed for

comparison since both codes are multilevel codes and suppress the low-frequency component in their spectrum. The better BER performance of Dicode coded NRZ gives us the opportunity to extend the reach of the system. The increase in demand for bandwidth in broadband communication has forced us to analyze our system performance by using data signal along with video signal. Due to this, we need to adopt a cost effective system which can provide high data rates. The hybrid WDM-PON is proposed [13, 178] to support such system. In Manchester coding, the dc component is absent and does not require any external clock signal due to self-clocking which helps to save power [177]. Similarly, Dicode coded NRZ is modified form of the NRZ level coding, have reduced zero dc components. Thus it is also a power efficient with more dispersion tolerance capacity [179, 180].

#### 5.4.1 Hybrid WDM-PON Setup

Simulation setup in Fig. 5.11 depicts the hybrid TDM-WDM-PON. This system is used for doublet services such as data and video transmission. This simulation architecture uses both arrayed waveguide grating for WDM multiplexing and a power splitter for TDM multiplexing. Since the 1X 16 WDM Demux and 1X 8, TDM power splitter is employed hence this hybrid system can provide the service to the 128 users. This hybrid system is capable of using the advantages of both WDM and TDM technology. At the optical line transmitter (OLT), we are transmitting data and video signals simultaneously by modulating with the help of much-Zehnder modulator to get decoded NRZ signal for real-time applications. A 10 Gbps downstream signal is transmitted to 20 km fiber without any repeater and dispersion compensation techniques after 20 Km fiber the signal is separated by arrayed waveguide grating.



Fig. 5.11 Simulation Setup of HPON.

These different wavelength signals are applied to power splitter to increases the capacity of the system in terms of the users. So in this case, 8 users are sharing the same wavelength, and total number of effective users is 128. At the optical networking unit (ONU), Bessel's filter is used to suppress the unwanted noise and the filtered output is detected by PIN diode. This received signal is analyzed with the help of BER / Eye analyzer to measure the bit error rate and Q factor.

#### 5.4.2 Results and Discussion

The present work gives the simulation results of hybrid PON system for data and video services for different modulation formats such as Manchester and Dicode-coded NRZ signal. The BER and Q factor for different distance for different level coding has been compared to observe possible transmission distance for optical network.

| S. No | Length(Km) | Q Factor | BER                    |
|-------|------------|----------|------------------------|
| 1     | 5          | 6.65     | 2.62*10 <sup>-11</sup> |
| 2     | 10         | 5.9      | 1.04*10-9              |
| 3     | 15         | 5.5      | 2.74*10-7              |
| 4     | 20         | 4.12     | 1.74*10 <sup>-5</sup>  |

Table 5.3 BER and Q Factor measurement for Manchester coding

Table 5.4 BER and Q factor measurement of Dicode-coded NRZ

| S. No | Length (km) | Q Factor | BER                    |
|-------|-------------|----------|------------------------|
| 1     | 10          | 8.29     | 3.92*10 <sup>-17</sup> |
| 2     | 20          | 7.63     | 8.70*10- <sup>15</sup> |
| 3     | 30          | 6.67     | 1.59*10 <sup>-11</sup> |
| 4     | 40          | 5.89     | 1.76*10 <sup>-9</sup>  |



Fig. 5.12 BER Performance of Manchester and Dicode coded NRZ.



Fig. 5.13 Eye diagram at 40 km for Dicode-coded NRZ.

For Manchester coding Bit error rate and Q factor is measured by varying different length of fiber i.e. 5, 10, 15, 20 Km. We observed the permissible values of BER & Q factor at 10 Km length as shown in Table 5. 3. Similarly Table 5.4 indicates the value of Q factor and BER for Dicode coded NRZ for different length of fiber & works satisfactory up to 40 Km. Fig. 5.12 shows the graphical representation how the BER vary with length of fiber with different level coding. Fig. 5.13 shows the eye diagram of Dicode coded NRZ at 40 Km, which gives clear eye opening and good quality of signal transmission. In the above section, the performance of hybrid passive optical network with 128 users is observed with different level coding. The simulation results

shows that for fixed input power of 10 dBm and data rate 10 Gbps, Dicode coded NRZ maximizes the transmission distance up to 40 km without any repeater and dispersion compensation.

#### **5.5 Chapter Summary**

The effects of reshaped spectra of different schemes such as NRZ (Non return to zero), Dicode-NRZ and Duobinary on Wavelength division multiplexing have been analyzed for two wavelength 1550 & 1551 nm with the system capacity of 80 Gbps. For a system with the capacity of 80 Gbps is achieved by wavelength division multiplexing of two 40 Gbps channels and fixed fiber length of 50 km/span. Dispersion is mitigated by fiber Bragg grating (FBG) with post compensation and nonlinear effects are reduced using Dicode. We evaluated the performance of three schemes in terms of bit error rate (BER) and Q factor. Numerical simulation results show that Dicode-NRZ provides better tolerance towards dispersion and nonlinearities compared to others and can transmit signal up to 2000 km. We have analyzed a ring-based wavelength-division-multiplexing passive optical network (WDM-PON) which provides Rayleigh backscattering (RB) noise mitigation as well as fiber-fault protection. For protection and restoration against fiber fault, the dual fiber ring is used. Here, we discussed and analyzed the characteristics of the downstream signals using 10 Gbps ON–OFF keying (OOK), 10 Gbps dicode-coded NRZ.

It is observed that there is significant improvement in BER and Q factor while using dicode-coded NRZ. Cost and complexity of receiver at the user side is less as compared to other coding techniques such as DPSK, OFDM. In the above section, the performance of the hybrid wavelength division multiplexing passive optical network (H-PON) is observed for different coding techniques such as Manchester code and Dicode coded Non Return to Zero (NRZ). The above stated two codes are employed for comparison since both suppress the low frequency component in their spectrum. The simulation results shows bit Error rates of  $8.7 \times 10^{-15}$  for fixed input power of 10 dBm, data rate 10 Gbps, Dicode coded NRZ maximizes the transmission distance up to 20 km without any repeater and dispersion compensation. Mean the better BER performance of Dicode coded NRZ gives us opportunity to extend the reach of the system up to 40 KM.

# Chapter 6

# **Enhancement of Links Performance by Incorporating EDFA and Dicode coding**

## **6.1 Introduction**

The everyday evolution and updates in technologies, the triple play, and other related services have a quench of higher bandwidth [10]. The wavelength division multiplexing (WDM) based network offers high transmission capabilities as several wavelengths can be transmitted through a single fiber [181]. In a conventional system, the transmitted wavelengths are multiplexed at the central office (CO) and are separated at the remote node (RN) with the help of arrayed waveguide grating [37]. WDM-PON is one of the most promising solutions for the next-generation broad-band access network to meet out the bandwidth requirement. PON uses point to multipoint architecture, which is considered as the best economic approach for fiber to the home solutions. The performance reaches a limit due to inter-symbol interference, reflection noise, spectral efficiency, dispersion, nonlinearity and Raman cross-talk [182, 183].

The cost of infrastructure and equipment are among major challenges in WDM-PON deployment. The laser source for each wavelength used at the optical line terminal is the main contributor of the costly set up [28, 57]. The challenge of high cost for set up can be controlled by Flexible hybrid PON, which can be implemented with the existing system [87]. It can be designed by using WDM then TDM, hence by utilizing the advantages of both, the set of wavelengths are shared by various optical network units (ONUs) at the same time instead of dedicated wavelength as in WDM-PON [183]. It offers low cost, sharing of single wavelength by multiple users with a limited channel bandwidth per user [183], which is compensated by using the high bit rate signal. The multiple wavelengths in same fiber can lead to an increase in the concentration of a low-frequency component at Remote Node (RN), thus limiting the power budget of the system due to noise, affecting the low frequency the most [151].

A dicode coding is a spectral reshaped spectrum which has reduced low-frequency spectra that are least affected in comparison to other coding systems, offering improved power budget in such type of networks [151]. The feasibility of hybrid PON with a different number of users is verified by reports of R. S Kaler et al. [13]. A system at 1.25 Gbps data rate hybrid WDM- PON system with triple play services i.e. data, voice and video for 128 users for different modulation schemes such as NRZ, RZ (return to zero) and Manchester was analysed by his team. The results express that NRZ scheme is suitable for transmitting information to all ONUs for up to 28 Km without using any repeater in the link. R. S. Penze et al. [146] proposed a PON extender using the power over fiber techniques which helps to eliminate the batteries at the remote node.

A set up with 32 split and 50 Km reach for XGPON/ GPON system is analysed with PON extender. A 10 G– EPON system is proposed by T. Tsutsumi et al. [155], the link budget of the access network is increased by using semiconductor optical amplifier (SOA). The setup is reported to be functional upto 41.5 Km length for 128 users together with N:1 optical subscriber unit (OSU) protection to improve reliability against OSU failure. Dicode coding in WDM-PON architecture is proposed by Q. Guo et al. [184] with aided improvement in the reach of the network as compared to NRZ coding. The proposed architecture is costly due to the allocation of dedicated wavelength to each user and increases the capacity of each user. It is the waste of bandwidth due to the lack of flexibility [87].

In this chapter, we propose a flexible Hybrid (WDM-TDM) PON using dicode coding that utilizes traffic rate of 40 Gbps as a downstream for 128 users up to 83 KM without using dispersion compensation techniques. The system depicts an improvement in the receiver sensitivity with dicode coding, which helps in increasing the number of users.

#### 6.2 Importance of flexible PON

In WDM-PON network, multiple wavelengths are transmitted from the transmitter into the same fiber. At the remote node, these wavelengths are demultiplexed, separated and provided to the end users. Normally, a set of multiple 10Gbps signals is used i.e. each user is getting an individual 10Gbps wavelength. However in practice, below 1Gbps data rates are required by each user. Therefore in WDM technology, most of the bandwidth is wasted. By using flexible PON architecture, the utilization of bandwidth can be increased. For this, rather than using pure WDM technology, a combination of WDM and TDM must be utilized. TDM PON enables the sharing of one 10Gbps signal to the multiple users and therefore increases the bandwidth utilization.

# 6.3 Hybrid WDM-PON with Dicode Coded None Return to Zero (NRZ)

The performance of the system is a function of characteristics of the modulation schemes employed [185]. The numerous significant applications and developments inspired us to study the different Line codes. These line codes are selected by some dominant consideration like timing, dc content, power Spectrum, performance monitoring as well as transparency. In this work, the property of line coding with reduced dc content in the power spectrum has been deployed. The power spectrum should be such that the maximum energy is contained in a small bandwidth to achieve high transmission efficiency [185].

Dicode is a special spectrum reshaping technique to reduce the low-frequency dc component compared with the conventional NRZ coding which will improve noise performance as well as nonlinearity. It is correlative coding which has the advantage to permit some amount of interference at the receiver end to reproduce the original message. The block diagram of dicode encoder is discussed and shown in Fig. 3.18. The effect of spectra on the system performance due to channel impairments such as dispersion is analyzed. The bandwidth required for the critical spectral content at the receiver end is determined by the waveform spectra.

**Hybrid WDM-PON**: - The Equipment cost is the main obstacle regarding WDM- PON technology (Need to have one laser at the OLT for each wavelength) [82]. Hybrid PON is the compromise of the WDM-PON and time division multiplexing (TDM-PON) network & use the current existing TDM system [45]. Efficiency of these network resources is enhanced by this architecture [87].

Hybrid WDM-TDM-PON classify mainly are two types [87]:-

- Broadcast and select (B&S) PON architecture
- ➢ A wavelength splitting (WS) PON


Fig. 6.1 Flexible hybrid PON.

In Fig. 6.1, if at RN1, the splitter is used, then it is broadcast and select (B&S) PON architecture.

Limitation:-

- > It has severe security issues and provides no data isolation.
- > It has poor power budget and therefore restricts the overall reach.

RN1 as an AWG or WDM, it is a wavelength splitting (WS) PON

Significantly improves the data security and the power budget, it reduces the overall flexibility available in a B&S-PON.

Now we consider a setup with RN1 as a WDM and RN2 as a splitter so it can provide moderate security as well as flexibility.

**6.3.1 Flexible Hybrid PON Setup:** A flexible hybrid PON is the cost-effective approach that can be used to fulfill the future bandwidth demand of subscriber with low cost. In our work, N is considered 4 and M are variable from 16 to 256 as shown in Fig. 6.2.



Fig. 6.2 (a) A flexible hybrid PON system (b) Optical network unit (ONU).

The block diagram of flexible hybrid PON is depicted in Fig 6.2 (a), in which, the central office consists of an optical source, modulator, electrical signal generator and a Mux/ Demux unit. A continuous wave (CW) laser of 0 dBm power, 10 MHz line width generates four frequencies at 193.1 THz, 193.2 THz, 193.3 THz and 193.4 THz with 100 GHz spacing light pulse as a downstream transmission. A PRBS generator generated a 10 Gbps binary data sequence and converted into electrical signal by using NRZ / dicode signal. This signal is converted into optical signal by a push-pull operation of Lithium Niobate (LiNbO<sub>3</sub>) Mach-Zehnder Intensity Modulator (MZM)

having an extinction ratio of up to 30 dB. These modulated signals are multiplexed by  $4\times1$  WDM multiplexer and transmitted through the optical fiber. Attenuation and dispersion losses of fiber are assumed to be 0.2 dB and 16.75 ps / nm-km respectively. At the other end of the remote node, an erbium-doped fiber amplifier (EDFA), a 1×4 Demux (RN1) and power splitters (RN2) are used. EDFA of 6 dB Noise figure and 22 dB gain, amplifies the received optical signal, however it is the main source of amplified spontaneous emission (ASE) noise, and this ASE noise is affected by the power spectra of the signal. These amplified signals get demultiplexed and further split at RN1 and RN2 respectively. The splitting ratio of RN2 can be varied from 1×2 to 1×256.

Splitting loss of the splitter depends on 3.5log<sub>2</sub>M. RN1 and RN2 can be employed for two-way, i.e., splitter- splitter / Demux- splitter to achieve the flexibility of the system [6] as per bandwidth requirement. The Splitter – Splitter combination reduces power and available bandwidth for each user. It also reduces the total fiber length. In our work, we are using the combination of RN to meet the future demand for bandwidth and a large number of users. As shown in Fig 6.2 (b) Optical Network Unit (ONU), received optical signal is converted into electrical signal by using PIN Photodetector having a responsivity of 1 A/W and dark current 10 nA. This electrical signal is filtered by low pass filter (LPF) to reduce noise. The performance of received signal is analyzed by Bit Error Rate (BER) Analyzer.

Upstream transmission is also possible for the proposed architecture by using 198.1 THz, 198.2 THz, 198.3 THz, 198.4 THz frequency band. The signal coming from Remote Node is amplified by EDFA and enters into circulator via bidirectional optical fiber and directed to the OLT demux. Here all upstream signals are separated and received by the respective receiver.

#### 6.3.2 Results and Discussion

 a) On/Off Extinction Analysis: - The impact of the high extinction ratio (ER) on the performance of the Hybrid PON system as shown in Fig. 6.2 for Dicode coded NRZ is presented below.

| <i>S. No.</i> | Extinction ratio(dB) | Dicode (Q) |
|---------------|----------------------|------------|
| 1             | 5                    | 2.32       |
| 2             | 10                   | 2.67       |
| 3             | 15                   | 4.1        |
| 4             | 20                   | 5.52       |
| 5             | 25                   | 5.97       |
| 6             | 30                   | 6.26       |

Table 6.1 Extinction ratio analysis

Table 6.1 shows that moving towards the higher extinction ratio (ER), performance improves regarding Quality factor (Q) and Bit Error rate. At 30 dB extinction ratio achieved Quality factor is 6.26 for 128 users at 67 KM fiber length.

b) Effects of thermal power density (receiver sensitivity):-

Receiver is responsible for performance of system as change in minimum receiver sensitivity causes change in required power budget. Thus, we analysed the effect of coding in the presence of highly sensitivity detector.

| S. No. | Thermal Power density<br>(W/ Hz) at Receiver | Dicode (Q) | NRZ(Q) |
|--------|--|------------|--------|
| 1      | 100×10 <sup>-23</sup>                        | 3.23       | 3.22   |
| 2      | 100×10 <sup>-24</sup>                        | 6.39       | 6.25   |
| 3      | 100×10 <sup>-25</sup>                        | 8.82       | 6.20   |
| 4      | 100×10 <sup>-26</sup>                        | 9.71       | 6.31   |
| 5      | 100×10 <sup>-27</sup>                        | 9.92       | 6.31   |

Table 6.2 Effects in the presence of thermal power density.

Table 6.2 shows that when thermal power density is reduced, receiver sensitivity improves. Dicode shows significant improvements in system performance, but there is no significant change in the case of NRZ coding. The value of Thermal Power density at photodetector is  $100 \times 10^{-24}$  W/Hz for our work.

c) Effects of Group velocity dispersion (GVD):-

| S. No. | GVD     | NRZ(Q) | Dicode (Q) | Remarks                |
|--------|---------|--------|------------|------------------------|
| 1.     | Absent  | 23.38  | 22.20      | NRZ performs better    |
| 2.     | Present | 6.22   | 9.66       | Dicode performs better |

Table 6.3 GVD Analysis at 10 Gbps data rate.

It is shown in Table 6.3 that when we consider GVD effects in Single mode optical fiber with length of 67 KM, dicode will perform better than NRZ. In the absence of GVD, NRZ will perform better than Dicode. But as we know that GVD effects cannot be negligible in standard single mode fiber for practical communication system, it follow that dicode is the best choice in case of high data rate system at 10 Gbps and above. A flexible hybrid PON system is designed as shown in Fig. 6.2, considering the 30 dB extinction ratio and  $100 \times 10^{-24}$  W/Hz thermal Power density.

The performance of the system is observed at the different lengths and number of users by using different coding viz. NRZ and Dicode. For designing of a system, it is important to have required power budget so it is ensured that enough power will reach to the receiver to maintain reliable performance during the entire lifetime of the system. Fig. 6.3 shows the BER vs optical received power at 10 Gbps for NRZ and dicode for with and without fiber. We can mathematically express the power budget and losses due to various components used in passive optical access network as,

Available Optical power Budget

= Input power - Optical receiver sensitivity + amplifier gain (if used)(2)

Minimum Receiver sensitivity can be obtained by Fig. 6.3 for NRZ and dicode is respectively -32.60 dBm and -36.20 dBm.



Fig. 6.3 Optical received power vs BER

For NRZ

= 0-(-32.60) + 22 dB = 54.60 dB

For Dicode

= 0-(-36.20)) + 22 dB = 58.20 dB

The available optical power budget is calculated for NRZ and dicode is 54.60 dB and 58.20 dB respectively.

As shown in Table 6.4, NRZ only satisfies the power budget for 64 users at 67 KM, and for other, it's not suitable, but dicode will perform satisfactorily in all the three case. Fig. 6.3 shows that the received optical power obtained for BER @ 10<sup>-9</sup> in the NRZ and Dicode coding in the back-to-back (0 km) is -37.50 dBm and -38.20 dBm respectively. Hence dicode provides 0.7 dBm of improvement in receiver sensitivity. At 67 Km fiber length, received optical power at BER 10<sup>-9</sup> is -32.60 and -36.20 dBm respectively, for NRZ and dicode. So dicode provides 3.6 dB improvements in power budget as compared to NRZ coding. This improvement in power budget could be used to increase the number of users.

|                           | 64 user at 67 Km                    | 128 user at 67 km        | 64 user at 77Km                           |
|---------------------------|-------------------------------------|--------------------------|---|
| Min Launch Power<br>(dBm) | 0                                   | 0                        | 0   |
| MZM insertion loss        | 5                                   | 5                        | 5   |
| SSMF Attenuation<br>(dB)  | 13.4<br>(0.2dB/km over 67km of SMF) |                          | 15.4 dB<br>(0.2dB/km over<br>77km of SMF) |
| Coupler/Mux/<br>Demux(dB) | (5+5)                               | (5+5)                    | (5+5)                                     |
| Optical Splitter(dB)      | 21                                  | 24.5                     | 21  |
| Other loss(dB)            | 5                                   | 5                        | 5   |
| Total System loss         | 54.4 dB                             | 57.9                     | 56.4                                      |
| NRZ (power budget )       | (54.6-54.4) =<br>0.2 dB             | -ve                      | -ve                                       |
| Dicode (Power<br>Budget)  | (58.20-54.4) =<br>3.80 dB           | (58.20-57.9)=<br>0.30 dB | (58.20-56.4) =<br>1.80dB                  |

 Table 6.4 Power Budget Analysis

### Power penalty analysis: -

Power penalty for dicode is  $(-36.20 - (-38.20)) = 2 \text{ dB} @ 10^{-9}$ For NRZ is  $(-32.60 - (-37.5)) = 4.90 \text{ dB} @ 10^{-9}$ 

Dicode's power penalty due to fiber is less than NRZ by 2.9 dB. Fig. 6.4 shows the performance of the system when varying the fiber length for 64 users. It is found that up to 67 KM length both coding provides good quality of signal but after that NRZ signal degrades severely and dicode provides good quality of signal up to 77 KM after that signal starts to degrade.



Fig. 6.4 Length v/s Q at 10 Gbps data rate.



Fig. 6.5 Q factor analysis with length and number of users.

Quality factor of the system with varying number of users at fixed fiber length is evaluated in Fig. 6.4 and the minimum required Q for system can be found for different coding schemes. As analyzed in Fig. 6.5, minimum required Q can achieved for NRZ at 67 KM up to 64 user and further increasing the length of system shows the low quality of signal irrespective of a number of the user. In the case of dicode, it gives the quality of signal at 67 KM up to 128 user, 77 KM up to 64 user and 83 KM for 16 users. Fig. 6.6 shows the maximum number of user at fiber length 67 KM and more. Possible maximum number of users at 67 km is 64 for NRZ and 67 KM, 77 KM and 83KM as 128, 64 and 16 users respectively for dicode.



Fig. 6.6 Length vs maximum No. of Users.

Fig. 6.7 gives the comparative graphical analysis for received signal. Further good eye opening as depicted in Fig. 6.7 (a) and Fig. 6.8 (b) ensure the quality of signal achievable for 64 users at 77 Km and 128 users at 67 km respectively for dicode coding.



Fig. 6.7 Eye diagram for 64 users at 77 Km Dicode coding.



Fig. 6.8 Eye diagram for 128 user at 67 KM for dicode.

ASE, thermal and shot noise in EDFA and PIN detector will affect NRZ signal due to presence of dc and low frequency spectra as compare to dicode coding. Therefore the use of dicode coding results in an improvement of 3.6 dB power budget which aids to improve the length of fiber and number of the users.

#### 6.3.3 Chapter Summery

In this work, we have analyzed the performance of flexible hybrid WDM-PONs with NRZ and dicode coding with 128×4 user at 67 KM. Further, we have also calculated the BER, Q factor and received optical power to observe the improvements in receiver sensitivity and power penalty as 0.7 dB and 2.9 dB respectively for the dicode. Hence, extra power budget of 3.6 dB for the dicode can increase the length up to 77 Km for 64 users or double the number of user at 67 Km compared to conventional NRZ coding. The possible increase in number of users helps to minimize the cost per user of the system.

### Chapter 7

## **Conclusion and Future trends**

We have discussed the background of optical fiber communication and passive optical networks in detail in chapter 2. We have also studied the merits and demerits of various PON generations like APON, BPON, GPON, next generation PON etc. Maturity, cost and power consumption of PONs also have been studied and observed the requirements to improve the network performance and various channel impairments like dispersion, attenuation, nonlinearity, etc. Chapter 3 described the different line coding such as NRZ, RZ, Manchester, DPSK, with their characteristics to make them suitable for the different applications and developments. To transmit the information signal into the fiber channel, different optical modulation schemes including OOK and DPSK have been studied.

Moreover, different modulations formats such as NRZ-OOK, RZ-OOK, DPSK, dicode, and duobinary have studied in detail. We found that dicode coding is more resilient to dispersion and nonlinearity as compared to other coding and therefore we utilized it for different designs described in upcoming chapters. Chapter 4 deals with the Raman crosstalk. Here, we investigated its effect on multilevel coding and observed that stimulus Raman scattering is more dominant in Duobinary than the dicode coding and the dicode coding gain is found to be 0.5 dB approximately. Raman scattering and fiber losses are compensated by amplifying the downstream data signal through remote Raman pump.

Also, The DPSK modulated distributed Raman amplifier in WDM passive optical network system is designed and analyzed parametrically. WDM-PON system provided the 55 Km transmission reach for upto 32 user and 0.3 bits/s/Hz higher spectral efficiency with 10 Gbps transmission speed. The network system efficiency is upgraded by increasing the transmission bit rate, distance and capacity of the system to support maximum number of users or subscribers. There are many more interesting challenges that remain for the future work. One can use bi-directional Raman pumping configuration for Raman amplification under the effect of temperature variation along the fiber cable. Here, we also analyzed the effects of dicode coding on GVD. Dicode

coding was found to be more tolerant to GVD and observed the power budget and receiver sensitivity improvement of up to 5.5 dB and 0.7 dB respectively. The improved power budget can be used to increase the fiber length or a number of users. The system showed the best performance for up to 128 users, and 72 Km reach. Thus a cost effective system can be designed with these obtained parameters. High bandwidth WDM-PON system and ring WDM-PON has been designed in chapter 5. We designed a high bit rate system using FBG that supports long haul networks. We compared the performance at 80Gbps using NRZ, Dicode-NRZ and Duobinary schemes and found Dicode-NRZ is more tolerant to dispersion, which can be used to increase fiber length and reduce system complexity.

A ring-based WDM-PON is also designed which provide RB noise mitigation as well as fiber-fault protection. Here, we analyzed the characteristics of downstream signal using 10 Gbps OOK and dicode-coded NRZ and found a significant improvement in BER and Q factor using dicode-coded NRZ. The performance of the hybrid passive optical network with 128 users was also observed for different level coding. For 10 dBm input power and 40km fiber distance, transmission using dicoded coded NRZ are achieved successfully without any repeater and dispersion compensation. In Chapter 6, the performance of flexible hybrid WDM-PONs with NRZ and dicode coding has been analyzed for 128×4 user and 67 km fiber reach. We observed the improvements in receiver sensitivity and a power penalty of 0.7 dB and 2.9 dB respectively for the dicode coding. Therefore total power budget up to 3.6dB is improved that can be used to further increase the length or users.

The increment in the number of supported users for the same system reduced effective cost per user. Designed system can be used to provide flexibility and future bandwidth demands. In near future we would like to extend our work by implementing the proposed line coding in wireless network integrated with optical networks. We will also design some new line coding by integrating some existing ones to improve the performance of passive optical network further. We would also like to focus on energy efficiency of the network, which will definitely reduce the emission of carbon foot print to design the environmental friendly network. Also, identified line coding will guarantee to support maximum users for the long run within the available energy resources.

## **Bibliography**

- 1. J. M. Senior, "Optical Fiber Communications," Prentice-Hall of India, 3rd edition, 2009.
- W. S. Huxford and J. R. Platt, "Survey of near infra-red communication systems," J. Opt. Soc. Am., 38, pp. 253–268, 1948.
- N. C. Beese, "Light sources for optical communication," Infrared Phys., 1, pp. 5– 16, 1961.
- 4. G. Keiser, "Optical Fiber Communications," McGraw-Hill, 5<sup>th</sup> edition, 2013.
- 5. G. P. Agrawal, "Fiber-Optic Communications Systems," 3<sup>rd</sup> edition, Wiley, 2002.
- T. Koonen, "Fiber to the Home/Fiber to the Premises: What, Where, and When?," Proceedings of the IEEE, vol. 94, no. 5, pp. 911-934, 2006.
- M. N. Tipán, G. V. Arévalo, I. N. Cano and F. Amaya-Fernández, "CATV, LTE and data convergence test in optical access networks," 2016 8th IEEE Latin-American Conference on Communications (LATINCOM), Medellin, pp. 1-3, 2016.
- 8. D. Kettler, H. Kafka, and D. Spears, "Driving fiber to the home," IEEE Communication Magazine, vol. 38, no. 11, pp. 106–110, 2000.
- V. B. Lawrence, L. J. Smithwick, J. J. Werner and N. A. Zervos, "Broadband access to the home on copper," Bell Labs Technical Journal, vol. 1, no. 1, pp. 100-114, 1996.
- S.J. Park, C. H. Lee, K. T. Jeong, H. –J. Park, J.G. Ahn and K. H. Song, "Fiber-tothe-home services based on wavelength-division-multiplexing passive optical network," Journal of Lightwave Technology, vol. 22, no. 11, pp. 2582-2591, 2004.
- T. Rokkas, D. Katsianis and D. Varoutas, "Techno-economic Evaluation of FTTC/VDSL and FTTH Roll-Out Scenarios: Discounted Cash Flows and Real Option Valuation," IEEE/OSA Journal of Optical Communications and Networking, vol. 2, no. 9, pp. 760-772, 2010.
- S. Verbrugge et al., "Research approach towards the profitability of future FTTH business models," 2011 Future Network & Mobile Summit, Warsaw, pp. 1-10, 2011.

- R. Goyal and R.S. Kaler, "A novel architecture of hybrid (WDM/TDM) passive optical networks with suitable modulation format," Optical Fiber Technology., vol. 18, pp. 518-522, 2012.
- 14. M. K. Weldon, T. Van Landegem, and E. S. Szurkowski, "Next-generation access networks: A preview," Bell Labs Technical Journal, vol. 13, no. 1, pp. 1-10, 2008.
- J. Wang, C. Qiao, Y. Li and K. Lu, "On guaranteed VoD services in next generation optical access networks," IEEE Journal on Selected Areas in Communications, vol. 28, no. 6, pp. 875-888, 2010.
- J. R. Stern, J. W. Ballance, D. W. Faulkner, S. Hornung and D. B. Payne, "Passive Optical Local Networks for Telephony Applications and Beyond," Electronics Letters, vol. 23, pp. 1255-1257, 1987.
- ITU-T Recommendation G983.1, "Broadband Optical Access System Based on Passive Optical Networks," 2001.
- F. Effenberger et al., "An introduction to PON technologies [Topics in Optical Communications]," IEEE Communications Magazine, vol. 45, no. 3, pp. S17-S25, 2007.
- 19. ITU-T Recommendation G983.4, "A broadband optical access system with increased service capability using dynamic bandwidth assignment," 2001.
- J. Kani, F.Bourgart, A.Gui, A. Rafel, M.Campbell, R.Davey, and S. Rodrigues, Next-generation PON—Part I: Technology roadmap and general requirements," IEEE Commun. Mag., vol. 47, no. 11, pp. 43–49, 2011.
- L. G. Kazovsky, W. T. Shaw, D. Gutierrez, N. Cheng and S. W. Wong, "Next-Generation Optical Access Networks," Journal of Lightwave Technology, vol. 25, no. 11, pp. 3428-3442, 2007.
- 22. Donald. E.A. Clarke and Tetsuya Kanada, "Broadband: The Last Mile," IEEE Communications Magazine, Vol.31, pp. 94-100, 1993.
- C. H. Lee, W. V. Sorin and B. Y. Kim, "Fiber to the Home Using a PON Infrastructure," Journal of Lightwave Technology, vol. 24, no. 12, pp. 4568-4583, 2006.
- P. Brandao Harboe and J. Rodolfo Souza, "Passive Optical Network: Characteristics, Deployment, and Perspectives," IEEE Latin America Transactions, vol. 11, no. 4, pp. 995-1000, 2013.

- A. A. M. Saleh and J. M. Simmons, "All-Optical Networking—Evolution, Benefits, Challenges, and Future Vision," Proceedings of the IEEE, vol. 100, no. 5, pp. 1105-1117, 2012.
- S. Kimura, "WDM/TDM-PON technologies for future flexible optical access networks," OECC 2010 Technical Digest, Sapporo, pp. 14-15, 2010.
- F. Saliou, P. Chanclou, F. Laurent, N. Genay, J. A. Lazaro, F. Bonada, and J. Prat, "Reach extension strategies for passive optical networks," J. Opt. Commun. Netw. Vol.1, no.4, pp. C51–C60, 2009.
- R. Bavey, J. Kani, F. Bourgart, and K.McCammon, "Options for future optical access networks," IEEE Commun. Mag., vol. 44, no. 10, pp. 50–56, 2006.
- 29. J. L. Wei, K. Grobe, and H. Griesser, "Cost-efficient high-speed modulation for next-generation PONs," in Proceedings of Photonic Networks, 2015.
- J. L. Wei,K. Grobe, C. Sanchez, E. Giacoumidis and H. Griesser, "Comparison of cost- and energy-efficient signal modulations for next generation passive optical networks," Optics Express, vol. 23, no. 22, pp. 28271-28281, 2015.
- C. H. Yeh, C. W. Chow, and H. Y. Chen, "Simple colorless WDM-PON with Rayleigh backscattering noise circumvention employing m-QAM OFDM downstream and remodulated OOK upstream signals," J. Lightw. Technol., vol. 30, no. 13, pp. 2151–2155, 2012.
- 32. K. Iwatsuki and J. i. Kani, "Applications and Technical Issues of Wavelength-Division Multiplexing Passive Optical Networks With Colorless Optical Network Units [Invited]," IEEE/OSA Journal of Optical Communications and Networking, vol. 1, no. 4, pp. C17-C24, 2009.
- M. Fujiwara, J. Kani, H. Suzuki, and K. Iwatsuki, "Impact of backreflection on upstream transmission in WDM single-fiberloop back access networks," J. Lightwave Technol., vol. 24, no.2, pp. 786–796, 2006.
- Y Zhang, P. Chowdhury, M. Tornatore, and B. Mukherjee, 'Energy Efficiency in Telecom Optical Networks," IEEE Communications Surveys & Tutorials, vol. 12, no. 4, pp. 441-458, 2010.
- 35. C. H. Yeh, C. W. Chow, and C.H. Hsu,"40-Gb/s Time-Division-Multiplexed Passive Optical Networks Using Downstream OOK and Upstream OFDM Modulations," IEEE Photonics Technology Letters, vol. 22, no. 2, pp.118-120, Jan. 2010.

- E. Forestieri and G. Prati, "Novel optical line codes tolerant to fiber chromatic dispersion," Journal of Lightwave Technology, vol. 19, no. 11, pp. 1675 -1684, 2001.
- A. Banerjee, Y. Park, F. Clarke, H. Song, S. Yang, G. Kramer, K. Kim and B. Mukherjee, "Wavelength-division-multiplexed passive optical network (WDM-PON) technologies for broadband access: a review," J. Opt. Netw. vol. 4, pp. 737-758, 2005.
- C. W. Chow, G. Talli, and P. D. Townsend, "Rayleigh noise reduction in 10-Gb/s DWDM-PONs by wavelength detuning and phase-modulation-induced spectral broadening," IEEE Photon. Technol. Lett., vol. 19, no. 6, pp. 423–425, 2007.
- Z. Li, Y. Dong, Y. Wang, and C. Lu, "A novel PSK Manchester modulation format in 10-Gb/s passive optical network system with high tolerance to beat interference noise," IEEE Photon. Technol. Lett., vol. 17, no. 5, pp. 1118–1120, 2005.
- Z. Al-Qazwini and H. Kim, "DC-balanced line coding for downlink modulation in bidirectional WDM PONs using remodulation," IEEE Photon. Technol. Lett., vol. 23, no. 18, pp. 1331–1333, 2011.
- 41. C. W. Chow and C. H. Yeh, "40-Gb/s downstream DPSK and 40-Gb/s upstream OOK signal remodulation PON using reduced, vol. 38, no. 11, pp. 106–110, 2000.
- J. L. Locicero and B.P. Patel, "Line coding," Mobile communications Handbook. Ed. Suthan, S. Suthersan, Boca Raton: CRC Press, 1999.
- G. Katsaros, I. Darwazeh and P. M. Lane, "Non-linear transmission effects in duobinary and dicode optical systems," IEE Proceedings - Optoelectronics, vol. 152, no. 6, pp. 344-352, 2005.
- R.A. Cryan and M.J.N. Sibley, "Minimising intersymbol interference in opticalfibre dicode PPM systems," IEE Proc.-Optoelectron., vol. 153, no. 3, pp. 93-100, 2006.
- 45. F. C. Court, "Minimum bandwidth dicode signalling," Electronic and Radio Engineers, Journal of the Institution of, vol. 56, no. 4, pp. 159-168, 1986.
- M. J. N. Sibley, "Dicode pulse-position modulation: a novel coding scheme for optical-fibre communications," IEE Proceedings - Optoelectronics, vol. 150, no. 2, pp. 125-131, 2003.
- G. Katsaros, P. M. Lane and M. Murphy, "A comparison of the impact of FWM on binary, duobinary and dicode modulation in DWDM systems," LEOS 2000. Rio Grande, pp. 27-28 vol.1, 2000.

- 48. Q. Guo and A. V. Tran, "Combined utilization of partial-response coding and equalization for high-speed WDM-PON with centralized light waves," Opt. Express, vol. 20, pp. 27981-27991, 2012.
- 49. C. Lange and A. Gladisch, "On the energy consumption of FTTH access networks," OFC/NFOEC'09, San Diego, CA, 2009.
- K. Cho, K. Fukuda, H. Esaki, and A. Kato, "The impact and implication of the growth in residential user-to-user traffic," Proc. ACM/ SIGCOMM (SIGCOMM2006), Pisa, Italy, pp. 207–218, 2006.
- R. Fildes and V. Kumar, "Telecommunications demand forecasting—a review," Int. J. Forecast., vol. 18, no. 4, pp. 489–522, 2002.
- 52. M. Balmes, J. Bourne and J. Mar, "Fiber to the home: the technology behind Heathrow," IEEE LCS, vol. 1, no. 3, pp. 25-29, 1990.
- 53. P. W. Shumate, "Fiber-to-the-Home: 1977–2007," Journal of Lightwave Technology, vol. 26, no. 9, pp. 1093-1103, 2008.
- C. T. Lin et al., "Hybrid Optical Access Network Integrating Fiber-to-the-Home and Radio-Over-Fiber Systems," IEEE Photonics Technology Letters, vol. 19, no. 8, pp. 610-612, 2007.
- 55. N. Cheung, "Fiber to 30 million homes," in IEEE Communications Magazine, vol. 43, no. 9, pp. 70-71, 2005.
- K. A. Janjua and S. A. Khan, "A comparative economic analysis of different FTTH architectures," in Int. Conf. on Wireless Communications, Networking and Mobile Computing, pp. 4979–4982, 2007.
- K. Grobe, M. Roppelt, A. Autenrieth, J. P. Elbers and M. Eiselt, "Cost and energy consumption analysis of advanced WDM-PONs," IEEE Communications Magazine, vol. 49, no. 2, pp. s25-s32, February 2011.
- 58. K. Mokushi, H. Yoshioka, K. Takaya and T. Tominaga, "The influence of the electrical characteristics of premises cables on the interference immunity of VDSL," 2008 International Symposium on Electromagnetic Compatibility - EMC Europe, Hamburg, pp. 1-6, 2008.
- 59. ITU-T Recommendation G.993.1, "Very high speed digital subscriber line", 2004.
- K. Oshima, T. Kitayama, M. Yamaki, T. Matsui and K. Ito, "Fiber-optic local area passive network using burst TDMA scheme," Journal of Lightwave Technology, vol. 3, no. 3, pp. 502-510, 1985.

- J. W. Kim , "An Optimized ATM-PON Based FTTH Access Network," International Conference on Information, Communications and Signal Processing ICICS '9 Singapore, vol. 3, pp. 1800 – 1804, 9-12, 1997.
- 62. B. D. Vos et al., "Demonstration of extended split APON," Optical Fiber Communication Conference and Exhibit, pp. 437-439, 2002.
- J. D. Angelopoulos, I. S. Venieris and G. I. Stassinopoulos, "A TDMA based access control scheme for APON's," Journal of Lightwave Technology, vol. 11, no. 5, pp. 1095-1103, 1993.
- R. Schoop, F. Fredricx, T. Koonen and C. Hardalov, "WDM Isolation Requirements for CATV in BPON," 2002 28TH European Conference on Optical Communication, Copenhagen, pp. 1-2, 2002.
- 65. A. V. Tran, Chang-Joon Chae and R. S. Tucker, "Bandwidth-efficient PON system for broad-band access and local customer Internetworking," IEEE Photonics Technology Letters, vol. 18, no. 5, pp. 670-672, 2006.
- 66. M. Nakamura, H. Ueda, S. Makino, T. Yokotani and K. Oshima, "Proposal of networking by PON technologies for full and ethernet services in FTTx," Journal of Lightwave Technology, vol. 22, no. 11, pp. 2631-2640, 2004.
- 67. M. Gagnaire, "An overview of broad-band access technologies," in Proceedings of the IEEE, vol. 85, no. 12, pp. 1958-1972, 1997.
- L. D. Lamb, "Status and Evolution of EPON: From IPTV Today to 10 Gb/s and Beyond," 2006 European Conference on Optical Communications, Cannes, France, pp. 1-2, 2006.
- B. Skubic, J. Chen, J. Ahmed, L. Wosinska and B. Mukherjee, "A comparison of dynamic bandwidth allocation for EPON, GPON, and next-generation TDM PON," IEEE Communications Magazine, vol. 47, no. 3, pp. S40-S48, 2009.
- Gigabit-Capable Passive Optical Networks (GPON): General Characteristics, Rec. ITU-T G.984.1, 2008.
- Y. Yi et al., "Simulations and Experiments on the Effective Optical Gain of FEC in a GPON Uplink," IEEE Photonics Technology Letters, vol. 19, no. 2, pp. 82-84, 2007.
- C. A. Salihovic and M. Ivekovic, "Gigabit Passive Optical Network GPON," 2007 29th International Conference on Information Technology Interfaces, Cavtat, pp. 679-684, 2007.

- 73. M. S. Salleh et al., "The challenge for active and passive components design in CWDM PON system co-exist in GEPON and 10 GEPON architecture," 2011 2nd International Conference on Photonics, Kata Kinabalu, pp. 1-5, 2011.
- E. Skaljo, M. Hodzic and I. Bektas, "Migration from G(E)PON to NGPON," 2009 International Conference on Ultra-Modern Telecommunications & Workshops, St. Petersburg, pp. 1-4, 2009.
- 75. H. G. Krimmel, W. Poehlmann, B. Deppisch, L. Jentsch and T. Pfeiffer, "10 Gbps / 2.5Gbps GPON coexistence by downstream bit-stacking," 2008 34th European Conference on Optical Communication, Brussels, pp. 1-2, 2008.
- X. Chen, Z. Zhang and X. Hu, "The evolution trends of PON and key techniques for NG-PON," 2013 9th International Conference on Information, Communications & Signal Processing, Tainan, pp. 1-6, 2013.
- F. J. Effenberger, "The XG-PON System: Cost Effective 10 Gb/s Access," in Journal of Lightwave Technology, vol. 29, no. 4, pp. 403-409, 2011.
- 78. Y. Luo et al., "Time- and Wavelength-Division Multiplexed Passive Optical Network (TWDM-PON) for Next-Generation PON Stage 2 (NG-PON2)," in Journal of Lightwave Technology, vol. 31, no. 4, pp. 587-593, 2013.
- C. W. Chow and C. H. Yeh, "Technology advances for the 2nd stage nextgeneration passive-optical-network (NG-PON2)," 2013 6th IEEE/International Conference on Advanced Infocomm Technology (ICAIT), Hsinchu, 2013, pp. 83-84.
- C. W. Chow, C. H. Yeh, K. Xu, J. Y. Sung and H. K. Tsang, "TWDM-PON With Signal Remodulation and Rayleigh Noise Circumvention for NG-PON2," in IEEE Photonics Journal, vol. 5, no. 6, pp. 7902306-7902306, 2013.
- K. Prince et al., "GigaWaM—Next-Generation WDM-PON Enabling Gigabit Per-User Data Bandwidth," in Journal of Lightwave Technology, vol. 30, no. 10, pp. 1444-1454, 2012.
- C. Bouchat, C. Dessauvages, F. Fredricx, C. Hardalov, R. Schoop, and P. Vetter, "WDM-upgrade PONs for FTTH and FTTBusiness," in Proc. Int Workshop Opt. Hybrid Access Netw., Florence, Italy, pp. 231–238, 2002.
- 83. T. Dong, Y. Bao, Y. Ji, A. P. T. Lau, Z. Li and C. Lu, "Bidirectional Hybrid OFDM-WDM-PON System for 40-Gb/s Downlink and 10-Gb/s Uplink Transmission Using RSOA Remodulation," in IEEE Photonics Technology Letters, vol. 24, no. 22, pp. 2024-2026, 2012.

- Z. Al-Qazwini and H. Kim, "Ultranarrow Spectrum-Sliced Incoherent Light Source for 10-Gb/s WDM PON," in Journal of Lightwave Technology, vol. 30, no. 19, pp. 3157-3163, 2012.
- J. Chen and L. Wosinska, "Performance analysis of protection schemes compatible with smooth migration from TDM-PON to hybrid WDM/TDM-PON," J. Opt. Netw., vol. 6, no. 5, pp. 514–526, 2007.
- F.-T. An, K. S. Kim, D. Gutierrez, S. Yam, E. Hu, K. Shrikhande, and L. G. Kazovsky, "SUCCESS: A next-generation hybrid WDM/TDM optical access network architecture," J. Lightw. Technol., vol. 22, no. 11, pp. 2557–2569, 2004.
- G. Das, B. Lannoo, A. Dixit, D.Colle, M. Pickavet and P. Demeester, "Flexible hybrid WDM/TDM PON architectures using wavelength selective switches," Optical Switching and Networking, vol. 9, pp.159-169, 2015.
- T. Mitsui *et al.*, "Simple and scalable WDM/TDMA-PON using spectral slicing and forward error correction," 2011 Optical Fiber Communication Conference and Exposition and the National Fiber Optic Engineers Conference, Los Angeles, CA, pp. 1-3, 2011.
- F. J. Effenberger, "PON resilience," EEE/OSA Journal of Optical Communications and Networking, vol. 7, no. 3, pp. A547-A552, 2015.
- 90. J. D. Reis, B. Neto, P. S. Andre and A. Teixeira, "Optimization of Passive Optical Networks by means of fiber nonlinearities interference reduction," 2009 11th International Conference on Transparent Optical Networks, Azores, pp. 1-4, 2009.
- 91. A. Nag, D. B. Payne and M. Ruffini, "N:1 Protection design for minimizing OLTs in resilient dual-homed long-reach passive optical network," IEEE/OSA Journal of Optical Communications and Networking, vol. 8, no. 2, pp. 93-99, 2016.
- 92. R. P. Khare, "Fiber Optics and Optoelectronics," Oxford University, 2004.
- 93. N. k. Chen, K. c. Hsu, S. Chi and Y. Lai, "Tunable Er3+-doped Fiber Amplifiers Covering S-and C + L-bands (1490 ~ 1610 nm) Using Discrete All-fiber ASE Suppressing Filters," LEOS 2006 - 19th Annual Meeting of the IEEE Lasers and Electro-Optics Society, Montreal, pp. 396-397, 2006.
- 94. A. Hill and D. Payne, "Linear crosstalk in wavelength-division-multiplexed optical-fiber transmission systems," Journal of Lightwave Technology, vol. 3, no. 3, pp. 643-651, 1985.
- 95. J. Zhou et al, "Cross talk in Multi-wavelength Optical Cross-Connect Networks," Journal of light wave technology, vol. 14, no. 6, 1996.

- 96. J. D. Reis, L. N. Costa and A. L. Teixeira, "Nonlinear effects prediction in ultradense WDM systems using volterra series," 2010 Conference on Optical Fiber Communication (OFC/NFOEC), collocated National Fiber Optic Engineers Conference, San Diego, CA, pp. 1-3, 2010.
- 97. B. Barua and M. Islam, "Evaluate the effect of Stimulated Raman Scattering in DWDM transmission system with direct detection binary WSK receiver," 2012 International Conference on Informatics, Electronics & Vision (ICIEV), Dhaka, pp. 698-701, 2012.
- A. Kobyakov, M. Sauer, and D. Chowdhury, "Stimulated Brillouin scattering in optical fibers," Adv. Opt. Photon. Vol. 2, pp. 1-59, 2010.
- G. P. Agrawal, "Self-Phase Modulation in Optical Fiber Communications: Good or Bad?," 2007 Conference on Lasers and Electro-Optics (CLEO), Baltimore, MD, pp. 1-2, 2007.
- 100.S. Kumar and Dong Yang, "Second-order theory for self-phase modulation and cross-phase modulation in optical fibers," in Journal of Lightwave Technology, vol. 23, no. 6, pp. 2073-2080, 2005.
- 101.Zhaoming Zhu and D. J. Gauthier, "XPM-induced pulse delay and advancement in optical fiber," 2006 Conference on Lasers, Electro-Optics and Quantum Electronics and Laser Science Conference, Long Beach, pp. 1-2, 2006.
- 102.S. Betti and M. Giaconi, "Effect of the cross-phase modulation on WDM optical systems: analysis of fiber propagation," IEEE Photonics Technology Letters, vol. 13, no. 4, pp. 305-307, 2001.
- 103.S. Fatimah Tuanku Abdullah and S. A. Jabar, "FWM effect in FTTH over DWDM for 320 Gbps link distance 80 km," 2012 International Conference on Computer and Communication Engineering (ICCCE), Kuala Lumpur, pp. 633-637, 2012.
- 104.E. Kretzmer, "An Efficient Binary Data Transmission System," IEEE Transactions on Communications Systems, vol. 12, no. 2, pp. 250-251, 1964.
- 105.H. Nyquist, "Certain factors affecting telegraph speed," The Bell System Technical Journal, vol. 3, no. 2, pp. 324-346, 1924.
- 106.D. Sadot, "Pushing optical fiber communications to the Shannon limit: Advanced modulation formats and digital signal processing," 2016 18th International Conference on Transparent Optical Networks (ICTON), Trento, pp. 1-3, 2016.

- 107.S. Jawla, R.K.Singh, "Different Modulation Formats Used In Optical Communication System," IOSR Journal of Electronics and Communication Engineering (IOSR-JECE),vol. 8, no. 4, pp. 15-18, 2013.
- 108.C. H. Yeh, C. W. Chow, C. H. Wang, Y. F. Wu, F. Y. Shih and S. Chi, "Using OOK Modulation for Symmetric 40-Gb/s Long-Reach Time-Sharing Passive Optical Networks," IEEE Photonics Technology Letters, vol. 22, no. 9, pp. 619-621, 2010.
- 109.P. J. Winzer and R. J. Essiambre, "Advanced Modulation Formats for High-Capacity Optical Transport Networks," Journal of Lightwave Technology, vol. 24, no. 12, pp. 4711-4728, 2006.
- 110.J. Vardakas, I. Moscholios, M. Logothetis and V. Stylianakis, "An Analytical Approach for Dynamic Wavelength Allocation in WDM–TDMA PONs Servicing ON–OFF Traffic," IEEE/OSA Journal of Optical Communications and Networking, vol. 3, no. 4, pp. 347-358, 2011.
- 111.A. J. Price and N. LeMercier, "Reduced bandwidth optical digital intensity modulation with improved chromatic dispersion tolerance," Electron. Lett., vol. 31, no. 1, pp. 58–59, 1995.
- 112.M. I. Hayee and A. E. Willner, "NRZ versus RZ in 10–40-Gb/s dispersionmanaged WDM transmission systems," IEEE Photon. Technol. Lett., vol. 11, no. 8, pp. 991–993, 1999.
- 113. C. Caspar, H. M. Foisel, A. Gladisch, N. Hanik, F. Küppers, R. Ludwig, A. Mattheus, W. Pieper, B. Strebel, and H. G. Weber, "RZ versus NRZ modulation format for dispersion compensated SMF-based 10 Gb/s transmission with more than 100-km amplifier spacing," IEEE Photon. Technol. Lett., vol. 11, no. 4, pp. 991–993, 1999.
- 114. D. Breuer and K. Petermann, "Comparison of NRZ- and RZ-modulation format for 40-Gb/s TDM standard-fiber systems," IEEE Photon. Technol. Lett., vol. 9, no.
  3, pp. 398–400, 1997.
- 115. H. Sunnerud, M. Karlsson, and P. A. Andrekson, "A comparison between NRZ and RZ data formats with respect to PMD-induced system degradation," IEEE Photon. Technol. Lett., vol. 13, no. 5, pp. 448–450, 2001.
- 116.M. Ohm and J. Speidel, "Quaternary optical ASK-DPSK and receivers with direct detection," IEEE Photon. Technol. Lett., vol. 15, no. 1, pp. 159–161, 2003.

- 117.T. Mizuochi, K. Ishida, T. Kobayashi, J. Abe, K. Kinjo, K. Motoshima, and K. Kasahara, "A comparative study of DPSK and OOK WDM transmission over transoceanic distances and their performance degradations due to nonlinear phase noise," J. Lightw. Technol., vol. 21, no. 9, pp. 1933–1943, 2003.
- 118.K. Yonenaga and S. Kuwano, "Dispersion-tolerant optical transmission system using duobinary transmitter and binary receiver," Journal of Lightwave Technology, vol. 15, no. 8, pp. 1530-1537, 1997.
- 119.X. Gu and L. C. Blank, "10 Gb/s unrepeatered three-level optical transmission over 100 km of standard fiber," Electron. Lett., vol. 29, pp. 2209–2211, 1993.
- 120.E. M. Kimber, X. Gu, S. J. Pycock, E. Pittuck, B. L. Patel, D. M. Spirit, J. Wakefield, A. D. Ellis, and C. J. Anderson, "Field technology demonstration of unrepeatered, 2 ×10 Gb/s narrow band WDM over 120 km of installed step-index fiber," in Proc. ECOC'94, post-deadline papers, pp. 31–34, 1994.
- 121.G. Talli and P. D. Townsend, "Hybrid DWDM-TDM long-reach PON for nextgeneration optical access," J. Lightw. Technol., Vol. 24, No. 7, pp. 2827–2834, 2006.
- 122.H. Kim, et al., "Performance Limitations of Hybrid WDM Systems Due to Stimulated Raman Scattering", IEEE Photon. Technol. Lett., Vol.13, No.10, pp1118-1120, 2001.
- 123.F. Coppinger et al., "Nonlinear Raman Cross-Talk in a Video Overlay Passive Optical Network", OFC2003, TuR5, 2003.
- 124.A. Agata, et al., "Influence of Stimulated Raman Scattering in Video Overlay EPON Systems on Downstream Data Signal", ECOC2005, Tu1.3.5, 2005.
- 125.J. Wang et al, "Mitigation of Raman Crosstalk in BPON Applications", FTTH Council Annual Conference, 2004.
- 126.A. Murakami, A. Agata, K. Imai, Y. Horiuchi, and N. Edagawa, "Study of a Raman crosstalk reduction scheme for analog video signal transmission over metro area network with Gigabit Ethernet an Fiber Channel signals," Proc. Optoelectronics and Commun. Conf., Anaheim, CA, 2005.
- 127.H. Kim, S. Jun, and Y. C. Chung, "Raman Crosstalk Suppression in CATV Overlay Passive Optical Network," IEEE Photonics Technology Letters, vol. 19, no. 9, 2007.

- 128.K. Acharya and M. Y. A. Raja, "SRS crosstalk mitigation in WDMPON using quadrature amplitude modulation," HONET 2007 Proc. IEEE Xplore, pp. 1–5, 2008.
- 129.K. Acharya and M. Y. A. Raja, "Cross-talk penalty compensation and reach extension in wavelength division multiplexed passive optical network using remote Raman pump" Optical Engineering vol. 49, no.8, 2010.
- 130.X. Chen, P. R. Horche and A. M. Minguez, "Analysis of signal impairment and crosstalk penalty induced by different types of optical filters in 100 Gbps PM-DQPSK based systems," 2014 19th European Conference on Networks and Optical Communications - (NOC), Milano, pp. 35-40, 2014.
- 131.D. Breuer et al., "Opportunities for Next Generation Optical Access," IEEE Commun. Mag., vol. 49, pp. 16–24, 2011.
- 132.M. Mahloo et al," Toward Reliable Hybrid WDM/TDM Passive Optical Networks," IEEE Communications Magazine, vol. 52, pp. S14-S13, 2014.
- 133.M. K. Gupta, J. Agarwal, A. Dhingra and G. Singh, "Performance Analysis and Optimization of 40 Gbps Transmission System over 4000 km with FBG", Proceedings of the International Conference on Recent Cognizance in Wireless Communication & Image Processing, Jaipur, pp.759-765, 2016.
- 134.F. Tian, R. Hui, B. Colella, and D. Bowler, "Raman crosstalk in fiber-optic hybrid CATV systems with wide channel separations," IEEE Photon. Technol. Lett., vol. 16, no. 1, pp. 344–346, 2004.
- 135.H. A. Fevrier and M. W. Chbat," Raman amplification technology for bandwidth extension," Proc. in the IEEE LEOS, San Diego, pp. 344-345, 2001.
- 136.Y. Lu, J. Liu, X. Hong and D. Zeng, "A Novel Evolution Method for Hybrid WDM/TDM-PON Based on DPSK/NRZ Orthogonal Modulation," Communications and Photonics Conferenc and Exhibition (ACP), Asia, pp. 1-6, 2009.
- 137.J. Kahn and K.-P. Ho, "Spectral efficiency limits and modulation/detection techniques for dwdm systems," Selected Topics in Quantum Electronics, IEEE Journal of, vol. 10, no. 2, pp. 259–272, 2004.
- 138.G. W. Lu, T. Miyazaki" Spectrum Efficient 80-Gb/s DPSK transmitter using phase interleaving technology without optical time or polarization division multiplexing" Opt. Eng., vol. 47, no.5, pp. 050501, 2008.

- 139.J. M. Khan and K. P. Ho, "Spectral efficiency limits and modulation/detection techniques for DWDM systems," IEEE journal of selected topics in quantum electronics, vol. 10, no. 2, pp. 259-272, 2004.
- 140.J. G. Proakis, Digital Communications, 4th Ed., McGraw-Hill, 2000.
- 141.C. Headley and G. P. Agrawal, "Raman Amplification in Fiber optical communication systems," Elsevier, 2005.
- 142.D. Kochera, R.S. Kalera and R. Randhawa, "Simulation of fiber to the home triple play services at 2 Gbit/s using GE-PON architecture for 56 ONUs," Optik -International Journal for Light and Electron Optics, vol. 124, no. 21, pp. 5007– 5010, 2013.
- 143.H. A. Bakarman, S. Shaari and M. Ismail, "Simulation of 1.25 Gb/s downstream transmission performance of GPON-FTTx," International Conference on Photonics 2010, Langkawi, Kedah, pp. 1-5, 2010.
- 144.R. Kalera and R.S. Kaler, "Simulation of Fiber to the Home at 10 Gbit/s using GE-PON architecture," Optik International Journal for Light and Electron Optics, vol. 122, No. 15, pp. 1362–1366, 2011.
- 145.N. Cheng, M. Zhou and F. J. Effenberger, "10 Gbit/s delay modulation using a directly modulated DFB laser for a TWDM PON with converged services [invited]," in IEEE/OSA Journal of Optical Communications and Networking, vol. 7, no. 1, pp. A87-A96, 2015.
- 146.R. S. Penze, J. B. Rosolem, U. R. Duarte, G. E. R. Paiva and R. B. Filho, "Fiber powered extender for XG-PON/G-PON applications," IEEE/OSA Journal of Optical Communications and Networking, vol. 6, no. 3, pp. 250-258, 2014.
- 147. ITU-T recommendation G.989.1, "40-Gigabit-capable passive optical networks (NG-PON2): General requirements," 2013.
- 148.S. M. Posada, G. Maier, F. Giannone, L. Valcarenghi, A. Marotta and C. Antonelli, "Comparison of multi-operator PON technologies beyond NG-PON2: A real greenfield case-study," 2016 18th International Conference on Transparent Optical Networks (ICTON), Trento, 2016, pp. 1-4.
- 149. J. Smith, R. Loveland, and B. Rao, "The migration to 10G GPON," FTTH Prism, vol. 7, pp. 19–25, 2010.
- 150. J. Li, H. He and W. Hu, "Power Depletion and Crosstalk Induced by Stimulated Raman Scattering in WDM Fronthaul," IEEE Photonics Technology Letters, vol. 28, no. 10, pp. 1069-1072, 2016.

- 151.J. Li, M. Bi and H. He, "Suppression of SRS induced crosstalk in RF-video overlay TWDM-PON system using dicode coding" Optics Express, vol. 22, no. 18, 2014.
- 152.P. Ossieur et.al, "A 135-km 8192-split carrier distributed DWDM-TDMA PON with 2 32 10 Gb/s capacity," J. Lightw. Technol., vol. 29, no. 4, pp. 463–474, 2011.
- 153.E. K. MacHale, G. Talli, P. D. Townsend, A. Borghesani, I. Lealman, D. G. Moodie, and D. W. Smith, "Extended-reach PON employing 10Gb/s integrated reflective EAM-SOA," presented at the Eur. Conf. Opt. Commun., Brussels, Belgium, Paper Th.2.F.1, 2008.
- 154.P .P. Iannone et al, "Bi-directionally amplified extended reach 40 Gb/s CWDM-TDMPON with burst-mode upstream transmission," in Proc. Opt. Fiber Commun. / Nat. Fiber Opt. Eng. Conf., pp. 1–3, 2011.
- 155.T. Tsutsumi and T. sakamoto, "Long-Reach and High-Splitting-Ratio 10G-EPON System with Semiconductor Optical Amplifier and N: 1 OSU Protection," Journal of Lightwave Technology, vol.33, pp. 1660-1665, 2015.
- 156.G. P. Agarwal, "nonlinear fibre optics," 3rd edition, 2001.
- 157.D. Marcuse, "Pulse distortion in single-mode fibers," Appl. Opt., vol. 19, pp. 1653-1660, 1980.
- 158.N. Sultana and M. S. Islam, "Analysis of XPM effect with SPM and GVD in WDM fiber optic transmission system," 2012 International Conference on Computer and Communication Engineering (ICCCE), Kuala Lumpur, pp. 263-266, 2012.
- 159.C. Peucheret, N. Hanik, R. Freund, L. Molle, and P. Jeppesen, "Optimization of Pre- and Post-Dispersion Compensation Schemes for 10-Gbits/s NRZ Links Using Standard and Dispersion Compensating Fibers," IEEE Photonics Technology Letters, vol. 12, no. 8, pp. 992-994, 2000.
- 160.N. Cheng and John C. Cartledge, "Power penalty due to the amplitude and phase response ripple of a dispersion compensating fiber Bragg grating for chirped optical signals," IEEE Journal of Lightwave Technology, vol. 24, no. 9, pp. 3363– 3369, 2006.
- 161.J. Vardakas, I. Moscholios, M. Logothetis and V. Stylianakis, "An Analytical Approach for Dynamic Wavelength Allocation in WDM–TDMA PONs Servicing ON–OFF Traffic," in IEEE/OSA Journal of Optical Communications and Networking, vol. 3, no. 4, pp. 347-358, 2011.

- 162.A. Ryset, S. Y. Set, I. A. Goncharenko, and R. I. Laming, "Linear and Nonlinear Dispersion Compensation of Short Pulses Using Midspan Spectral Inversion," IEEE Photonics Technology Letters, vol. 8, no. 3,pp. 449-451, 1996.
- 163.Gigabit-Capable Passive Optical Network (GPON), ITU-T Recommendation G. 984.1, 2003.
- 164.S. Jain, F. Effenberger and A. Szabo, "World's First XG-PON Field Trial," Journal of Lightwave Technology, vol. 29, no. 4, pp. 524-528, 2011.
- 165.S. G. Mun, J. H. Moon, H. K. Lee, J. Y. Kim, and C. H. Lee, "A WDM-PON with a 40 Gb/s capacity based on wavelength-locked Fabry–Perot laser diodes," in Proc. ECOC, Brussels, Belgium, 2008.
- 166.H. Kimura, Y. Sakai, N. Hyama, and K. Kumozaki, "Service rate matching optical receiver module with bit rates over 40 Gb/s for PS-based WDM PON systems," in Proc. LEOS, Newport Beach, CA, 2008.
- 167.C. M. Weinert, R. Ludwig, W. Pieper, H. G. Weber, D. Breuer, K. Petermann and F. Kuppers, "40 Gb/s and 4×40 Gb/s TDM/WDM Standard Fiber Transmission," IEEE Journal Of Lightwave Technology, vol. 17, no. 11, pp. 2276–2284, 1999.
- 168.P. N. Suman, R. chandan and D. Singh, "Optimization of WDM PON System for Long Haul Optical Communication based on FBG", International Journal of Scientific Engineering and Technology, vol. 2, no. 7, pp. 749-751, 2013.
- 169.G. Gnanagurunathan and F. A. Rahman, "Comparing FBG and DCF as dispersion compensators in the long haul narrowband WDM systems," IEEE International Conference on Wireless and optical communications networks, pp. 1-4, 2006.
- 170. Y. Khan, X. Xin, A. Hussain, L. Bo and S. Shafique, "Generation and transmission of dispersion tolerant 10-Gbps RZ-OOK signal for radio over fiber link," Front. Optoelectron, vol. 5, no. 3, pp. 306–310, 2012.
- 171.C. W. Chow, L. Xu, C. H. Yeh ,H. K. Tsang, W. Hofmann, and M. C. Amann, "40 Gb/s upstream transmitters using directly-modulated 1.55 μm VCSEL array for high-split-ratio PONs," IEEE Photon. Technol. Lett., vol. 22, no. 5, pp. 347–349, 2010.
- 172.S. L. Lee, "Performance enhancement of WDM-PON systems with DMLs and RSOAs," presented at the Annu. Wireless and Opt. Commun. Conf., Taichung, Taiwan, 2008.
- 173.C. H. Yeh, C. W. Chow, Y. L. Liu, and C. L. Pan, "40Gb/s on-off keying downstream and 10 Gb/s on-off keying remodulated upstream signals in long-

reach access network with multi-video services," Opt. Eng., vol. 50, pp. 125008, 2011.

- 174.C. H. Yeh, C. W. Chow, S. P. Huang, J. Y. Sung, Y. L. Liu, and C. L. Pan, "Ring-Based WDM Access Network Providing Both Rayleigh Backscattering Noise Mitigation and Fiber-Fault Protection," J. Lightw. Technol., vol. 30, pp. 3211-3218, 2012.
- 175.C. H. Yeh, C. W. Chow, F. Y. Shih, Y. F. Wu, and J. Y. Sung, "Fiberfault protection WDM-PON using new apparatus in optical networking unit," Opt. Commun., vol. 285, pp. 1803–1806, 2012.
- 176.Z. X. Wang, X. F. Sun, C. Lin, C. K. Chan, and L. K. Chen, "A novel centrally controlled protection scheme for traffic restoration in WDM passive optical networks," IEEE Photon. Technol. Lett., vol.17, no.3, pp. 717–719, Mar. 2005.
- 177.S. Y. Kim, S. B. Jun, Y. Takushima, E. S. Son, and Y. C. Chung, "Enhanced performance of RSOA-based WDM PON by using manchester coding," J. Opt. Netw., vol. 6, pp. 624–630, 2007.
- 178.M. S. Ahsan, M. S. Lee, S. H. Shah Newaz and S. M. Asif, "Migration to the Next Generation Optical Access," Journal of networks, vol. 6, no. 1, pp. 18-25, 2011.
- 179.M.K. Gupta, P. Dali and G. Singh , "A novel approach to enhance the performance of ring based WDM PON," 2014 International Conference on Signal Propagation and Computer Technology (ICSPCT), Ajmer, India, pp. 329-331, 2014.
- 180.D. Kocher, R. S. Kaler and R. Randhawa "Simulation of fiber to the home triple play services at 2 Gbit/s using GE-PON architecture for 56 ONUs," Optik, vol. 124, pp. 5007–5010, 2013.
- 181.S. Singh and R.S. Kaler, "Placements of Hybrid/conventional optical amplifiers in a multi-wavelength dense metropolitan area network," Optoelectronics and Advanced Materials-Rapid Communications, vol.8, pp. 762-769, 2014.
- 182.J. S. Wey et al., "Physical layer aspects of NG-PON2 standards—Part 1: Optical link design," IEEE/OSA Journal of Optical Communications and Networking, vol. 8, no. 1, pp. 33-42, 2016.
- 183.R. Goyal and R. S. Kaler, "Investigation on polarization dependent bidirectional hybrid (WDM/TDM) utilizing QAM modulation with different amplifiers," Optoelectronics and Advanced Materials-Rapid Communications, vol.8, pp. 631-634, 2014.

- 184.Q. Guo and A. V. & Tran, "Performance Enhancement in RSOA-Based WDM Passive Optical Networks Using Level Coding," Journal of Lightwave Technology. vol. 31, pp. 67-73, 2013.
- 185.M. K. Gupta, D. Kumar, A. K. Garg, and G. Singh, Analysis and mitigation of Raman cross talk in WDM-TDM system using partial response coding and Raman pump. IEEE International Conference on Microwave and Photonics. Proc. IEEE, 1-2, 2015.

## **Publications**

#### Published:

- Mukesh Kumar Gupta and Ghanshyam Singh "Group Velocity Dispersion Tolerant WDM TDM Passive Optical Networks" 2017 IEEE International Conference on Computer, Communications and Electronics: COMPTELIX, pp. 521-526, 2017.
- Mukesh Kumar Gupta, Ghanshyam Singh "2×40 Gbps WDM-PON Long Haul system using FBG and Dicode" IEEE International Conference, WRAP-2015, IISC Bangalore, India, 16-17 Dec. 2015, *IEEE Xplore Digital Library* DOI: 10.1109/WRAP.2015.7806001, 05 January 2017.
- Mukesh Kumar Gupta; Jyoti Agarwal; A. Dhingra; Ghanshyam Singh, "Performance Analysis and Optimization of 40 Gbps Transmission System over 4000 km with FBG", Proc. of ICRCWIP 2015, ISBN e-Book/hardcover: 978-81-322-2638-3/978-81-322-2636-9, pp.759-765, DOI 10.1007/978-81-322-2638-3, pp 759-768, Springer Publication Jaipur, April 2016
- Mukesh Kumar Gupta, Dinesh Kumar, Amit Kumar Garg, Ghanshyam Singh "Analysis and mitigation of Raman cross talk in WDM-TDM system using partial response coding and Raman pump" IEEE Conference, ICMAP 2015, *IEEE Xplore Digital Library, print ISBN: 978-1-4673-6897-1, p.p. 1-2, DOI: 10.1109/ICMAP.2015.7408763*, December 11-13, 2015.
- Mukesh Kumar Gupta, Ghanshyam Singh "Extend the Reach of Hybrid WDM-PON using Dicode coded NRZ" *Proceedings of IEICE international conference*; Thailand-Japan Microwave (TJMW 2014), Bangkok, November 26<sup>th</sup> – 28<sup>th</sup> 2014.
- Mukesh Kumar Gupta., Dali P. P., Singh Ghanshyam "A novel approach to enhance the performance of ring based WDM PON" IEEE International Conference on Signal Propagation and Computer Technology (ICSPCT), 2014, *IEEE Xplore Digital Library, print ISBN: 978-1-4799-3139-2, pp 329-331 DOI:* 10.1109/ICSPCT.2014.6885012, 12-13 July 2014.

#### Communicated (Under Review):

- Mukesh Kumar Gupta, Michael V. Dashkov and Ghanshyam Singh "System Capacity and Reach improvement in flexible Hybrid WDM-TDM- PON using Dicode coding" Submitted to Optica Applicata, 2017.
- Mukesh Kumar Gupta, Ambrish and Ghanshyam Singh "Next Generation PON with Enhanced Spectral Efficiency: Analysis and Design" Submitted to Optik-International Journal for Light and Electron Optics, (Under review), March 2016.

### National Conferences:

- Deepak Singh, Mukesh Kumar Gupta and Ghanshyam Singh "Mitigation of Raman cross talk in co-existed GPON and XGPON" Proceedings of OSA Young Student Congress on Photonic Technologies, MNIT Jaipur, pp.-22, April 16 – 17, 2016.
- Amit Kumar Garg, Mukesh Kumar Gupta, Ghanshyam Singh and Vijay Janyani, "Flexible Hybrid WDM–TDM–PON Architecture" Proceedings of IETE Zonal Seminar on ICT in Healthcare, MNIT Jaipur, pp.-13, 5th March 2016.
- 3. Srishti Gupta, Mukesh Kumar Gupta, and Ghanshyam Singh "WDM-PON architecture based on Central single Fiber ring topology and secondary tree networks," Proceedings of IETE Zonal Seminar (National level conference) on future trends in communication technologies, Manipal University Jaipur, 22-23 august 2014.
- 4. Shivika Pathania, Mukesh Kumar Gupta, and Ghanshyam Singh "Ultra DWDM with 64 channels at 5Gbps, Analyzing compensation schemes and effect of nonlinearity," Proceedings of IETE Zonal Seminar (National level conference) on future trends in communication technologies, Manipal University Jaipur, 22-23 August 2014.
- 5. Mukesh Kumar Gupta, Gaurav Garg, and Ambrish Kumar "Generation and Simulation of 80 GHz MM Signal using Stimulated Brillouin Scattering for Radio over Fiber System," Proceedings of National Conference on Recent Developments in Wireless and Optical Technologies, MNIT Jaipur, Oct. 2012.

# **Profile Summary**

Mukesh Kr Gupta received the Bachelor of Engineering degree in Electronics and Communication Engineering from University of Rajasthan, Jaipur (Year: 2006) and Master of Technology in Microwave & optical Communication from Delhi Technological University (Formerly Delhi College of Engineering), New Delhi (Year: 2012). He is pursuing Ph.D. degree from MNIT Jaipur. He is an active member of Society of Photo-Optical Instrumentation Engineers (SPIE) and the Optical Society of America (OSA). His current research includes optical communication, optoelectronics devices and high speed Gigabit Passive Optical Networks (PONs).